## SECURITIES AND EXCHANGE COMMISSION

## FORM 8-K

Current report filing

Filing Date: **2022-07-01** | Period of Report: **2022-07-01** | SEC Accession No. 0001193125-22-186782

(HTML Version on secdatabase.com)

## **FILER**

## POPULAR, INC.

CIK:**763901**| IRS No.: **660667416** | Fiscal Year End: **1231** Type: **8-K** | Act: **34** | File No.: **001-34084** | Film No.: **221059876** 

SIC: 6022 State commercial banks

Mailing Address P.O. BOX 362708 SAN JUAN PR 00936-2708

Business Address
209 MUNOZ RIVERA AVE
POPULAR CENTER
BUILDING
HATO REY PR 00918
7877659800

# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

F	orn	n 8	-K

## CURRENT REPORT Pursuant to Section 13 OR 15(d)

of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): July 1, 2022

## POPULAR, INC.

(Exact Name of Registrant as Specified in its Charter)

Puerto Rico (State or Other Jurisdiction of Incorporation) 001-34084 (Commission File Number) 66-0667416 (IRS Employer Identification No.)

209 Muñoz Rivera Avenue Hato Rey, Puerto Rico (Address of Principal Executive Offices)

00918 (Zip Code)

(787) 765-9800 (Registrant's telephone number, including area code)

## NOT APPLICABLE

(Former name, former address and former fiscal year, if changed since last report)

	-		
	e appropriate box below if the Form 8-K filing is provisions ( <i>see</i> General Instruction A.2. below):	intended to simultaneously satisfy the	he filing obligation of the registrant under any of the
	Written communications pursuant to Rule 425 u	under the Securities Act (17 CFR 230	.425)
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)		
	□ Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))		
	Pre-commencement communications pursuant	to Rule 13e-4(c) under the Exchange	Act (17 CFR 240.13e-4(c))
Securities	registered or to be registered pursuant to Section 1	2(b) of the Act:	
	Title of each class	Trading Symbols(s)	Name of each exchange on which registered
	Common Stock (\$0.01 par value) 6 Cumulative Monthly Income Trust Preferred Securities	BPOP BPOPM	The NASDAQ Stock Market The NASDAQ Stock Market
	y check mark whether the registrant is an emergin r Rule 12b-2 of the Securities Exchange Act of 193		le 405 of the Securities Act of 1933 (§230.405 of this
Emerging	growth company □		
	ging growth company, indicate by check mark if th financial accounting standards provided pursuant t	2	extended transition period for complying with any new $\Box$

## Item 1.01. Entry into a Material Definitive Agreement.

### Second Amended and Restated Master Service Agreement

As previously disclosed on February 24, 2022, Popular, Inc. ("Popular" or the "Corporation") entered into an Asset Purchase Agreement, dated as of February 24, 2022 (as amended from time to time, the "Asset Purchase Agreement"), by and among Popular, its wholly-owned subsidiary Banco Popular de Puerto Rico ("BPPR"), Evertec, Inc. ("Evertec") and Evertec Group, LLC ("Evertec Group"). Completion of the transactions contemplated by the Asset Purchase Agreement (collectively, the "Evertec Transaction") occurred on July 1, 2022 (the "Closing Date").

In connection with the closing of the Evertec Transaction and as previously disclosed on February 24, 2022, Popular, BPPR and Evertec Group entered into a Second Amended and Restated Master Service Agreement (the "Second A&R MSA") on July 1, 2022, pursuant to which Evertec Group will continue to provide various key information technology and various transaction processing services to Popular, BPPR and their respective subsidiaries, which services were previously provided under the Amended and Restated Master Service Agreement (the "Original MSA"), dated September 30, 2010, among Popular, BPPR and Evertec Group. In addition, the Second A&R MSA provides for the matters described below.

Under the Second A&R MSA, Popular and BPPR are no longer subject to exclusivity provisions that existed under the Original MSA and required Popular and BPPR to obtain certain services from Evertec, nor are they subject to rights of first refusal that Evertec had under the Original MSA with respect to certain technology projects. In connection with the elimination of exclusivity provisions under the Original MSA, Evertec Group is now entitled to receive monthly payments from Popular and BPPR to the extent that Evertec Group's revenues under the Second A&R MSA fall below certain agreed minimum amounts on an annualized basis (each, an "Annual Minimum"). The Annual Minimum will equal (i) \$170,000,000 for each one-year period from the Closing Date through September 30, 2025 (which is approximately 93% of the annual fees paid under the Original MSA in 2021 (disregarding the fees that were paid for services that have been removed from the scope of the Second A&R MSA, and disregarding the increase in fees attributable to the Annual MSA CPI Escalation (as defined below) that became effective as of October 1, 2021)); (ii) \$165,000,000 for each one-year period from October 1, 2025 through September 30, 2026; and (iii) \$160,000,000 for each one-year period from October 1, 2026 through September 30, 2028 (in each case, pro-rated for any partial one-year period).

Under the Original MSA, Evertec was entitled to increase annually the fees charged under the agreement based on the annual increases in the Consumer Price Index (the "<u>Annual MSA CPI Escalation</u>"), subject to an annual cap of 5%. Under the Second A&R MSA, the Annual MSA CPI Escalation that became effective as of October 1, 2021 has been retroactively eliminated, and Popular and BPPR are entitled to a credit against fees payable under the Second A&R MSA equal to the amount by which the fees paid by Popular or BPPR for the period from October 1, 2021 through the Closing Date were increased as a result of the most recent Annual MSA CPI Escalation. Additionally, the cap on the Annual MSA CPI Escalation has been reduced relative to the Original MSA, which provided for an Annual MSA CPI Escalation of up to 5% each year. Specifically, under the Second A&R MSA, the Annual MSA CPI Escalation is capped (i) at 1.5% for each one-year period beginning on the Closing Date through September 30, 2025, and (ii) at 2% for each one-year period from October 1, 2025 through September 30, 2028 (or if lower, at the percentage by which the CPI increase during the prior one-year period exceeded 2%). In addition, beginning in October 2025, BPPR will receive a 10% fee discount for services provided under the Second A&R MSA.

The Second A&R MSA requires Evertec Group to deliver to BPPR a core application programming interface (the "<u>Core API Layer</u>"). The Core API Layer will be a middle layer to connect the assets acquired by BPPR in connection with the Evertec Transaction to existing Evertec Group core applications in a manner expected to enable BPPR's ability to enhance its customer facing digital channels in the future. The deadline for the delivery of the Core API Layer is January 31, 2024, subject to certain terms and conditions. If the delivery of the Core API Layer is delayed beyond the applicable deadline, Evertec Group will be required to make monetary payments to BPPR in accordance with an agreed schedule, and if the delay persists beyond a period of 12 months to 18 months following such deadline (such period depending on the delivery date that is determined for the Core API Layer), Popular or BPPR may terminate the Second A&R MSA.

The Second A&R MSA provides for a term that commenced on the Closing Date and ends on September 30, 2028 (a three-year extension of the term of the Original MSA), unless earlier terminated in accordance with its terms. The Second A&R MSA provides for termination by a party (i) for the other party's material breach of the agreement, (ii) for a failure by the other party to pay any properly submitted invoice for a material amount in the aggregate that

is undisputed for a period of more than 60 days, or (iii) for a prohibited assignment of the agreement by the other party. Additionally, Popular or BPPR will have the right to terminate services provided under the Second A&R MSA at any time with 180 days prior written notice, and will have the right to terminate the Second A&R MSA in its entirety if certain deadlines are missed with respect to the delivery of the Core API Layer, as discussed above. Additionally, Evertec Group may terminate certain services provided under the Second A&R MSA that have been unused or unconsumed by Popular or BPPR for longer than 90 days by providing 60 days prior written notice.

Under the Second A&R MSA, Evertec Group has agreed to provide transition assistance to Popular, BPPR and their respective subsidiaries in connection with a full termination of the Second A&R MSA or the termination of individual services provided under the Second A&R MSA.

As of the Closing Date, Popular owns approximately 10.6% of Evertec's issued and outstanding common stock.

The foregoing description of the Second A&R MSA is qualified in its entirety by reference to the full text of such agreement, a copy of which is filed as an exhibit hereto and is fully incorporated by reference.

## Item 1.02. Termination of a Material Definitive Agreement.

## **Termination of the Stockholder Agreement**

In connection with the closing of the Evertec Transaction, on the Closing Date the parties terminated the Stockholder Agreement (the "Stockholder Agreement"), dated April 17, 2012, among Carib Latam Holdings, Inc., as amended on March 27, 2013, June 30, 2013 and November 13, 2013. Under the Stockholder Agreement, Popular was entitled to, among other things, (1) nominate two directors for election to Evertec's board of directors, (2) limited pre-emptive rights and (3) various registration rights with respect to Evertec Common Stock.

## Item 7.01 Regulation FD Disclosure.

On July 1, 2022, the Corporation issued a press release announcing the closing of the Evertec Transaction. A copy of the press release is attached hereto as Exhibit 99.2.

Popular does not intend for this Item 7.01 or Exhibit 99.2 to be treated as filed under the Securities Exchange Act of 1934, as amended, or incorporated by reference into its filings under the Securities Act of 1933, as amended.

## Item 9.01. Financial Statements and Exhibits.

Exhibit 99.1	Second Amended and Restated Master Service Agreement, dated as of July 1, 2022, among Popular, Inc., Banco Popular de		
	erto Rico, and Evertec Group, LLC and its Subsidiaries.		

Exhibit 99.2 Press Release dated July 1, 2022.

Pursuant to Rule 406 of Regulation S-T, the cover page is formatted in Inline XBRL (Inline eXtensible Business Reporting

Language).

104 Cover Page Interactive Data File (embedded within the Inline XBRL document and included in Exhibit 101).

## **Cautionary Note Regarding Forward-Looking Statements**

This Current Report on Form 8-K contains "forward-looking statements" within the meaning of the U.S. Private Securities Litigation Reform Act of 1995, including without limitation those regarding Popular's acquisition of certain assets and assumption of certain liabilities from Evertec and the Evertec Transaction described in this Current Report on Form 8-K and Popular's business, financial condition, results of operations, plans, objectives and future performance. These statements are not guarantees of future performance, are based on management's current expectations and, by their nature, involve risks, uncertainties, estimates and assumptions. Potential factors, some of which are beyond Popular's control, could cause actual results to differ materially from those expressed in, or implied by, such forward-looking statements. Risks and uncertainties include, without limitation, the effect of competitive and economic factors, and our reaction to those factors, the adequacy of the allowance for loan losses, delinquency trends, market risk and the impact of interest rate changes, capital market conditions, capital adequacy and liquidity, the effect of legal and regulatory proceedings, new accounting standards on Popular's financial condition and results of operations, the scope and duration of the COVID-19 pandemic (including the appearance of new strains of the virus), actions taken by governmental authorities in response thereto, and the direct and indirect impact of the pandemic on Popular, our customers, service providers and third parties. Other factors include Popular's ability to successfully transition and integrate the assets acquired as part of the Evertec Transaction; related operations; employees and third party contractors; unexpected costs, including, without limitation, costs due to exposure to any unrecorded liabilities

or issues not identified during the due diligence investigation of the Evertec Transaction or that are not subject to indemnification or reimbursement by Evertec; operational risks that may affect Popular and other risks arising from the acquisition of the acquired assets or by adverse effects on relationships with customers, employees and service providers and business; and other risks arising from the extension of Popular's current commercial agreements with Evertec. All statements contained herein that are not clearly historical in nature, are forward-looking, and the words "anticipate," "believe," "continues," "expect," "estimate," "intend," "project" and similar expressions, and future or conditional verbs such as "will," "would," "should," "could," "might," "can," "may" or similar expressions, are generally intended to identify forward-looking statements.

More information on the risks and important factors that could affect Popular's future results and financial condition is included in our Annual Report on Form 10-K for the year ended December 31, 2021 and in our Form 10-Q for the quarter ended March 31, 2022. Our filings are available on Popular's website (www.popular.com) and on the Securities and Exchange Commission website (www.sec.gov). Popular assumes no obligation to update or revise any forward-looking statements or information which speak as of their respective dates.

## **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, as amended, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

## POPULAR, INC.

(Registrant)

Date: July 1, 2022 By: /s/ José R. Coleman Tió

José R. Coleman Tió Executive Vice President and Chief Legal Officer

## Exhibit 99.1

## EXECUTION VERSION

SECOND AMENDED AND RESTATED

MASTER SERVICE AGREEMENT

AMONG

POPULAR, INC.,

BANCO POPULAR DE PUERTO RICO,

AND

EVERTEC GROUP, LLC AND ITS SUBSIDIARIES

DATED AS OF JULY 1, 2022

## TABLE OF CONTENTS

		Pag
		ARTICLE ONE - GENERAL PROVISIONS
1.1	Definitions	2
1.2	Survival	15
1.3	Relationship between the Parties	15
1.4	Non-Exclusive	15
1.5	Assignment	16
1.6	Binding Effect	18
1.7	No Third Party Beneficiaries	18
1.8	Entire Agreement	18
1.9	Interpretation	18
1.10	Severability	19
1.11	Waiver	19
1.12	Governing Law	19
1.13	Trial by Jury	19
1.14	Dispute Resolution; Arbitration	19
1.15	Cumulative Remedies	20
1.16	Subcontracting the Services	21
1.17	Non-solicitation	22
1.18	Prohibition on Publicity	23
1.19	Business Days and Legal Holidays	23
1.20	Notices	23
1.21	Incorporation	24
1.22	Headings	24
1.23	Language	24
1.24	Counterparts	24
1.25	Representations and Warranties	25
1.26	Specific Performance	25
1.27	Limitation of Actions	25
1.28	Release of Existing Claims	25
1.29	Additional Assurances	26
1.30	No BPPR Guarantee	26
1.31	EVERTEC Change of Control	26
		ARTICLE TWO - THE SERVICES
2.1	Services	27
2.2	Annual Minimums	29
2.3	Service Personnel	32
2.4	Service Level Agreement	32
2.5	Reports and Errors	33
2.6	Specifications for Services	33
2.7	Modifications to Services	34

2.8	Mandatory Enhancements	35
2.9	Non-Circumvention Covenants	37
2.10	Non-Compete Covenants	38
2.11	Equipment and Other Resources	39
2.12	Import/Export Control	40
2.13	Contingency Planning	40
2.14	Procedures Manual	42
	ARTICLE THREE - PAYMENT FOR SERVICES	
3.1	Fees	43
3.2	Terms of Payment	44
3.3	Services Rendered during Legal Holidays	44
3.4	Additional Services	44
3.5	Out-of-pocket and Third Party Expenses	44
3.6	Taxes	45
3.7	Disputed Charges; Requests for Information	45
3.8	Supporting Documentation	45
3.9	No Right to Set-Off	46
3.10	Core API Layer SOW	46
3.11	Phase 2 SOW	48
	ARTICLE FOUR - DISCLAIMER OF WARRANTIES & LIMITED LIABILITY	
4.1	DISCLAIMER OF WARRANTIES	48
4.2	Reliance on COMPANY Provided Data	49
4.3	Force Majeure	49
4.4	Systems and/or Services Not Provided by EVERTEC	50
4.5	LIMITATION OF LIABILITY	50
	ARTICLE FIVE - CONFIDENTIALITY, PRIVACY & DATA PROTECTION	
5.1	Confidential Information	52
5.2	Protection of Confidential Information	53
5.3	Security of Personal Data	54
5.4	Required Disclosures	59
5.5	Remedies	59
5.6	Term of Obligation	60
5.7	Destruction	60
	ARTICLE SIX - SECURITY, BPPR DATA & RECORDS	
6.1	Authorized Persons	60
6.2	Security Measures	61
6.3	Ownership of BPPR Data	63
6.4	Records Retention.	64
6.5	Data Backup	64

	ARTICLE SEVEN - INTELLECTUAL PROPERTY	
7.1	Title	64
7.2	General	65
7.3	Developments	65
7.4	Intellectual Property Infringement	66
	ARTICLE EIGHT - REGULATORY COMPLIANCE, AUDIT & SERVICE REVIEWS	
8.1	Compliance	68
8.2	Legal Requirements	68
8.3	BSA/AML/OFAC	69
8.4	Audit	69
8.5	IT Infrastructure Network Diagram	71
8.6	Data Flow Diagram	71
8.7	Service Center Reviews	71
	ARTICLE NINE - TERM & TERMINATION	
9.1	Term	73
9.2	Termination for Cause	73
9.3	Termination without Cause	74
9.4	Effect upon Termination	75
9.5	Transition Assistance	76
	ARTICLE TEN - INSURANCE AND INDEMNIFICATION	
10.1	Insurance	78
10.2	Indemnity	81
10.3	Indemnification Procedures	82
	ARTICLE ELEVEN - SERVICES FOR CLIENTS	
11.1	Scope of Services	83
11.2	Special Representations Regarding EVERTEC's Performance of Services under Service Agreements with Governmental Agencies	84
11.3	Client Agreements	84

## **EXHIBITS**

- Exhibit A Intentionally Omitted
- Exhibit B Services and Fees
- Exhibit C Service Level Agreement General Terms and Conditions
- Exhibit D Strategic Clients
- Exhibit E Intentionally Omitted
- Exhibit F Outsourcing Policies and Procedures
- Exhibit G Data Protection and Privacy Minimum Data Safeguards for Popular Infrastructure
- Exhibit H Shortfall Payments and Bi-Annual True-Up Calculation

This Second Amended and Restated Master Service Agreement (the "Master Agreement") is made as of July 1, 2022 (the "Effective Date"), among Popular, Inc. ("Popular" or the "COMPANY"), a corporation organized under the laws of the Commonwealth of Puerto Rico, Banco Popular de Puerto Rico ("BPPR" and, together with Popular, the "Popular Parties"), a bank organized under the laws of the Commonwealth of Puerto Rico, and Evertec Group, LLC, a limited liability company organized under the laws of the Commonwealth of Puerto Rico, and its Subsidiaries (hereinafter referred to as "EVERTEC").

#### WITNESSETH

WHEREAS, COMPANY, acting through its Subsidiaries, is a financial services provider based in Puerto Rico with operations in Puerto Rico, the United States and the Caribbean:

WHEREAS, EVERTEC is in the business of, among other things, providing data processing, application processing, check imaging, transmission, telecommunications, credit and debit card transaction processing, and related operational, technical, and consulting services, and such other services as EVERTEC may engage in from time to time;

WHEREAS, Popular and EVERTEC previously entered into a Master Service Agreement on April 1, 2010 (the "April 2010 MSA") and an Amended and Restated Master Service Agreement on September 30, 2010 (the "September 2010 MSA");

WHEREAS, the Popular Parties and EVERTEC have entered into that certain Asset Purchase Agreement, dated as of February 24, 2022 (the "Asset Purchase Agreement"), pursuant to which BPPR is, contemporaneously with entering into this Master Agreement, acquiring a portion of the business of Evertec, Inc., a corporation organized under the laws of the Commonwealth of Puerto Rico and the ultimate parent entity of EVERTEC ("EVERTEC Parent");

WHEREAS, the applicable Popular Parties and EVERTEC are simultaneously on the date hereof amending and restating the Amended and Restated ATH Network Participation Agreement, dated as of September 30, 2010 (as amended and restated as of the date hereof, the "ATH Network Agreement"), the Amended and Restated Independent Sales Organization Sponsorship and Services Agreement, dated as of September 30, 2010 (as amended and restated as of the date hereof, the "ISO Agreement"), the Technology Agreement dated as of September 30, 2010 (as amended and restated as of the date hereof, the "Technology Agreement"), and certain other documents; and

WHEREAS, the Popular Parties and EVERTEC desire to amend and restate the September 2010 MSA in order to, among other things and as further described in the terms of this Master Agreement, provide for certain changes to services provided under and revisions to the terms of the September 2010 MSA.

**NOW THEREFORE**, in consideration of the payments to be made and services to be performed hereunder, upon the terms and subject to the conditions set forth in this Master Agreement and intending to be legally bound, the parties hereto agree to the following terms and conditions:

#### ARTICLE ONE- GENERAL PROVISIONS

- 1.1 <u>Definitions</u>. Capitalized terms not otherwise defined herein will have the meanings set forth in this <u>Section 1.1</u>:
- (1) "Affected Persons" has the meaning set forth in Section 5.3(e)(ii)(1).
- (2) "Affiliate" means, with respect to any Person, any other Person, directly or indirectly, through one or more intermediaries, Controlling, Controlled by, or under common Control with, such Person. Notwithstanding the foregoing, with respect to COMPANY (to the extent that at the time of determination it is engaged in a private equity or similar business), the term "Affiliate" shall not include portfolio companies of COMPANY or its Controlled Affiliates.
- (3) "Annual Minimum" has the meaning set forth in Section 2.2.
- (4) "April 2010 MSA" has the meaning set forth in the Recitals.
- (5) "Asset Acquirer" has the meaning set forth in Section 1.5(c).
- (6) "Asset Purchase Agreement" has the meaning set forth in the Recitals.
- (7) "Assignee Sub" has the meaning set forth in Section 1.5(b).
- (8) "ATH Network Agreement" has the meaning set forth in the Recitals.
- (9) "Audit" has the meaning set forth in Section 8.4(b).
- (10) "Audit Report" has the meaning set forth in Section 8.4(c).
- (11) "<u>Authorized Locations</u>" means the Data Centers and other locations owned or leased by EVERTEC, or COMPANY, BPPR, or one of their respective Affiliates, from time to time, for providing the Services and/or maintaining, processing, or storing BPPR Data under this Master Agreement.
- (12) "Authorized Persons" has the meaning set forth in Section 6.1.
- (13) "Base Date" has the meaning set forth in Section 9.3(d)(i).
- (14) "beneficially owned", "beneficial ownership" and similar phrases have the same meanings as such terms have under Rule 13d-3 (or any successor rule then in effect) under the Exchange Act, except that in calculating the beneficial ownership of any Person, such Person shall be deemed to have beneficial ownership of all securities that such Person has the right to acquire, whether such right is currently exercisable or is exercisable upon the occurrence of a subsequent event. Notwithstanding the foregoing, no Person shall be deemed to beneficially own any securities beneficially owned by another Person who is not an Affiliate of such Person.

- (15) "Best Efforts" means the efforts that a prudent Person desirous of achieving a result would use in similar circumstances to ensure that such result is achieved in the time period expressly contemplated or, in the absence of an expressly contemplated time period, within a commercially reasonable time period.
- (16) "Bi-Annual True-Up" has the meaning set forth in Section 2.2(c).
- (17) "BPPR" has the meaning set forth in the Preamble.
- (18) "BPPR Data" means data, records and information of COMPANY, BPPR, or one of their respective Subsidiaries that is maintained or processed by EVERTEC, including all Personal Data and all data and information derived from or works of any of the foregoing.
- (19) "BPPR Infrastructure" means (i) the servers, storage, connectivity and other infrastructure used in hosting Services or applications owned or licensed from a third party by COMPANY or BPPR; and (ii) any additional IaaS or data center hosting or colocation Services provided for COMPANY or BPPR under this Master Agreement that are dedicated solely to COMPANY and/or BPPR.
- (20) "BPPR Instructions" has the meaning set forth in Section 5.3(f).
- (21) "BPPR Pass Through License" means a license to Intellectual Property provided by a Third Party that is assigned, novated or otherwise transferred to COMPANY or BPPR in connection with the Asset Purchase Agreement.
- (22) "Business Continuity Plan" means the documented processes, preventive arrangements, and measures taken by a party to be able to respond, recover, resume or restore to a predefined level of operation following an Event in order to be able to continue offering its services or to prevent or mitigate the adverse effects of an Event. A Business Continuity Plan must include specific elements, such as incident response, disaster recovery and crisis management.
- (23) "Business Day" will be each day from Monday through Friday, except for Legal Holidays.
- (24) "Client" means any person or entity who establishes or maintains a contractual relationship with COMPANY, BPPR, or one of their respective Subsidiaries for obtaining any service or product offered by EVERTEC and/or COMPANY, BPPR, or one of their respective Subsidiaries.
- (25) "Client Agreements" has the meaning set forth in Section 11.1(b).
- (26) "COI" has the meaning set forth in Section 10.1(b).
- (27) "COMPANY" has the meaning set forth in the Preamble.
- (28) "COMPANY Resources" has the meaning set forth in Section 2.11(d).

- (29) "Competitor" has the meaning set forth in Section 2.9(c).
- (30) "Confidential Information" means any and all confidential and/or proprietary non-public information or material, whether in electronic or hard copy format, that is generated, collected or utilized in COMPANY's, BPPR's (or any of their respective Subsidiary's) or EVERTEC's business or operations. Confidential Information includes:
  - (i) information received from or concerning any Third Party (such as Clients, customers or suppliers);
  - (ii) product or services information concerning any of COMPANY's, BPPR's (or one of their respective Subsidiary's) or EVERTEC's research, or products or services being provided or developed, algorithms, supplier or vendor information, plans, ideas, drawings, specifications, documentation, quality guidelines, and analyses of strengths or weaknesses of each product or service;
  - (iii) trade secrets, source code and object code, other confidential Intellectual Property, technical and technological information, including designs, processes, methods, technical specifications, drawings, prototypes, troubleshooting guidelines, formulas, data, files, performance characteristics, computer software information and related documentation, databases, programs, systems engineering, research and development plans, disks and other storage media, drawings, notes, proposals, reports, photographs, hardware, card files and recordings;
  - (iv) financial information, such as overhead costs, profit margins, banking and financing data, budgeting data and reports, and pricing policies;
  - (v) organizational information, information concerning the utilization of facilities, merger, acquisition and expansion information, equipment utilization information;
  - (vi) marketing and sales information, such as marketing and sales techniques and data, identification and authorizing source, account status and ability to pay, product development and delivery schedules, transportation and logistical requirements, market research and forecasts, strategic planning, and marketing and advertising plans, techniques and budgets;
  - (vii) COMPANY's, BPPR's (or one of their respective Subsidiary's) or EVERTEC's overall strategies, the specific programs and strategies utilized by COMPANY, BPPR (one of their respective Subsidiaries) or EVERTEC; and the success or lack of success of those programs and strategies; and
  - (viii) Personal Data.

For the avoidance of doubt, Confidential Information includes, with respect to each party to this Master Agreement, such party's Data Protection Program (including the Data Safeguards established pursuant to such Data Protection Program), its Business Continuity Plan, proprietary non-public materials with respect to EVERTEC's physical security systems, access control systems, recovery equipment and techniques and all reports and information regarding the Services other than reports and information that are specifically designated as public. For the avoidance of doubt, COMPANY's and BPPR's Confidential Information includes all BPPR Data. EVERTEC's Confidential Information shall not include any (1) Intellectual Property, books and records or other information or materials acquired pursuant to the Asset Purchase Agreement (except to the extent that any books and records are designated as confidential information of EVERTEC pursuant to the Asset Purchase Agreement), or (2) modifications, enhancements or derivative works based on such Intellectual Property in the immediately preceding clause (1).

- (31) "Control," and its correlative meanings, "Controlling," and "Controlled," means the possession, direct or indirect, or the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract or otherwise.
- (32) "Control Acquirer" has the meaning set forth in Section 1.31(a).
- (33) "Core API Completion" has the meaning set forth in Section 3.10(a)(i).
- (34) "Core API Deadline" means the deadline set forth in the Core API Layer SOW.
- (35) "Core API Delay Window" has the meaning set forth in Section 9.3(d)(i).
- (36) "Core API Layer SOW" means the Core API Layer SOW dated, July 1, 2022.
- (37) "Covered Revenue" has the meaning set forth in Section 2.2(d).
- (38) "CPI" means the All Items Consumer Price Index All Urban Consumers, U.S. City Average (1982-84 100), which is published by the U.S. Department of Labor, Bureau of Labor Statistics.
- (39) "CVSS" has the meaning set forth in Exhibit G.
- (40) "Data Cap" has the meaning set forth in Section 4.5(c).
- (41) "Data Centers" means the facilities and data centers where EVERTEC performs Data Processing of BPPR Data, including alternate sites.
- (42) "<u>Data Processing</u>" means any operation or set of operations which is performed on BPPR Data or on sets of BPPR Data, whether or not by automated means, physical or virtual, on premises or in the cloud, such as access, collection, recording, storing, organization, structuring, adaptation or alteration, retrieval, consultation, use, disclosure by transmission, dissemination or otherwise making available, alignment or combination, restriction, erasure or destruction.

- (43) "<u>Data Protection Measures</u>" has the meaning set forth in <u>Section 5.3(d)(ii)</u>.
- (44) "<u>Data Protection Program</u>" means a written program that complies with applicable Legal Requirements, establishing the policies, procedures, controls and Data Safeguards reasonably designed and implemented in accordance with applicable Industry Standards to protect the confidentiality, privacy, security, integrity or availability of BPPR Data, the proper Data Processing of BPPR Data, and the security and reliability of the related infrastructure in order to prevent, detect and respond to any threats or hazards or intentional or accidental unauthorized access, use, disclosure, alteration, loss, destruction, or disposal.
- (45) "<u>Data Safeguards</u>" means reasonable administrative, technical and physical data protection processes, procedures and controls designed and implemented in accordance with Industry Standards.
- (46) "<u>Data Subcontractors</u>" has the meaning set forth in <u>Section 1.16(a)</u>.
- (47) "DFD" has the meaning set forth in Section 8.6.
- (48) "Direction" has the meaning set forth in Section 5.7.
- (49) "Disabling Device" means any program, mechanism, programming device, malware or other computer code: (i) designed to disrupt, disable, harm, or otherwise impede in any manner the operation of any software program or code, or any computer system or network (commonly referred to as "malware", "spyware", "viruses" or "worms"); (ii) that would disable or impair the operation thereof or of any software, computer system or network in any way based on the elapsing of a period of time or the advancement to a particular date or other numeral (referred to as "time bombs", "time locks", or "drop dead" devices); (iii) designed to or that could reasonably be used to permit any Person to gain unauthorized or unintended access any computer system or network (referred to as "trojans", "traps", "access codes" or "trap door" devices); or (iv) designed to or that could reasonably be used to permit any unauthorized Person to track, monitor or otherwise report the operation and use of any software program or any computer system or network by any Person.
- (50) "Disaster Recovery Plan" means the processes, policies, and procedures related to preparing for recovery or continuation of technology infrastructure, systems, and applications, which are vital to an organization after a disaster or other disruption. Disaster recovery focuses on the information or technology systems that support business functions, as opposed to business continuity, which involves planning for keeping all aspects of a business functioning in the midst of disruptive events. A Disaster Recovery Plan is a subset of a Business Continuity Plan.

- (51) "<u>Disaster Recovery Services Addendum</u>" means the Disaster Recovery Services Addendum referenced in <u>Schedule 2</u> to <u>Exhibit B</u>, as amended.
- (52) "Disaster Recovery Strategy" is a strategy for restoring IT infrastructure, data, and systems with a coordinated effort for the recovery of Data Centers, networks, servers, storage, service monitoring, user support, and related software during adverse events (e.g., natural disasters, infrastructure failures, technology failures, unavailability of staff, or cyber-attacks) in order to return such operations to a normalized state with minimum disruption. A Disaster Recovery Strategy is a subset of a Business Continuity Plan.
- (53) "Disclosing Party" has the meaning set forth in Section 5.1(a).
- (54) "Dispute" has the meaning set forth in Section 1.14.
- (55) "Effective Date" has the meaning set forth in the Preamble.
- (56) "Encumbrances" means any direct or indirect encumbrances, lien, pledge, security interest, claim, charges, option, right of first refusal or offer, mortgage, deed of trust, easement, or any other restriction or third party right, including restrictions on the right to vote equity interests.
- (57) "Errors" has the meaning set forth in Section 2.5(b).
- (58) "Event" means any event, occurrence or circumstance that has resulted in, or could reasonably result in any impairment or disruption of a Person's normal business and operations or its ability to provide services.
- (59) "EVERTEC" has the meaning set forth in the Preamble.
- (60) "EVERTEC Change of Control" means, with respect to EVERTEC, any:
  - (i) merger, consolidation or other business combination of EVERTEC (or any Subsidiary or Subsidiaries that alone or together represent all or substantially all of EVERTEC's consolidated business at that time) or any successor or other entity holding all or substantially all the assets of EVERTEC and its Subsidiaries that results in the stockholders of EVERTEC (or such Subsidiary or Subsidiaries) or any successor or other entity holding all or substantially all the assets of EVERTEC and its Subsidiaries or the surviving entity thereof, as applicable, immediately before the consummation of such transaction or a series of related transactions, holding, directly or indirectly, less than 50% of the voting power of EVERTEC (or such Subsidiary or Subsidiaries) or any such successor, other entity or surviving entity, as applicable, immediately following the consummation of such transaction or series of related transactions; provided that this clause (i) shall not be deemed applicable to any merger, consolidation or other business combination, if, as a result of any such merger, consolidation or other business combination, no Person or Group of Persons that had not had "control" of EVERTEC immediately prior to such transaction, as such term is defined under the Bank Holding Company Act of 1956, as amended, shall have obtained such "control";

- (ii) Transfer (other than in the form of a pledge, hypothecation or similar grant of a security interest only and which shall not involve the grant of a proxy or other right with respect to the voting of such equity), in one or a series of related transactions, of equity representing 50% or more of the voting power of EVERTEC (or any Subsidiary or Subsidiaries that alone or together represent all or substantially all of EVERTEC's consolidated business at that time) or any successor or other entity holding all or substantially all the assets of EVERTEC and its Subsidiaries to a Person or Group of Persons;
- (iii) transaction in which a majority of the board of directors or equivalent governing body of EVERTEC (or any successor or other entity holding all or substantially all the assets of EVERTEC and its Subsidiaries) immediately following or as a proximate cause of such transaction is comprised of persons who were not members of the board of directors or equivalent governing body of EVERTEC (or such successor or other entity) immediately prior to such transaction; or
- (iv) sale or other disposition in one or a series of related transactions of all or substantially all of the assets of EVERTEC and its Subsidiaries (or any successor or other entity holding all or substantially all the assets of EVERTEC and its Subsidiaries) to a Person who is not an Affiliate of EVERTEC at such time.
- (61) "EVERTEC Equipment" has the meaning set forth in Section 2.11(a).
- (62) "EVERTEC Parent" has the meaning set forth in the Recitals.
- (63) "EVERTEC Pass Through License" means a license to Intellectual Property owned by a Third Party procured by EVERTEC at the written direction of, or with the written consent of, BPPR, COMPANY or any of their respective Affiliates and used in connection with the provision of the Services.
- (64) "EVERTEC Security Requirements" means reasonable security policies and procedures provided in writing to COMPANY and BPPR with reasonable advance notice by EVERTEC; provided the EVERTEC Security Requirements will not include any of the following:

  (a) representations, warranties or disclaimers; (b) indemnities, cost reimbursements, suspension or termination rights, or other remedies for EVERTEC; (c) rights for EVERTEC to conduct audits or assessments of BPPR or COMPANY networks or Systems; or (d) obligations for BPPR or COMPANY to provide certifications or conduct audits or assessments.

- (65) "Exchange Act" means the Securities Exchange Act of 1934.
- (66) "Exclusive Mandatory Enhancement" has the meaning set forth in Section 2.8(b).
- (67) "Extended Transition Period" has the meaning set forth in Section 9.5(e).
- (68) "FFIEC" means the Federal Financial Institutions Examination Council.
- (69) "Firm" has the meaning set forth in Section 8.7.
- (70) "For Cause Audit" has the meaning set forth in Section 8.4(b).
- (71) "Force Majeure Event" has the meaning set forth in Section 4.3(a).
- (72) "General Liability Cap" has the meaning set forth in Section 4.5(c).
- (73) "Governmental Authority" means the government or any agency or court thereof, of any nation or country (including the European Union, if applicable), state, commonwealth (including the Commonwealth of Puerto Rico), city, municipality or political subdivision thereof, any entity exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to any such government that have regulatory, supervisory, and/or examination authority with respect to COMPANY, BPPR, or one of their respective Subsidiaries and/or of EVERTEC with respect to the matters covered by the Services or their respective operations or financial condition, any quasi-governmental entity or arbitral body, any SRO and any applicable stock exchange.
- (74) "Group of Persons" means a group of Persons that would constitute a "group" as determined pursuant to Section 13(d) of the Exchange Act and the rules and regulations promulgated thereunder.
- (75) "Incident" means the actual or reasonably suspected occurrence of an intentional or accidental unauthorized event that could reasonably pose a threat to the confidentiality, privacy, security, integrity or availability of BPPR Data, including: (1) unauthorized access or disclosure of BPPR Data; (2) unauthorized access or intrusion to information Systems; (3) loss/theft of an electronic or storage device containing BPPR Data; (4) loss/theft of printed materials containing Personal Data; and (5) unauthorized Disabling Devices. For the avoidance of doubt, an Incident will not be deemed to exist solely as a result of EVERTEC's receipt of an automated system alert or a ping or scan of EVERTEC's Systems, unless and until such time as EVERTEC reasonably believes that an Incident has occurred or has reasonable grounds to suspect that an Incident has occurred (provided, that EVERTEC has appropriate systems and procedures for the establishment and use of such automated systems).
- (76) "Incident Response Program" means a written program that complies with applicable Legal Requirements establishing policies, procedures, activities and internal controls reasonably designed and implemented in accordance with applicable Industry Standards to prevent, detect, identify, mitigate, address and notify relevant Persons of Incidents.

- (77) "Incremental BPPR Requirements" has the meaning set forth in Section 8.7(b).
- (78) "Indebtedness" means, with respect to any Person, (a) all indebtedness of such Person, whether or not contingent, for borrowed money, and (b) all obligations of such Person evidenced by notes, bonds, debentures or other similar debt instruments.
- (79) "Indemnified Party" has the meaning set forth in Section 10.1.
- (80) "Indemnifying Party" has the meaning set forth in Section 10.1.
- (81) "Industry Standards" means the procedural and technical rules and controls generally followed by financial and information technology and data processing providers when providing services to banks and their Affiliates, as they may be adopted from time to time, including as may be required under applicable Legal Requirements.
- (82) "Infrastructure-as-a-Service" or "IaaS" means the capability provided to COMPANY or BPPR to provision processing, storage, networks, and other fundamental computing resources where COMPANY or BPPR is able to deploy and run arbitrary software, which can include operating systems and applications. COMPANY and BPPR do not manage or control the underlying cloud infrastructure but have control over operating systems, storage, deployed applications, and possibly limited control of select networking components (e.g., host firewalls).
- (83) "Initial Term" has the meaning set forth in Section 9.1.
- (84) "Intellectual Property" means all world-wide know-how, ideas, concepts, techniques, designs, materials, formulae, computer programs and related documentation, writings, works of authorship, trade secrets, Confidential Information (as defined herein), proprietary information, research, developments, registered and unregistered copyrights, copyrightable materials, derivatives, trademarks, mask works, moral rights, design rights, industrial property rights, discoveries, inventions, rights in and to inventions, improvements, patent rights, patent applications, patents and any other intellectual property rights in any jurisdiction in the world including modifications and other applications using or based upon the resulting technology or required for the use of such technology.
- (85) "Interchange Fee" means a fee paid by a merchant or acquirer for the purpose of compensating an issuer for its involvement in an electronic debit transaction. It is paid via the network according to its compensation rules and subject to federal regulations when a debit account is involved.

- (86) "IP Developments" has the meaning set forth in Section 7.3.
- (87) "IP Indemnified Parties" has the meaning set forth in Section 7.4(a).
- (88) "ISO" has the meaning set forth in Section 10.1(a)(i)(2).
- (89) "ISO Agreement" has the meaning set forth in the Recitals.
- (90) "Legal Holiday" means Saturday, Sunday or any federal and/or local legal holiday in the Commonwealth of Puerto Rico, except as otherwise provided in the Service Level Agreement, a Service Addendum, or as agreed to in writing between the parties.
- (91) "Legal Requirements" means any applicable federal, state, Puerto Rico, local, municipal, foreign, international, multinational, or other administrative order, constitution, law, ordinance, principle of law, regulation, statute, rule, regulation, guidance or treaty, and any policy statement or interpretation that has the force of law, issued by a Governmental Authority, and any standards or guidance issued by the FFIEC (for clarity, with respect to EVERTEC, including the Bank Service Company Act and any other Legal Requirement to which EVERTEC is subject as a result of EVERTEC's provision of services to COMPANY, BPPR or other financial institutions).
- (92) "Loss(es)" means losses, liabilities, claims, damages, fines, expenses, penalties, interest expense, costs and fees and disbursements (including reasonable legal counsel and experts' fees and disbursements), net of any amounts recovered with respect thereto under insurance policies covering any liability thereof if and to the extent applicable in each case, individually or collectively.
- (93) "Mandatory Enhancements" has the meaning set forth in Section 2.8(a).
- (94) "Master Agreement" has the meaning set forth in the Preamble.
- (95) "Material Adverse Effect" has the meaning set forth in Section 9.2(b)(ii).
- (96) "Material Breach" has the meaning set forth in Section 9.2(b)(i).
- (97) "Monthly Benchmark" has the meaning set forth in Section 2.2(a).
- (98) "Notice of Dispute" has the meaning set forth in Section 1.14.
- (99) "Notice of Election" has the meaning set forth in Section 10.3(a).
- (100) "Outsourcing Policy" has the meaning set forth in Section 1.16(a).
- (101) "Penetration Tests" has the meaning set forth in Section 6.2(c)

- (102) "Permitted Assignment" means a Permitted Subsidiary Assignment or a Permitted Third-Party Assignment.
- (103) "Permitted Subsidiary Assignment" means an assignment by EVERTEC of any of its rights, duties or obligations under this Master Agreement to an Assignee Sub in compliance with the provisions of Section 1.5(b).
- (104) "Permitted Third-Party Assignment" means an assignment by EVERTEC of all its rights, duties and obligations under this Master Agreement to an Asset Acquirer in compliance with the provisions of Section 1.5(c).
- (105) "Person" means any individual, corporation, partnership, trust, incorporated or unincorporated association, joint venture, joint stock company, limited liability company, Governmental Authority or other entity of any kind, and will include any assignee and/or successor (by merger or otherwise) of such entity in connection therewith.
- (106) "Personal Data" means any information in any medium or format that alone or in conjunction with other information, identifies, or can reasonably be used to identify or authenticate, directly or indirectly, a natural person or household related to BPPR's past, present or prospective customers, guarantors, claimants, beneficiaries, employees or agents. Personal Data includes, individuals' (1) names, date and place of birth, biometric data, home or business addresses, legal domicile, telephone numbers, email addresses, photographs, handwriting, human race, religion, family status, or criminal record; (2) health information such as genetics, medical history or treatment, or medical insurance data; (3) government-issued identification numbers (such as social security number, passport number, driver's license number); (4) online information such as computer IP address, user names, passwords or PINs, answers to security questions, online identifier, geolocation data, internet or other website activity information; (5) financial information such as financial account numbers, income or other financial information, transaction records, credit report information, applications, loan or claim information; and (6) other information of or relating to an individual that is protected from unauthorized disclosure by Legal Requirements related to data protection and privacy.
- (107) "Personnel" means, with respect to any Person, such Person's Representatives and Subcontractors.
- (108) "Phase 2 SOW" has the meaning set forth in the Asset Purchase Agreement.
- (109) "Popular" has the meaning set forth in the Preamble.
- (110) "Popular Parties" has the meaning set forth in the Preamble.
- (111) "Popular Services and Products" has the meaning set forth in Section 11.1(b).
- (112) "Procedures Manual" has the meaning set forth in Section 2.14(a).

- (113) "Proprietary Application Information" has the meaning set forth in Section 2.14.
- (114) "Reasonable and Customary Costs" has the meaning set forth in Section 4.5(f).
- (115) "Receiving Party" has the meaning set forth in Section 5.1(a).
- (116) "Recovery Point Objectives" or "RPOs" means the point in time to which data used by an activity is restored to enable the resumption of business functions upon occurrence of an Event. The RPO is expressed backwards in time from the point of disruption and can be specified in increments of time (e.g., minutes, hours, or days).
- (117) "Recovery Time Objectives" or "RTOs" means the maximum amount of time that a system resource can remain unavailable before there is an unacceptable impact on other system resources and business processes.
- (118) "Region" means Puerto Rico, the U.S. Virgin Islands and the British Virgin Islands.
- (119) "Release Event" has the meaning set forth in the Technology Agreement.
- (120) "Released Parties" has the meaning set forth in Section 1.28.
- (121) "Releasing Parties" has the meaning set forth in Section 1.28.
- (122) "Reports" has the meaning set forth in Section 2.5(a).
- (123) "Representative" means with respect to a particular Person, any director, officer, partner, member, employee, agent, consultant, advisor, or other representative of such Person, including legal counsel, accountants, and financial advisors.
- (124) "Requested Enhancement" has the meaning set forth in Section 2.7(b).
- (125) "Resolution Forum" has the meaning set forth in Section 1.14.
- (126) "Role-Based Access Controls" means documented controls that require that each individual providing Services be assigned one or more roles, and that each role be granted and limited to specific System and data access and administration privileges permitted for individuals in those roles.
- (127) "September 2010 MSA" has the meaning set forth in the Recitals.
- (128) "Service Addendum" has the meaning set forth in Section 2.1(d).
- (129) "Service Center Review" has the meaning set forth in Section 8.7(a).
- (130) "Service Level Agreement" has the meaning set forth in Section 2.4(a).

- (131) "Service Levels" has the meaning set forth in Section 2.4(a).
- (132) "Services" means the services and functions set forth on Exhibit B.
- (133) "Shortfall Payment" has the meaning set forth in Section 2.2(b).
- (134) "SOC 1" has the meaning set forth in Section 8.7(a).
- (135) "SOC 2" has the meaning set forth in Section 8.7(a).
- (136) "Solvent" with regard to any Person, means that (i) the sum of the assets of such Person, both at a fair valuation and at a present fair salable value, exceeds its liabilities, including contingent, subordinated, unmatured, unliquidated, and disputed liabilities; (ii) such Person has sufficient capital with which to conduct its business; and (iii) such Person has not incurred debts beyond its ability to pay such debts as they mature. For purposes of this definition, "debt" means any liability on a claim, and "claim" means (x) a right to payment, whether or not such right is reduced to judgment, liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, secured or unsecured or (y) a right to an equitable remedy for breach of performance to the extent such breach gives rise to a payment, whether or not such right to an equitable remedy is reduced to judgment, fixed, contingent, matured, unmatured, disputed, undisputed, secured or unsecured. With respect to any such contingent liabilities, such liabilities shall be computed at the amount which, in light of all the facts and circumstances existing at the time, represents the amount which can reasonably be expected to become an actual or matured liability.
- (137) "SOW" has the meaning set forth in Section 2.1(f).
- (138) "Specifications" has the meaning set forth in Section 2.6(a).
- (139) "SRO" means any domestic or foreign securities, broker-dealer, investment adviser or insurance industry self-regulatory organization.
- (140) "SSAE 18" has the meaning set forth in Section 8.7(a).
- (141) "Subcontractor" has the meaning set forth in Section 1.16(a).
- (142) "Subsidiary" means, with respect to any Person, any corporation, association, partnership, limited liability company or other business entity of which 50% or more of the total voting power or equity interests (including partnership interests) entitled (without regard to the occurrence of any contingency) to vote in the election of directors, managers, representatives or trustees thereof is at the time owned or Controlled, directly or indirectly, by (a) such Person, (b) such Person and one or more Subsidiaries of such Person, or (c) one or more Subsidiaries of such Person.

- (143) "Supplemental Mandatory Enhancements" has the meaning set forth in Section 2.8(d).
- (144) "System" means any hardware, software, technology, applications, network infrastructure and/or combination thereof used to provide or use the Services under this Master Agreement.
- (145) "Technology Agreement" has the meaning set forth in the Recitals.
- (146) "Term" means the Initial Term and any subsequent renewal term that the parties may agree to in an amendment to this Master Agreement.
- (147) "Termination Notice Date" has the meaning set forth in Section 9.5(a).
- (148) "Third Party" means any Person that is not a party to this Master Agreement and is not an Affiliate of any party to this Master Agreement.
- (149) "Transfer" means any direct or indirect sale, assignment, transfer, conveyance, gift, bequest by will or under intestacy laws, pledge, hypothecation or other Encumbrance, or any other disposition, of the stated security (or any interest therein or right thereto, including the issuance of any total return swap or other derivative whose economic value is primarily based upon the value of the stated security) or of all or part of the voting power (other than the granting of a revocable proxy) associated with the stated security (or any interest therein) whatsoever, or any other transfer of beneficial ownership of the stated security, with or without consideration and whether voluntarily or involuntarily (including by operation of law).
- (150) "Transition Assistance" has the meaning set forth in Section 9.5(d).
- (151) "Transition Period" has the meaning set forth in Section 9.5(c).
- (152) "True-Up Calculation Period" has the meaning set forth in Section 2.2(c).
- (153) "Virtusa Work Order" has the meaning set forth in Section 3.10(a)(iii).
- 1.2 <u>Survival</u>. Provisions relating to limitation of liability, indemnity, payment, transition services and other provisions that by their nature are intended to survive shall survive the termination or expiration of this Master Agreement.
- 1.3 <u>Relationship between the Parties</u>. The parties hereto are independent contractors, and this Master Agreement will not be construed in any way as establishing a partnership, joint venture, or express or implied agency relationship between or among them.
- 1.4 <u>Non-Exclusive</u>. Except as otherwise set forth herein or agreed to by the parties in writing, the parties hereto acknowledge that this Master Agreement is not exclusive and nothing contained herein will be construed to create an exclusive relationship between EVERTEC, on the one hand, and COMPANY, BPPR, or any of their respective Subsidiaries, on the other. As such, EVERTEC will not be limited in entering into similar agreements with other Persons to provide the same or similar services.

### 1.5 Assignment.

- (a) Other than a Permitted Assignment pursuant to Section 1.5(b) or (c), this Master Agreement may not be assigned by any party without the prior written consent of the other parties; provided, that any party may assign its rights, duties and obligations under this Master Agreement to its financing sources solely in connection with the granting of a security interest and the enforcement of all rights and remedies that the assigning party has against the other party under this Master Agreement, subject to the claims, defenses and rights, including rights of set off, that such other party may have against the assigning party.
- (b) Assignment to Subsidiaries. EVERTEC may assign any of its rights, duties or obligations to a direct or indirect wholly-owned Subsidiary of EVERTEC (an "Assignee Sub") if (i) such Assignee Sub is identified by EVERTEC to COMPANY and BPPR at least 20 Business Days prior to the consummation of the proposed assignment; (ii) (A) such proposed assignment is legally required in order for EVERTEC to provide to COMPANY, BPPR or their respective Subsidiaries, in the country, state, territory or other jurisdiction in which the Assignee Sub is organized, the specific services to be performed pursuant to the assignment of this Master Agreement, and only (x) to the extent of such legal requirement and (y) if EVERTEC provides a written opinion of qualified counsel that opines that such legal requirement is applicable and is based upon reasonable assumptions with respect to such legal requirement or (B) COMPANY has provided its prior written consent, such consent not to be unreasonably delayed, withheld or conditioned; (iii) such Assignee Sub will be Solvent immediately after and giving effect to such proposed assignment and COMPANY is reasonably satisfied with the terms and conditions of the proposed assignment; (iv) COMPANY is a third-party beneficiary to the assignment agreement, which is in form and substance that is reasonably satisfactory to COMPANY, and which provides that the Assignee Sub's rights under the assignment agreement will be terminated if the Assignee Sub ceases to be a wholly-owned Subsidiary, directly or indirectly, of EVERTEC and (v) EVERTEC remains fully liable with respect to the performance of all its obligations under this Master Agreement and EVERTEC guarantees the performance of all of the obligations of EVERTEC to COMPANY assumed by Assignee Sub under this Master Agreement, which guarantee provides that, for the avoidance of doubt, after any termination of the proposed assignment, EVERTEC shall continue to be obligated with respect to any obligation undertaken by Assignee Sub prior to such term
- (c) Assignment to Third Parties. EVERTEC may assign all of its rights, duties and obligations (or those rights, duties and obligations arising after the effectiveness of the assignment) in a transaction with a third-party assignee (an "Asset Acquirer") if (i) such Asset Acquirer is identified by EVERTEC to COMPANY and BPPR at least 30 Business Days prior to the consummation of the proposed assignment; (ii) such Asset Acquirer (A) acquires at least 90% of the consolidated gross assets (excluding cash) of EVERTEC and its Subsidiaries and (B) assumes at least 90% of the consolidated gross liabilities (excluding Indebtedness) of EVERTEC and its Subsidiaries (including the assignment and assumption of all commercial agreements between EVERTEC or any of its Subsidiaries, on the one hand, and COMPANY, BPPR or any of their respective Subsidiaries, on the other hand) through one legal entity;

(iii) neither the Asset Acquirer nor any of its Affiliates is engaged, directly or indirectly, in the banking, securities, insurance or lending business, from which they derive aggregate annual revenues from Puerto Rico in excess of \$50 million unless none of them has a physical presence in Puerto Rico that is used to conduct any such business; (iv) the Asset Acquirer will be Solvent immediately after and giving effect to such proposed assignment; and (v) EVERTEC reasonably believes that the Asset Acquirer, after completion of the proposed purchase and assumption transaction, will be capable of providing the Services at the level of service required under this Master Agreement.

- (d) Cooperation. EVERTEC shall use its Best Efforts to cooperate with COMPANY and BPPR in evaluating whether any proposed assignment pursuant to this Section 1.5 would be in compliance with the requirements of the provisions contained in this Section 1.5, including the ability of Assignee Sub or Asset Acquirer, as applicable, to comply with the terms of this Master Agreement, including, in each case, by providing any non-confidential information regarding the purposes and plans in connection with such proposed assignment other than information that would create any potential liability under applicable Legal Requirements, violate any confidentiality obligation, or that reasonably would be expected to result in the waiver of any attorney-client privilege.
- (e) *Notice of Objection*. COMPANY or BPPR shall notify EVERTEC in writing within 15 Business Days following receipt of EVERTEC's notice of the proposed assignment of any objection to any proposed assignment to an Assignee Sub or Asset Acquirer unless EVERTEC has failed to satisfy its obligations pursuant to Section 1.5(d) and COMPANY or BPPR asserts such failure prior to the expiration of the 15 Business Day objection period, in which case such 15 Business Day period shall be tolled until EVERTEC satisfies its obligations pursuant to Section 1.5(d). If BPPR or COMPANY fails to timely object to such proposed assignment (taking into account any tolling of the 15 Business Day objection period), it shall be deemed to have consented to such proposed assignment.
- (f) *Invalidity of Impermissible Assignments*. Any attempted or purported assignment in violation of this <u>Section 1.5</u> hereof shall be null and void and the assignee's rights assigned pursuant to any assignment made in compliance with this <u>Section 1.5</u> will terminate in the event and to the extent of the termination of this Master Agreement.
- (g) BPPR Asset Transfer. If BPPR or any of its Subsidiaries transfers, in a single transaction or series of related transactions (including in a merger, business combination, reorganization, or similar transaction (including by operation of law)) 50% or more of BPPR's consolidated assets in the Region as of the time of transfer, or assets that generate 50% or more of BPPR's consolidated revenues in the Region for the full twelve-month period ending at the time of transfer, to any Person, then BPPR shall assign (or cause its applicable Subsidiaries to assign) to such Person its rights, duties and obligations under this Master Agreement in respect of the Services provided to the applicable transferors and shall cause such Person to assume its liabilities under this Master Agreement in respect of the Services provided to the applicable transferor. For the avoidance of doubt, no such assignment shall relieve COMPANY, BPPR or any of their respective Subsidiaries of their obligations under this Master Agreement to the extent COMPANY, BPPR or any of their respective Subsidiaries survive any such sale of assets, merger, business combination, reorganization, or similar transaction.

- 1.6 <u>Binding Effect</u>. This Master Agreement and all the provisions hereof will be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns. The parties hereto intend that this Master Agreement will not benefit or create any right or cause of action in, or on behalf of, any Person other than the parties hereto.
- 1.7 No Third Party Beneficiaries. Each party intends that this Master Agreement will not benefit, or create any right or cause of action in or on behalf of, any Person other than COMPANY, BPPR, and their respective Subsidiaries, and EVERTEC.
- 1.8 Entire Agreement. This Master Agreement, the Asset Purchase Agreement, the Technology Agreement and the other Ancillary Agreements (as such term is defined in the Asset Purchase Agreement) contain the entire understanding of all agreements between the parties hereto with respect to the subject matter hereof and supersede any prior agreement or understanding, oral or written, pertaining to any such matters, which other agreements or understandings will be of no force or effect for any purpose. This Master Agreement may not be amended or supplemented in any manner except by mutual agreement of the parties and as set forth in a writing signed by the parties hereto or their respective permitted successors in interest.

## 1.9 Interpretation.

- (a) The general terms and conditions of this Master Agreement and the Exhibits, Addenda, and Schedules made a part hereof from time to time will be interpreted as a single document. However, in the event of a conflict between the general terms and conditions of this Master Agreement and the terms of any Exhibit, Addendum or Schedule hereto, then the terms of the Exhibits, Addenda, and Schedules will prevail and control the interpretation of this Master Agreement with respect to the subject matter of the applicable Exhibits, Addenda, and/or Schedules; provided, however, that the specific provisions of each SOW and other written instructions listed in Schedule 3 to Exhibit B relating to the term and termination of such SOW and other written instructions shall be governed in accordance with Schedule 3 to Exhibit B. Furthermore, in the event of any conflict or inconsistency between this Master Agreement and any other document referenced herein (including the EVERTEC Security Requirements), regarding the interpretation of the terms of this Master Agreement, this Master Agreement together with its Exhibits, Addenda, and Schedules will prevail and control.
- (b) For purposes of this Master Agreement: (i) "include", "includes" or "including" shall be deemed to be followed by "without limitation"; (ii) "hereof", "herein", "hereby", "hereto" and "hereunder" shall refer to this Master Agreement as a whole and not to any particular provision of this Master Agreement; (iii) "extent" in the phrase "to the extent" shall mean the degree to which a subject or other item extends and shall not simply mean "if"; (iv) "dollars" and "\$" shall mean United States dollars; (v) the singular includes the plural and vice versa; (vi) reference to a gender includes the other gender; (vii) "any" shall mean "any and all"; (viii) "or" is used in the inclusive sense of "and/or"; (ix) reference to any agreement, document or instrument means such agreement, document or instrument as amended, supplemented, modified and in effect from time to time in accordance with its terms; and (x) reference to any statute or regulation means such statute or regulation as amended from time to time and includes any successor legislation thereto and any regulations promulgated thereunder.

- 1.10 Severability. Any term or provision of this Master Agreement which is held by a court of competent jurisdiction or as the result of an arbitration proceeding pursuant to Section 1.14, to be invalid or unenforceable in any jurisdiction shall, as to that jurisdiction, be ineffective to the extent of such invalidity or unenforceability without rendering invalid or unenforceable the remaining terms and provisions of this Master Agreement or affecting the validity or enforceability of any of the terms or provisions of this Master Agreement in any other jurisdiction; provided that this Section 1.10 shall not apply if such severability would materially and adversely affect the economic benefits of the parties resulting from this Master Agreement. Upon a determination that any term or other provision is invalid, illegal, or unenforceable, the parties hereto shall negotiate in good faith to modify this Master Agreement so as to effect the original intent of the parties as closely as possible in a mutually acceptable manner in order that this Master Agreement be consummated as originally contemplated to the greatest extent possible.
- 1.11 <u>Waiver</u>. The tardiness or failure by any of the parties hereto in exercising any right or privilege pursuant to this Master Agreement will not operate as a waiver thereof, nor will the exercise of any right by any party serve as an obstacle to the exercise of any other rights, powers or privileges, or any portion thereof. The waiver of any breach of any provision under this Master Agreement by any party will not be deemed to be a waiver of any preceding or subsequent breach under this Master Agreement. No such waiver will be effective unless in writing.
- 1.12 Governing Law. This Master Agreement will be governed by and interpreted in accordance with the laws of the Commonwealth of Puerto Rico applicable to contracts made and entirely to be performed therein, without regard to the place of performance hereunder.
- 1.13 <u>Trial by Jury</u>. The parties hereby mutually agree that no party, nor any permitted assignee, successor, heir or Representative thereof, will seek a jury trial in any lawsuit, proceeding, counterclaim, or any other litigation procedure based upon or arising out of this Master Agreement, or any related agreement or instrument among the parties. None of the parties will seek to consolidate any such action, in which a jury trial has been waived, with any other action in which a jury trial has not been waived. The provisions of this <u>Section 1.13</u> have been fully negotiated by the parties. The waiver contained herein is irrevocable, constitutes a knowing and voluntary waiver, and will be subject to no exceptions.
- 1.14 <u>Dispute Resolution</u>; <u>Arbitration</u>. Except as otherwise provided in writing with respect to EVERTEC's failure to achieve or maintain a Service Level (as such term is defined below), or except as may otherwise be agreed to in writing among the parties, any dispute, controversy or claim between EVERTEC, on the one hand, and any of the Popular Parties or their respective Subsidiaries, on the other, or against any Representative of one of the parties, related to this Master Agreement, and any dispute or claim related to the relationship or duties contemplated hereunder, including the validity of this <u>Section 1.14</u> (a "<u>Dispute</u>") will be resolved in accordance with this <u>Section 1.14</u>. Each party will give written notice (a "<u>Notice of Dispute</u>") to the others of any Dispute claimed by it within thirty (30) days of learning of the cause of such a Dispute. The Notice of Dispute will include a reasonable description of the basis of the Dispute, including, (i) the specific charge or charges being disputed, (ii) if available and/or applicable, the supporting documentation that is reasonably required for verification of the charge or charges, and (iii) any amounts being withheld. Following delivery of a Notice of Dispute, a Representative of each party will meet and will attempt in good faith to resolve the Dispute. Any Dispute that remains

unresolved for more than twenty (20) days after the receipt of a Notice of Dispute shall be referred to designated representatives of the parties hereto who shall negotiate in good faith to resolve such dispute (the "Resolution Forum"). If a Dispute is not resolved in the Resolution Forum, the Dispute shall be submitted to the consideration of a representative from the senior management of EVERTEC who shall be identified in a written notice delivered to the Popular Parties from time to time, the Chief Operating Officer, the Chief Financial Officer or the Chief Information Officer of BPPR and the Chief Operating Officer or the Chief Financial Officer of COMPANY. Any Disputes that may remain unresolved for more than ninety (90) days following the receipt of a Notice of Dispute may be referred to binding arbitration at the request of any party upon written notice to the other. Such arbitration proceeding will be administered by the American Arbitration Association in accordance with the then-current Commercial Arbitration Rules and will be aired in the Commonwealth of Puerto Rico. The arbitration will be governed by the United States Arbitration Act, 9 U.S.C. §§ 1-16 to the exclusion of any provision of state law inconsistent therewith or which would produce a different result. A panel of three neutral arbitrators will determine the Dispute of the parties and render a final award in accordance with the applicable substantive law. If the Dispute is between EVERTEC, on the one hand, and one or both of the Popular Parties and their respective Subsidiaries, on the other hand, each of EVERTEC and the Popular Parties shall select one neutral arbitrator and, unless those parties agree on a third neutral arbitrator, such two arbitrators shall select the third arbitrator (subject to such limitations, if any, mutually agreed by those parties). If the Dispute is between the Popular Parties, then each of the Popular Parties shall select one neutral arbitrator and, unless those parties agree on a third neutral arbitrator, such two arbitrators shall select the third arbitrator (subject to such limitations, if any, mutually agreed by those parties). Strict confidentiality will govern the arbitration proceedings, including all information submitted to the arbitrators and the decision or award entered by the arbitrators. Any court having jurisdiction may enter judgment upon the award rendered by the arbitrators. The terms hereof will not limit any obligation of a party to defend, indemnify or hold harmless another party against court proceedings or other Losses. The procedures specified in this Section 1.14 will be the sole and exclusive procedures for the resolution of Disputes among the parties arising out of or relating to this Master Agreement; provided, however, that a party may request temporary remedies in a court of law to maintain the status quo or to protect goods or property until the arbitration has initiated and the selected arbitrators have had the opportunity to resolve the request for temporary relief. Each party is required to continue to perform its obligations under this Master Agreement pending final resolution of any Dispute arising out of or relating to this Master Agreement, unless to do so would be impossible or impracticable under the circumstances.

1.15 <u>Cumulative Remedies</u>. Except as otherwise expressly provided, all rights and remedies provided for in this Master Agreement will be cumulative and in addition to and not in lieu of any other rights and remedies available to any party at law, in equity or otherwise and will not serve to exclude the exercise of any right or remedy provided by law.

#### 1.16 Subcontracting the Services.

- (a) EVERTEC shall provide COMPANY and BPPR with the written notice required below in this Section 1.16(a) prior to outsourcing, delegating or subcontracting the performance of any part of the Services to a Third Party vendor, including any Third Party computer programmers. software developers or Data Center, collocation, hosting or cloud service providers (each, a "Subcontractor"), to the extent such Subcontractor would store, process, transport or otherwise have access to BPPR Data or other Confidential Information or Intellectual Property of COMPANY or BPPR or their respective Subsidiaries (a Subcontractor with such access, including those who perform Data Processing of Personal Data or provide Data Center services, a "Data Subcontractor"), except to the extent such Data Subcontractor is in use by EVERTEC as of the Effective Data. The Data Subcontractors used by EVERTEC as of the Effective Date are listed in Schedule 1.16. For the avoidance of doubt, such information related to EVERTEC's Data Centers, Data Subcontractors and Subcontractors shall be considered EVERTEC Confidential Information and EVERTEC may otherwise engage Subcontractors who are not Data Subcontractors without the obligation to inform COMPANY or BPPR, as applicable, subject to the terms of this Master Agreement and to applicable Legal Requirements. EVERTEC agrees that, prior to assigning a Data Subcontractor under this Master Agreement: EVERTEC must (i) provide COMPANY and BPPR with at least 180 days' written notice for Data Subcontractors that will perform Services at a new Data Center and at least 30 days' written notice for all other Data Subcontractors (or such shorter period as may be necessary to assign a Data Subcontractor in order to remedy an Incident or other similar emergency event); (ii) within 10 days of a request from COMPANY or BPPR, provide COMPANY or BPPR the information described in Section 1.16(c) to evaluate the proposed Data Subcontractor during the applicable notice period provided above; and (iii) comply with the outsourcing policies and procedures set forth in Exhibit F (the "Outsourcing Policy") and applicable Legal Requirements relating to the outsourcing of services. Without limiting the foregoing, prior to the commencement of any work by any Data Subcontractor, EVERTEC must: (i) perform the necessary due diligence to manage related vendor risks and ensure it only assigns Subcontractors that (A) are qualified industry vendors that EVERTEC reasonably believes to be competent, (B) meet or exceed any requirements as may be required under applicable Legal Requirements and (C) have the requisite skills to perform the subcontracted obligations; (ii) enter into written agreements that bind the Data Subcontractors to data security and confidentiality obligations no less stringent than to those herein; and (iii) monitor its Subcontractor's performance to ensure it is consistent and compliant with all the requirements of the Outsourcing Policy.
- (b) EVERTEC shall monitor its Data Subcontractors' performance to ensure such performance is consistent and compliant with all the applicable requirements of this Master Agreement and the Outsourcing Policy. Notwithstanding the foregoing, EVERTEC shall remain exclusively and fully responsible and liable towards COMPANY, BPPR, and their respective Subsidiaries for the due performance of such Services, compliance with this Master Agreement, and any gross negligence, fraud or willful misconduct by such Subcontractors (including any Fourth Party) and there shall be no direct relationship whatsoever between COMPANY, BPPR, or their respective Subsidiaries, on the one hand, and such Subcontractors or Fourth Parties, on the other, solely by virtue of this Master Agreement.
- (c) <u>Subject to Section 1.16(f)</u>, at COMPANY's or BPPR's request, upon reasonable advance written notice, EVERTEC will provide COMPANY or BPPR with copies of any documents in EVERTEC's possession that are related to EVERTEC's due diligence and risk analysis of its Data Subcontractors; <u>provided</u>, that (i) such requests shall be conducted in such a manner as not to interfere with the normal operations of EVERTEC's business, (ii) disclosure of such information by EVERTEC shall be subject to, and limited by, applicable Legal Requirements, (iii) EVERTEC may redact from such copies information related solely to EVERTEC or customers other than COMPANY, BPPR or their respective Subsidiaries and (iv) EVERTEC shall not be obligated to provide any information if doing so would reasonably be expected to result in the waiver of any attorney-client or other legal privilege.

- (d) EVERTEC represents that its vendor risk management policies and procedures include commercially reasonable audit rights with respect to its Data Subcontractors.
- (e) If a Data Subcontractor uses or permits another party (a "Fourth Party") to store, transport, process otherwise access Personal Data or Confidential Information or Intellectual Property of COMPANY, BPPR, or their respective Subsidiaries, EVERTEC will comply with this <u>Section 1.16</u> with respect to such Fourth Party, subject to <u>Section 1.16(f)</u>.
- (f) If there are terms in EVERTEC's contracts existing as of the Effective Date with its Data Subcontractors that limit EVERTEC's rights to provide the information requested by COMPANY or BPPR under Section 1.16(c) or (e), EVERTEC will use Best Efforts to obtain the right to provide the requested information but will not be in breach of this Master Agreement to the extent it is unable to do so as a result of such contract terms. If a Data Subcontractor proposes to limit EVERTEC's rights in this regard in any new contract or contract renewal after the Effective Date, EVERTEC will notify COMPANY and BPPR reasonably in advance of the contract execution date of the issue and use Best Efforts to address any concerns relating to the issue that COMPANY or BPPR may raise with EVERTEC.

### 1.17 Non-solicitation.

- (a) Each party agrees that, during the period commencing on the execution of this Master Agreement and ending upon the one (1) year anniversary of the end of the Transition Period, without the prior written consent of the other party, such party shall not, and it shall cause its Subsidiaries not to, directly or indirectly, (i) induce or encourage any employee of the other party to terminate his or her employment with the other party or (ii) solicit for employment or any similar arrangement any employee of the other party; provided that such party and its Subsidiaries shall not be restricted from (A) accepting referrals for employment made by a placement agency or employment service so long as such placement agency or employment service has not targeted employees of the other party, (B) making any general advertisement not targeted at employees of the other party appearing in a newspaper, magazine, Internet sites or trade publication, (C) soliciting any person who has not been an employee of the other party for at least one hundred eighty (180) days prior to being solicited or hired by such party or its Subsidiaries and whom neither the other party nor any of its Subsidiaries, subject to clauses (A) and (B) of this proviso, have solicited over such 180-day period, or (D) soliciting any person who the other party or its Subsidiaries are permitted to solicit under the Asset Purchase Agreement in accordance with the terms thereof.
- (b) Each of COMPANY and BPPR agrees that, during the period commencing on the execution of this Master Agreement and ending upon the one (1) year anniversary of the end of the Transition Period, without the prior written consent of EVERTEC, such party shall not, and it shall cause its Subsidiaries not to, directly or indirectly, hire or assist any other Person in hiring any Restricted Employee; <u>provided</u> that COMPANY, BPPR and their Subsidiaries shall not

be restricted from hiring any person who has not been an employee of the other party for at least one hundred eighty (180) days prior to being solicited or hired by such party or its Subsidiaries and whom neither the other party nor any of its Subsidiaries, subject to clauses (A) and (B) of the proviso in clause (a) above, have solicited over such 180-day period. For purposes of this Section 1.17(b), "Restricted Employee" means any employee of the following EVERTEC groups or divisions: Business Solutions division, Payment Services division, Information Technology and Quality and Change Management groups within EVERTEC's Information Technology and Operations division, and Information Security division (including any functional successor thereto).

- (c) EVERTEC agrees that, during the period commencing on the execution of this Master Agreement and ending upon the one (1) year anniversary of the expiration or termination of this Master Agreement, without the prior written consent of COMPANY, EVERTEC shall not, and it shall cause its Affiliates not to, directly or indirectly, hire or assist any other Person in hiring any Restricted Employee; provided that EVERTEC shall not be restricted from hiring any person who has not been an employee COMPANY, BPPR or any of its Affiliates for at least one hundred eighty (180) days prior to being solicited or hired by EVERTEC or its Affiliate and whom neither EVERTEC nor any of its Affiliates, subject to clauses (A) and (B) of the proviso in clause (a) above, have solicited over such 180-day period. For purposes of this Section 1.17(c), "Restricted Employee" means (A) any employee of COMPANY or BPPR's Digital Banking and Payment Division, Information Technology Division and Cybersecurity Division (including any functional successor thereto); and (B) any former employee of EVERTEC that is hired by COMPANY, BPPR or their Subsidiaries in connection with solicitations permitted under the Asset Purchase Agreement.
- 1.18 <u>Prohibition on Publicity</u>. Neither EVERTEC nor the Popular Parties (or their respective Subsidiaries) may advertise or promote using the name or description of the other parties or party, respectively, without in each instance the express written consent of EVERTEC or the Popular Parties, as applicable.
- 1.19 <u>Business Days and Legal Holidays</u>. Except as expressly agreed to otherwise for a particular Service, in the event that any action, payment, or time period, under this Master Agreement is scheduled to occur on or becomes due on a day that is a Legal Holiday, such action, payment or time period will be performed and/or expire, as applicable, on the next Business Day immediately following the Legal Holiday.
- 1.20 Notices. Except as expressly agreed to otherwise for a particular Service, all notices, requests, demands, consents and other communications given or required to be given under this Master Agreement and under the related documents will be in writing and may be given in any manner described below to the address or number set out below, and shall be deemed effective as follows: (a) if delivered personally to the Person designated below, (b) when the same is actually received, if sent by express overnight courier service, with charges prepaid and return receipt requested, or (c) if sent by email, if notice of nondelivery is not received, (i) at or prior to 5:00 pm local time of the recipient on a Business Day, on that Business Day or (ii) later than 5:00 pm local time of the recipient, on the next succeeding Business Day; provided notices of default, non-renewal or termination must be delivered personally or by courier as provided in Sections 1.20(a) or (b); provided further that all such notices, requests, demands, consents and other communications shall be accompanied by an email notification.

#### To EVERTEC:

Evertec, Inc.

Cupey Center Building, Road 176, Kilometer 1.3

San Juan, Puerto Rico 00926 Facsimile: 787-766-4585

Email: luis.rodriguez@evertecinc.com

Attention: Luis A. Rodríguez, General Counsel and Executive Vice President of

Corporate Development

To the Popular Parties:

Popular, Inc. PO Box 362708

San Juan, Puerto Rico 00936-2708 Email: legaldivision@popular.com

Attention: Jose R Coleman Tio, Executive Vice President and Chief Legal Officer

Any party hereto may change its address or contact details from time to time by giving notice to that effect as provided in this Master Agreement.

- 1.21 <u>Incorporation</u>. All Exhibits, Addenda, Schedules, certificates, agreements and other documents attached hereto and to which reference is made herein are incorporated by reference as if fully set forth herein.
- 1.22 <u>Headings</u>. The headings used in this Master Agreement are inserted for purposes of convenience of reference only and will not limit or define the meaning of any provisions of this Master Agreement.
- 1.23 Language. This Master Agreement has been executed in the English language and all Exhibits, Addenda, and Schedules to this Master Agreement shall be in English, except that any Addenda in effect prior to the date hereof may be in Spanish.
- 1.24 Counterparts. This Master Agreement may be executed in counterparts, each of which when so executed shall be deemed to be an original and all of which when taken together shall constitute one and the same instrument. Any signature to this Master Agreement may be delivered by facsimile, electronic mail (including pdf) or any electronic signature complying with the U.S. federal ESIGN Act of 2000 or the New York Electronic Signature and Records Act or other transmission method and any counterpart so delivered shall be deemed to have been duly and validly delivered and be valid and effective for all purposes to the fullest extent permitted by applicable Legal Requirements. For the avoidance of doubt, the foregoing also applies to any amendment, extension or renewal of this Master Agreement and any Addenda, Schedule, SOW or other document to be executed by the parties pursuant hereto or thereto.

1.25 Representations and Warranties. EVERTEC, COMPANY and BPPR each represent and warrant that (i) it has the power and authority to grant the rights and perform the obligations to which it commits herein; (ii) the execution of this Master Agreement by the person representing it will be sufficient to render this Master Agreement binding upon it; (iii) neither its performance hereunder nor the exercise by the other parties of rights granted by the warranting party hereunder will violate any applicable Legal Requirements, or the legal rights of any Third Parties, or the terms of any other agreement to which the warranting party is or becomes a party; and (iv) it has and will maintain an adequate system of internal controls and procedures for financial reporting. Each party is separately responsible for ensuring that its performance and grant of rights do not constitute any such violation during the Term. Each of the foregoing representations and warranties and any other representations and warranties made throughout this Master Agreement will be deemed provided by the parties on the Effective Date hereof and will be continuous in nature throughout the Term.

1.26 Specific Performance. COMPANY and EVERTEC agree that if an act or omission of one of the Popular Parties or any of their respective Subsidiaries, on the one hand, or EVERTEC, on the other hand, results in a breach of Section 1.5 (Assignment), Section 1.17 (Non-solicitation), Section 1.18 (Prohibition on Publicity), Section 1.31 (EVERTEC Change of Control), Section 2.9 (Non-Circumvention Covenants), Section 2.10 (Non-Compete Covenants), Article Five (Confidentiality, Privacy & Data Protection), Article Six (Security, BPPR Data & Records), Article Eight (Regulatory Compliance, Audit & Service Reviews) or Section 9.5 (Transition Assistance), EVERTEC or the Popular Parties, as applicable, will be irreparably damaged, no adequate remedy at law would exist and damages would be difficult to determine, and that EVERTEC or one of the Popular Parties, as applicable, shall be entitled to an injunction or injunctions to prevent such breach, and to specific performance of the aforementioned Sections and Articles as the case may be, in addition to any other remedy at law or equity, without having to post bond or any financial undertaking. By listing the provisions above the Parties are not intending to limit either Party's right to seek injunctive relief, specific performance or other equitable remedies under applicable Legal Requirements.

1.27 <u>Limitation of Actions</u>. No action, regardless of form, arising out of any claimed breach of this Master Agreement or the Services provided hereunder, may be brought by any party more than two (2) years after such party has obtained actual knowledge of the cause of action or after the statute of limitations prescribed by Puerto Rico law, whichever is less.

1.28 <u>Release of Existing Claims</u>. Each party hereto, on behalf of itself, its Affiliates and Subsidiaries (and each of their respective Representatives), including any successors and assigns of the foregoing (in such capacity, the "<u>Releasing Parties</u>"), hereby generally, completely and irrevocably releases and discharges each other party hereto and all of such other party's Affiliates and Subsidiaries (and each of their respective Representatives), including any successors and assigns of the foregoing (in such capacity, the "<u>Released Parties</u>"), of and from any and all claims, demands, debts, liabilities, contracts, obligations, accounts, suits, torts, causes of action, Disputes, or claims for relief of whatever kind or nature (other than Third Party claims) solely to the extent arising out of facts, circumstances or occurrences which were known or reasonably should have been known to the Releasing Parties, both at law and equity, which the Releasing Parties ever had, now has, or may have (including attorneys' fees and costs incurred) relating to, resulting from or arising out of the Released Parties' breach of any representations, warranties or covenants under

the September 2010 MSA or the April 2010 MSA; provided, that, this Section 1.28 shall not apply with respect to (i) any rights that cannot be waived under applicable Legal Requirements, (ii) the continuation on or after the Effective Date of any act or omission that commenced prior to the Effective Date, to the extent that such continuation of such act or omission following the Effective Date constitutes a breach of any representations, warranties or covenants under this Master Agreement, or (iii) any act or omission that commenced following the Effective Date which results in a breach of any representations, warranties or covenants under this Master Agreement. The Releasing Parties represent and warrant that they are respectively the sole and lawful owners of, and have not assigned or transferred to any Third Party, all rights, title and interest in and to all released matters referred to in this Section 1.28.

1.29 <u>Additional Assurances</u>. The parties covenant and agree that subsequent to the execution and delivery of this Master Agreement and without any additional consideration, each will execute and deliver any further legal instruments and perform any acts that are or may become necessary to effectuate the purposes of this Master Agreement.

1.30 No BPPR Guarantee. The parties acknowledge and agree that notwithstanding anything to the contrary contained in this Master Agreement, BPPR is party to this Master Agreement only to the extent that it receives Services from EVERTEC, and BPPR shall be liable to EVERTEC only for the performance of its (and its Subsidiaries') duties, obligations and payments under this Master Agreement. BPPR shall not guarantee or otherwise be liable for the performance of any duties, obligations or payments of any of BPPR's Affiliates (other than BPPR itself and BPPR's Subsidiaries) or for the performance of any duties, obligations or payments of COMPANY or COMPANY's Subsidiaries arising under this Master Agreement.

# 1.31 EVERTEC Change of Control.

(a) EVERTEC Change of Control. COMPANY and BPPR shall have the right, subject to Section 1.31(c), to terminate this Master Agreement up to thirty (30) days following the later of (i) the occurrence of an EVERTEC Change of Control or (ii) the date on which EVERTEC provides COMPANY and BPPR written notice that an EVERTEC Change of Control has occurred or is likely to occur (provided that if EVERTEC has not satisfied its obligations pursuant to Section 1.31(b) and that COMPANY or BPPR asserts such failure prior to the expiration of the thirty (30)-day period then such thirty (30)-day period shall be tolled until EVERTEC satisfies its obligations under Section 1.31(b) and provided further that if an EVERTEC Change of Control occurs, and EVERTEC fails to provide COMPANY and BPPR written notice thereof within thirty (30) days thereof, then COMPANY and BPPR shall have an unqualified right to terminate this Master Agreement), unless (w) the Person or Group of Persons proposing to engage in such proposed EVERTEC Change of Control transaction (the "Control Acquirer") is identified to COMPANY by EVERTEC at least thirty (30) Business Days prior to such proposed EVERTEC Change of Control; (x) neither the Control Acquirer nor any of its Affiliates is engaged, directly or indirectly, in the banking, securities, insurance or lending business, from which they derive aggregate annual revenues from Puerto Rico in excess of \$50 million unless none of them has a physical presence in Puerto Rico that is used to conduct any such business; (y) EVERTEC (or its successor, as applicable) will be Solvent immediately after and giving effect to such proposed EVERTEC Change of Control; and (z) EVERTEC (or its successor, as applicable), after the proposed EVERTEC Change of Control, will be capable of providing the Services at the level of service that is required under this Master Agreement.

- (b) Cooperation. EVERTEC shall use its Best Efforts to cooperate with COMPANY and BPPR in evaluating whether any proposed EVERTEC Change of Control would be in compliance with the requirements of this Section 1.31 including the ability of EVERTEC (or its successor, as applicable) to comply with the terms of this Master Agreement, including, in each case, by providing any non-confidential information regarding the purposes and plans in connection with such proposed EVERTEC Change of Control other than information that would create any potential liability under Legal Requirements, violate any confidentiality obligation, or that reasonably would be expected to result in the waiver of any attorney-client privilege.
- (c) Notice of Objection. If EVERTEC provides at least 30 days' written notice to COMPANY and BPPR prior to an EVERTEC Change of Control, BPPR and/or COMPANY shall notify EVERTEC in writing within 15 Business Days following receipt of EVERTEC's notice of the proposed EVERTEC Change of Control on the basis that it does not satisfy the criteria set forth in clauses (w) through (z) of Section 1.31(a) (unless EVERTEC has failed to satisfy its obligations pursuant to Section 1.31(b) and COMPANY or BPPR asserts such failure prior to the expiration of the 15 Business Day objection period, in which case such 15 Business Day objection period shall be tolled until EVERTEC satisfies its obligations pursuant to Section 1.31(b)). If BPPR or COMPANY fails to timely object to such proposed EVERTEC Change of Control (taking into account any tolling of the 15 Business Day objection period), it shall be deemed to have consented to such proposed EVERTEC Change of Control and waived its right of termination under Section 1.31(a).

### ARTICLE TWO- THE SERVICES

#### 2.1 Services.

- (a) EVERTEC will provide to COMPANY, BPPR, and their respective Subsidiaries the Services which are listed in Exhibit B, attached hereto, including the additional descriptions of certain of such Services in the document Rate Matrix and Printing & Distribution included as Schedule 1 to Exhibit B.
- (b) None of the Services shall be subject to any exclusivity. COMPANY, BPPR, and their respective Subsidiaries may use a Third Party to provide any of the Services and COMPANY, BPPR, and their respective Subsidiaries may perform any of the Services themselves or through their Subsidiaries, irrespective of whether any such Service has been terminated in whole or in part. To the extent there are exclusivity terms included in a Service Addendum or SOW existing as of the Effective Date, those exclusivity terms shall not apply after the Effective Date. To the extent there are minimum commitments or termination fees included in a Service Addendum or SOW as of the Effective Date, those termination fees and minimum commitments shall not apply after the Effective Date other than under the Statement of Work for Amazon Connect ACD and IVR Services dated November 3, 2021, as amended.

(c) If COMPANY or BPPR performs itself, or retains a third party to perform, any services that interface or interact with the Services, or that are or formerly were, part of the Services, EVERTEC will, at COMPANY or BPPR's cost, cooperate and coordinate with COMPANY and BPPR or such third party as reasonably required for COMPANY and BPPR or the third party to perform such services, EVERTEC's cooperation and coordination with COMPANY, BPPR and any such third party will include, as applicable: (i) providing reasonable access to the equipment and software used in providing the Services via API or other reasonable method agreed by the parties (subject to any applicable EVERTEC Security Requirements); (ii) providing such reasonable information regarding the operating environment, system constraints and other operating parameters as a person with reasonable commercial skills and expertise would find reasonably necessary for COMPANY and BPPR or the third party to perform its work; (iii) providing access to BPPR Data stored by EVERTEC; and (iv) reasonably cooperating and participating in meetings to identify the causes of incidents and problems and to agree on action plans to resolve them; provided that EVERTEC shall have sole discretion to deny any party from performing programming services on or accessing the source code of applications owned or licensed by EVERTEC. COMPANY or BPPR, as applicable, shall (and shall require any such third party engaged by COMPANY or BPPR to) comply with EVERTEC Security Requirements and with the other applicable terms of this Master Agreement when accessing EVERTEC's systems in connection with this Section 2.1(c). Subject to Section 4.5, COMPANY or BPPR will reimburse EVERTEC for any Losses incurred by EVERTEC in connection with any unauthorized access or intrusion to EVERTEC's Systems, or any Disabling Device introduced to EVERTEC's Systems, as a result of a breach of this Master Agreement, failure to comply with the EVERTEC Security Requirements, gross negligence, fraud or willful misconduct of COMPANY, BPPR or any such third party in connection with its access to EVERTEC's Systems pursuant to this Section 2.1(c).

(d) If EVERTEC and the Popular Parties, or any of their respective Subsidiaries, agree from time to time upon terms and prices for certain Services, then EVERTEC, on the one hand, and the Popular Parties or any of their appropriate respective Subsidiaries, on the other hand, will execute a separate addendum (each, a "Service Addendum") setting forth the mutually agreed upon terms and prices for such Services. Each such Service Addendum will be incorporated and, to the extent not incompatible with the terms of such Service Addendum, will be subject to the terms and conditions of this Master Agreement. Nothing herein will be interpreted as imposing an obligation upon EVERTEC to develop new Services, or upon COMPANY, BPPR, or any of their respective Subsidiaries to acquire any additional Services from EVERTEC. The parties agree that regardless of whether a Service is set forth in a Service Addendum or not, the pricing with respect to any and all Services will be set forth in Exhibit B, which will serve as a master list of the Services and corresponding pricing. The parties agree that Exhibit B will be reviewed and updated on an ongoing basis following the Effective Date hereof. Any changes to Exhibit B shall be agreed to by both parties by executing an amended and restated Exhibit B clearly denoting the date any such changes become effective; provided that EVERTEC may amend Exhibit B upon notice to, but without the consent of, the Popular Parties in order to reflect the price adjustments provided in Section 3.1(b), or any change, modification, enhancement or upgrade of the Services permitted pursuant to Section 2.1(b), but for the avoidance of doubt subject to the consent requirements of Sections 2.6 and 2.7.

(e) The Service Addenda set forth in Schedule 2 to Exhibit B are hereby incorporated into this Master Agreement by this reference.

- (f) Each Statement of Work ("<u>SOW</u>") pursuant to which EVERTEC provides Services(s) to COMPANY, BPPR, and/or their respective Subsidiaries as of the date hereof (except for any SOW that relates to a non-recurring definitive project) is set forth in <u>Schedule 3</u> to <u>Exhibit B</u> and is hereby incorporated into this Master Agreement.
- (g) Each SOW to be entered into after the date of this Master Agreement, pursuant to which EVERTEC shall provide Services(s) to COMPANY, BPPR, and/or their respective Subsidiaries after the date hereof shall be set forth in <a href="Schedule 4">Schedule 4</a> to <a href="Exhibit B">Exhibit B</a> and shall be incorporated into this Master Agreement.

### 2.2 Annual Minimums.

- (a) The Popular Parties agree to minimum annual service revenue (the "<u>Annual Minimum</u>") for all Services under this Master Agreement, payable on a monthly basis in an amount not to exceed the amounts described below (the "<u>Monthly Benchmark</u>") and subject to adjustment in accordance with this <u>Section 2.2</u> and pro-ration in the case of any partial months, as follows:
  - (i) with respect to the period from the Effective Date through September 30, 2025, the Monthly Benchmark shall be \$14,166,667 (for clarity, based on an Annual Minimum equal to \$170,000,000);
  - (ii) with respect to the period from October 1, 2025 through September 30, 2026, the Monthly Benchmark shall be \$13,750,000 (for clarity, based on an Annual Minimum equal to \$165,000,000); and
  - (iii) with respect to the period from October 1, 2026 through September 30, 2028, the Monthly Benchmark shall be \$13,333,333 (for clarity, based on an Annual Minimum equal to \$160,000,000).
- (b) Shortfall Payments and Cap. If Covered Revenue for a particular month during the periods referenced in Section 2.2(a) is less than the applicable Monthly Benchmark, then EVERTEC will include the amount of such shortfall on the invoice for the relevant month (prorated for partial months during the Initial Term) (a "Shortfall Payment"); provided, however, that Shortfall Payments shall be invoiced solely to the extent that they would not result in the sum of (x) Covered Revenue with respect to the period from January 1 of the calendar year in which the invoice was given through the invoice date and (y) aggregate Shortfall Payments previously invoiced during such period exceeding aggregate Monthly Benchmarks with respect to such period.
- (c) *True-Up*. If the sum of (x) the Covered Revenue plus (y) the aggregate Shortfall Payments made for the period beginning January 1 and ending June 30 or the period beginning July 1 and ending December 31 (each a "<u>True-Up Calculation Period</u>") exceeds the aggregate Monthly Benchmarks for the True-Up Calculation Period, EVERTEC shall refund to the Popular Parties an amount equal to the sum of the amounts referred to in clauses (x) and (y) of

this sentence *minus* the aggregate Monthly Benchmarks for the True-Up Calculation Period (the "<u>Bi-Annual True-Up</u>"); provided, however, that the Bi-Annual True-Up shall be capped at an amount equal to the aggregate Shortfall Payments made for the applicable True-Up Calculation Period; provided, further, that the Shortfall Payments and Bi-Annual True-Up shall be calculated consistently with the illustrative calculation included as <u>Exhibit H</u> hereto. EVERTEC will deliver its proposed calculation of any Bi-Annual True-Up (including if such amount would equal \$0, based on EVERTEC's proposed calculation) not later than thirty (30) days following the end of each True-Up Calculation Period and will provide the Popular Parties with such supporting information for such calculation as the Popular Parties may reasonably request. The amount of any Bi-Annual True-Up, as calculated by the Popular Parties in good faith taking into account EVERTEC's proposed calculation, will be applied as a credit against any amounts due to EVERTEC under this Master Agreement; provided, after June 30, 2028, and following any termination of this Master Agreement, the Popular Parties may elect to receive the Bi-Annual True-Up as a cash payment rather than a credit in which case EVERTEC will make such cash payment within thirty (30) days of a request from the Popular Parties. The amount of any Bi-Annual True-Up shall be subject to the fee dispute provisions set forth in this Master Agreement.

- (d) Excluded and Imputed Revenue. "Covered Revenue" means, with respect to any period, the sum of all amounts payable to EVERTEC in connection with the Popular Parties' receipt of Services or purchases from EVERTEC of other products or services under this Master Agreement; provided, however, that Covered Revenue shall be subject to adjustment as follows:
  - (i) The following amounts will be excluded in the calculation of Covered Revenue (i.e., such amounts will be disregarded in determining whether the Monthly Benchmark is met with respect to any particular month and in determining whether the Annual Minimum has been met):
    - (1) any fees that are part of the ATH Network Agreement or the ISO Agreement, as amended from time to time, that are not chargeable as fees in respect of the Services provided hereunder; and
    - (2) Covered Revenue shall exclude sales of hardware solely to the extent that such hardware was not purchased by COMPANY or BPPR in connection with the provision or consumption of Services.
  - (ii) The following amounts will be included in the calculation of Covered Revenue (i.e., such amounts will be deemed to have been paid to EVERTEC in determining whether the Monthly Benchmark is met with respect to any particular month):
    - (1) To the extent that the Popular Parties terminate any Service for cause or any Service is not provided (i) for a period of at least two (2) weeks due to a Force Majeure Event or (ii) any breach of this Master Agreement by EVERTEC, Covered Revenue shall be deemed to include the fees that would otherwise have been payable with respect to such Service; <u>provided</u>, that in the case of any Service with volume-based fees, the amount of such fees for purposes of this sentence shall be determined based on the amount of volume-based fees actually charged during the twelve (12) months preceding the date of termination of such Service.

- (2) If EVERTEC is requested to provide, but declines or is unable to provide, any change, modification, enhancement or upgrade of any Service because EVERTEC has a demonstrated lack of capacity or technical expertise to fulfill such request, then Covered Revenue shall be deemed to include any amounts that would have been payable to EVERTEC had EVERTEC agreed to such request, as determined based on (x) the costs and fees payable by COMPANY or BPPR, as applicable, to an independent third party engaged to provide such change, modification, enhancement or upgrade, or (y) if COMPANY or BPPR are unable to engage any such third party, the costs and fees to obtain professional services required to provide such change, modification, enhancement or upgrade (calculated based on programming hours) based on the rates charged by EVERTEC set forth in Exhibit B; provided, that COMPANY or BPPR shall promptly notify EVERTEC of the costs and fees determined in accordance with the foregoing clause (x) or (y), as applicable, and EVERTEC shall have ten (10) days after delivery of such notification to provide COMPANY and BPPR with a written commitment to meet such request (and for the avoidance of doubt, if EVERTEC meets such request, then this paragraph shall not apply and, instead, amounts payable to EVERTEC in respect of such request shall be counted toward Covered Revenue); provided, further, that this paragraph shall not apply in the case of any request that EVERTEC is unable to fulfill on commercially reasonable terms, so long as EVERTEC shall have used Best Efforts to maintain, develop and engage one or more third parties to provide such capacity; provided, further, that this paragraph shall only apply to the extent that (i) the request arises out of the obsolescence of a Service or (ii) the request relates to a Mandatory Enhancement.
- (3) If EVERTEC is requested to provide, but declines or is unable to provide, any new service that is reasonably related to the Services (including transition or conversion services), then Covered Revenue shall be deemed to include any amounts that would been payable to EVERTEC had EVERTEC agreed to such request, as determined based on the methodology described in the foregoing paragraph (2); provided, that in the case of any request by COMPANY or BPPR request for services that are not related to the Services, EVERTEC's failure to meet such request shall not be a basis for imputing toward Covered Revenue the fees that would have been earned with respect to such services had EVERTEC agreed to such request.
- (iii) The Monthly Benchmarks and the Annual Minimum are exclusive of taxes required to be paid by the Popular Parties in accordance with this Master Agreement.

(iv) The amount of any Service Level Agreement Credits will not alter the Monthly Benchmark for any month for purposes of the calculations in this Section 2.2. For the sake of clarity, credits under this Master Agreement, including any ordinary course Service Level Agreement Credits and Bi-Annual True-Ups and the credit delivered at the closing of the transactions contemplated under the Asset Purchase Agreement in accordance with Section 1.7 of the Asset Purchase Agreement, will be counted on a dollar-for-dollar basis toward the Monthly Benchmarks.

#### 2.3 Service Personnel.

- (a) EVERTEC agrees that it will (i) assign qualified, adequately trained, and efficient Personnel to the performance of the Services and (ii) cause such Personnel to use their Best Efforts to discharge EVERTEC's obligations under this Master Agreement in an efficient and timely manner and to exercise reasonable care in performing the Services, subject to the terms, conditions and prices established in the corresponding Service Addendum or Statement of Work.
  - (b) EVERTEC shall provide reasonable access, if requested by the Popular Parties, to the relevant Personnel.

### 2.4 Service Level Agreement.

- (a) The Services will be rendered in a commercially reasonable manner, in accordance with (i) the performance standards and service levels applicable to the particular Service in question and generally accepted industry practices and procedures used in performing services of a like-kind to the Services, but no less than with the same degree of care and diligence practiced prior to the Effective Date and (ii) unless agreed to otherwise in writing by the parties, the general terms and conditions and performance standards ("Service Levels") in the Service Level Agreement attached hereto as Exhibit C (together with Schedule 1 to Exhibit C, the "Service Level Agreement"), which is incorporated into this Master Agreement by reference.
- (b) Furthermore, the parties agree that EVERTEC's performance levels and procedures for Services shall, unless otherwise mutually agreed to by the parties, be no less than the performance levels and procedures prior to the Effective Date or, to the extent any such Services were not provided to COMPANY, BPPR, and their respective Subsidiaries prior to the Effective Date, such Services shall be provided at Service Levels that are at least consistent with then prevailing industry standards.
- (c) The parties agree that regardless of whether a Service is set forth in a Service Addendum or not, if the Service is subject to Service Levels, such Service Levels will be set forth in Schedule 1 of the Service Level Agreement which will serve as a master list of all Service Levels applicable to the Services COMPANY, BPPR, and their respective Subsidiaries receive hereunder. The parties agree that Schedule 1 will be reviewed and updated on an ongoing

basis following the Effective Date. Any changes to the Service Levels set forth in Schedule 1 shall be agreed to by the parties by executing an amended and restated Schedule 1 clearly denoting the date any such changes become effective. The parties agree that they shall negotiate in good faith with respect to any disputes arising from the updating of Schedule 1.

## 2.5 Reports and Errors.

- (a) COMPANY and BPPR will be responsible for reviewing and reconciling, the reports, statements, files, and any notice, correspondence or communication (collectively, the "Reports") it or its Subsidiaries receives from EVERTEC and each of COMPANY and BPPR agrees to exercise Best Efforts to do so in a commercially reasonable time period taking into account any deadlines imposed by any Legal Requirements.
- (b) Each of COMPANY and BPPR shall exercise Best Efforts to detect and inform EVERTEC in writing of any errors and/or discrepancies ("Errors") in the Reports received from EVERTEC within a commercially reasonable time period taking into account any deadlines imposed by any Legal Requirements.
- (c) The notice given by COMPANY or BPPR to EVERTEC shall specify and describe the Errors detected by COMPANY or BPPR, as the case may be, and the parties shall use their respective Best Efforts and cooperate with each other to obtain and/or provide to each other any available information that may be necessary, relevant and/or useful to identify the cause of any Errors and correct and resolve such Errors.
- (d) EVERTEC shall exercise Best Efforts to correct, resolve and/or reprocess the Errors reported to EVERTEC by COMPANY or BPPR within a commercially reasonable time period.
- (e) Any party may require that any controversy related to or arising under the provisions of this <u>Section 2.5</u> be processed as a Dispute pursuant to <u>Section 1.14</u>, but in such case the Dispute shall be immediately addressed by each party's Representative negotiating in good faith without having to comply with the Notice of Dispute and time periods set forth in <u>Section 1.14</u>.

## 2.6 Specifications for Services.

- (a) Based on COMPANY's and BPPR's instructions, EVERTEC will establish the processing parameter settings, features and options (collectively, the "Specifications") for the Services that will apply to COMPANY, BPPR, and their respective Subsidiaries (including all control and verification mechanisms utilized in each processing function related to the Services).
- (b) The parties agree that unless otherwise agreed to in writing by the parties, the Specifications will be derived from or in compliance with the respective terms and conditions established or documented by the parties using means such as, but not limited to: (i) electronic systems used to register service requests; (ii) operational manuals of COMPANY, BPPR, or their respective Subsidiaries provided to EVERTEC; (iii) SOWs; (iv) service standard forms

provided by COMPANY and/or BPPR to EVERTEC; and (v) any other written instructions that may be provided by COMPANY or BPPR to EVERTEC from time to time; provided, that in the event any Specification is documented by the parties pursuant to clauses (i) - (v) in this Section 2.6(b) and such Specification relates to a recurring Service (as opposed to a definitive project), Exhibit B shall be amended to include such Specification and the corresponding fee agreed to by the parties.

## (c) Changes to Specifications:

- (i) To the extent a particular Service is provided solely to COMPANY, BPPR, and their respective Subsidiaries (and not to EVERTEC's other customers), EVERTEC will disclose to COMPANY, and COMPANY will have a reasonable period of time to approve, all Specifications and modifications thereof before their activation in the production environment. If the Specifications or modifications are required by Legal Requirements, and COMPANY does not approve or provide alternative Specifications or modifications to EVERTEC, EVERTEC will not be responsible for any failure of such Service to comply with the applicable Legal Requirements.
- (ii) In cases where EVERTEC makes a change to Specifications for Services provided to EVERTEC's other customers, or in order to comply with this Master Agreement or Legal Requirements applicable to EVERTEC, EVERTEC will provide COMPANY and BPPR with reasonable prior written notice of the change and meet with COMPANY to address any concerns COMPANY may have about the change.
  - (iii) EVERTEC will perform system processing and provide the Services in accordance with the Specifications.
- (d) Should the Specifications call for the use of software owned by or under license to EVERTEC, or should EVERTEC be required to procure hardware, software or other items in order to perform the Services in accordance with the Specifications, (i) COMPANY, BPPR, or one of their respective Subsidiaries may subscribe to a separate license agreement for such software; and/or (ii) without limiting EVERTEC's responsibilities or liabilities under this Master Agreement, EVERTEC will pass through and assign to COMPANY, BPPR, or one of their respective Subsidiaries all warranties provided by the manufacturer(s) and/or licensor(s) of such items; provided, however that all disclaimers and/or limitations of liabilities relating to such software, hardware or other items will be deemed extended to include EVERTEC.

### 2.7 Modifications to Services.

(a) Subject to Section 8.2, EVERTEC reserves the right to change, modify, enhance or upgrade the manner in which it renders the Services, at any time, provided, however, that any change, modification, enhancement or upgrade does not adversely affect the functionality of the Services, the fees for such Service or the agreed upon Service Levels, as applicable, and provided that (i) EVERTEC provides written notice of such modification to COMPANY and BPPR at least forty-five (45) days prior to implementation of any such change, modification, enhancement or upgrade, (ii) such notice describes in reasonable detail the change, modification, enhancement or upgrade to be made by such modification and EVERTEC promptly answers any reasonable inquiries of COMPANY, BPPR, or one of their respective Subsidiaries regarding such

change, modification, enhancement or upgrade and (iii) neither COMPANY nor BPPR delivers a written notice to EVERTEC prior to such implementation that it reasonably believes that such change, modification, enhancement or upgrade would be likely to adversely affect its or one of its Subsidiaries' compliance with applicable Legal Requirements.

- (b) Any change, modification, enhancement or upgrade requested or required by COMPANY or BPPR that is not a Mandatory Enhancement or a Supplemental Mandatory Enhancement (a "Requested Enhancement") will require written notice to EVERTEC. Upon receipt of such notice, EVERTEC will exercise Best Efforts to prepare and present to COMPANY or BPPR, as soon as possible, a written estimate of the costs for the Requested Enhancement and any adjustment in fees that may be necessary as a result thereof. EVERTEC's Best Efforts will take into consideration the business needs of COMPANY, BPPR, and their respective Subsidiaries and any timeframes for implementation related thereto. COMPANY, BPPR, and their respective Subsidiaries may decide that the Requested Enhancement will not be developed by EVERTEC and this Master Agreement will continue in full force in effect under the then current terms and conditions, and COMPANY, BPPR, and their respective Subsidiaries shall, notwithstanding anything to the contrary contained in this Master Agreement but subject to the terms and conditions of the Technology Agreement, be permitted to develop such Requested Enhancement themselves or engage a Third Party, in each case at no cost or expense to EVERTEC. Following the completion of such development, EVERTEC shall, if requested by COMPANY or BPPR, cooperate in good faith and otherwise use its Best Efforts to implement the Requested Enhancement at the then present rates in Exhibit B. If implementation of the Requested Enhancement results in a material increase in costs to EVERTEC as opposed to the provision of the applicable Service without such Requested Enhancement, (i) EVERTEC will provide COMPANY and BPPR with a reasonably detailed explanation of the incremental costs for implementing and operating such Requested Enhancement and (ii) any adjustments to the fees necessary to address such incremental costs shall be agreed among the parties in good faith.
- (c) Without limiting the terms in <u>Sections 2.7(a)</u> and (b), if EVERTEC intends to replace any product used to provide the Services under this Master Agreement with an alternate product, EVERTEC will provide COMPANY and BPPR with at least 180 days' notice of the replacement.

# 2.8 Mandatory Enhancements.

- (a) Subject to <u>Section 8.2</u> and the remainder of this <u>Section 2.8</u>, EVERTEC will provide changes, modifications, enhancements or upgrades to the Services to ensure that those Services permit COMPANY, BPPR, or their respective Subsidiaries to comply with mandatory changes in COMPANY's, BPPR's, or their respective Subsidiaries' applicable Legal Requirements, including as requested by COMPANY, BPPR or their respective Subsidiaries (the "<u>Mandatory Enhancements</u>").
- (b) EVERTEC shall notify COMPANY and BPPR of its intention to implement a Mandatory Enhancement as soon as practicable following EVERTEC's becoming aware of such Mandatory Enhancement and, in the 30 days following such notice, shall (i) consult with COMPANY and BPPR, as well as other EVERTEC customers affected by the relevant Mandatory Enhancement, regarding the interpretation of the relevant Legal Requirements as well as

the strategy for implementation of the Mandatory Enhancement and (ii) provide an initial estimate of the potential applicable costs. If the Mandatory Enhancement is to be funded or used exclusively by or for COMPANY or BPPR or their Affiliates (an "Exclusive Mandatory Enhancement"), the Mandatory Enhancement will reflect COMPANY and BPPR's interpretation and requirements for the Mandatory Enhancement. If the Mandatory Enhancement is not an Exclusive Mandatory Enhancement, the design and implementation of the Mandatory Enhancement shall be based on EVERTEC's reasonable interpretation of any relevant Legal Requirement applicable to EVERTEC (taking into account input from COMPANY and BPPR) and its understanding of the Services affected thereby.

- (c) EVERTEC may charge COMPANY or BPPR at the time and material rates set forth in Exhibit B, as applicable, for development hours expended on Mandatory Enhancements for COMPANY, BPPR, or any of their respective Subsidiaries on a prorated basis. The costs will be proportionately allocated among EVERTEC's affected customers. Any such charges shall be disclosed to COMPANY and BPPR as soon as reasonably possible following the consultations described in Section 2.8(b), including a description of how they are to be calculated and allocated among COMPANY, BPPR, and their appropriate respective Subsidiaries, on the one hand, and EVERTEC's other affected customers, on the other. If (i) COMPANY, BPPR or any applicable Subsidiary thereof inform EVERTEC in writing that they are not subject to the Legal Requirement that is the subject of a given Mandatory Enhancement within 30 days following EVERTEC's notice in clause (b) above, or (ii) COMPANY or BPPR elect not to use the Mandatory Enhancement by provision of written notice within 45 days following EVERTEC's notice in clause (b) above, COMPANY, BPPR, and their appropriate respective Subsidiaries shall not be charged any development fees for such enhancement and EVERTEC may determine, in its sole discretion, to not make such enhancement available to COMPANY, BPPR and their Subsidiaries.
- (d) In the event that COMPANY, BPPR, or any of their respective Subsidiaries requests other changes, modifications, enhancements or upgrades to the Services that are different from or in addition to, but requested in connection with, Mandatory Enhancements (the "Supplemental Mandatory Enhancements"), then, provided EVERTEC agrees to make such changes, modifications, enhancements or upgrades (which agreement shall not be unreasonably delayed, withheld or conditioned), COMPANY, BPPR, or any of their respective Subsidiaries that request a Supplemental Mandatory Enhancement shall be charged a development fee at the time and material rates specified in Exhibit B, as applicable.
- (e) EVERTEC may charge COMPANY, BPPR, or any of their respective Subsidiaries that request a Supplemental Mandatory Enhancement an ongoing usage fee for any new Service resulting from (i) Mandatory Enhancements (provided that if such Mandatory Enhancements are provided to EVERTEC's other customers, such charges may be imposed solely to the extent charged to such other customers), or (ii) Supplemental Mandatory Enhancements.
- (f) For the avoidance of doubt, COMPANY, BPPR, or any of their respective Subsidiaries can develop themselves or use a Third Party for any Requested Enhancement, Mandatory Enhancement or a Supplemental Mandatory Enhancement; subject to and in accordance with <u>Section 2.1(c)</u>.

#### 2.9 Non-Circumvention Covenants.

- (a) EVERTEC agrees that it shall not, and shall cause its Affiliates not to, directly initiate or procure any negotiations or contacts with any Client of COMPANY, BPPR, or any of their respective Subsidiaries that are designed or intended to:
  - (i) induce, cause, or propose to any such Client to terminate any of the accounts and/or any other business relationships such Client may have directly with COMPANY, BPPR, or any of their respective Subsidiaries (including accounts or relationships for which COMPANY, BPPR, or any of their respective Subsidiaries shall have engaged EVERTEC to provide any of the Services), or
  - (ii) induce, cause, or propose to any Client of COMPANY, BPPR, or any of their respective Subsidiaries to (A) reduce or curtail the amount or type of services or goods that COMPANY, BPPR, or any of their respective Subsidiaries provides or sells to such Client, (B) reduce the amount of business with, prices paid by, or revenue from, a Client or (C) to seek to transfer such accounts and/or business relationships from COMPANY, BPPR, or their respective Subsidiaries to another Person.
- (b) The Popular Parties agree that they shall not, nor cause any of their respective Subsidiaries to, directly initiate or procure any negotiations or contacts with any of EVERTEC's clients to which EVERTEC provides services that are substantially similar to the Services provided to the Popular Parties under this Master Agreement or any service providers of EVERTEC identified on Schedule 2.9 with respect to the application(s) listed for each such service provider in Schedule 2.9 that are designed or intended to:
  - (i) induce, cause, or propose to any such client or service provider to terminate any of the accounts and/or any other business relationships such client or service provider may have directly with EVERTEC, or
  - (ii) induce, cause, or propose to any such client or service provider of EVERTEC to (A) reduce or curtail the amount or type of services or goods that EVERTEC provides or sells to such client or receives from such service provider, (B) reduce the amount of business with, prices paid to, or revenue from, a client, (C) reduce the amount of business with, or increase the prices paid to a service provider, or (D) seek to transfer such accounts and/or business relationships from EVERTEC to another Person;
  - (iii) <u>provided</u>, that the mere fact that COMPANY, BPPR, or one of their respective Subsidiaries shall have engaged EVERTEC to provide any of the Services to its Clients will not cause such Clients to be considered "clients" or "accounts" and/or "business relationships" of EVERTEC for purposes of this <u>Section 2.9</u>.
- (c) Nothing in this Section 2.9 shall be interpreted to preclude or prevent either party from (i) distributing marketing materials on its products and services to the general public; (ii) engaging one or more third parties to provide services to such party in connection with a transition or termination of the Services or services provided under other agreements between the parties; or (iii) responding to requests for products and services from the other party's respective

clients or other Persons, but, in the case of EVERTEC, subject to the limitations imposed under Section 2.10(a). In addition, nothing in paragraphs (a) or (b) of this Section 2.9 shall be interpreted to preclude or prevent EVERTEC from entering into any agreement with a Person in the same industry or a similar industry which offers similar products or services as COMPANY (a "Competitor") pursuant to which such Competitor will provide to its clients any one or more services similar to the Services; provided, however, that EVERTEC in its dealings with such Competitor shall at all times comply with its confidentiality obligations under this Master Agreement. The provision of services pursuant to the ISO Agreement or pursuant to an exception expressly set forth in the ISO Agreement will not be deemed a breach of this provision.

(d) Neither party shall instruct or intentionally procure a third party to take actions that would, if taken by such party directly, violate this Section 2.9.

### 2.10 Non-Compete Covenants.

(a) Notwithstanding the provisions of Section 2.9, EVERTEC agrees that, without COMPANY's, or its relevant Subsidiary's, prior written consent, it shall not offer, provide or market any of the Restricted Payment Processing Services (as defined below) to any of the Strategic Clients (as defined below). With respect to Restricted Payment Processing Services for which COMPANY or its relevant Subsidiaries have provided such written consent, the parties agree that they shall cooperate with each other to provide, offer and market the Restricted Payment Processing Services to Strategic Clients. In the event that BPPR ceases to offer a service included in the definition of "Restricted Payment Processing Services" to a Strategic Client, such service shall no longer be included in the list of Restricted Payment Processing Services with respect to such Strategic Client, and EVERTEC shall be permitted to offer such service to such Strategic Client without any restriction. In the event that BPPR offers any service not provided by EVERTEC that has features and functionalities that are materially similar to those of any service included in the definition of "Restricted Payment Processing Services" to any of its clients, such service shall no longer be included in the list of Restricted Payment Processing Services and EVERTEC shall be permitted to offer such service without any restriction.

#### (b) For purposes of this Section 2.10(b):

- (i) "Restricted Payment Processing Services" means the payment processing services offered by COMPANY, BPPR, or one of their respective Subsidiaries to Clients as of the Effective Date, and that are listed below:
  - (1) Multi Merchant Pay
  - (2) Call Center Pay
  - (3) Kiosk
  - (4) Check-Out Payment
  - (5) PER/PER WEB
  - (6) Lockbox Retail/Wholesale

- (7) Telepago Online
- (8) IVRU
- (9) Payment Station
- (10) Payments on BPPR ATMs
- (11) Any other payment or collection services, other than such services for which BPPR receives, or has been offered and refused to receive, (i) Interchange Fees and (ii) MAB Revenue Sharing, P2B Revenue Sharing (as each such term is defined in the ISO Agreement), or any other revenue sharing arrangement that is no less favorable to BPPR than the terms of the MAB Revenue Sharing and P2B Revenue Sharing (as each such term is defined in the ISO Agreement).
- (ii) "<u>Strategic Clients</u>" means the Clients listed in <u>Exhibit D</u> hereto, which shall be subject to the provisions of this <u>Section 2.10</u> so long as such Clients maintain an account and/or business relationship with COMPANY, BPPR, or one of their respective Subsidiaries. <u>Exhibit D</u> may be amended only with the prior written consent of COMPANY and EVERTEC.

# 2.11 Equipment and Other Resources.

- (a) EVERTEC will retain all right, title or interest in any EVERTEC equipment supplied to COMPANY, BPPR, or any of their respective Subsidiaries as part of the Services ("EVERTEC Equipment"), and no ownership rights in such EVERTEC Equipment will transfer to COMPANY, BPPR, or any of their respective Subsidiaries. COMPANY, BPPR, and their respective Subsidiaries will, as applicable, provide a suitable and secure environment, free from environmental hazards, and electric power for such EVERTEC Equipment located in premises operated or controlled by COMPANY, BPPR, or one of their respective Subsidiaries, and will keep EVERTEC Equipment free from all liens, charges, and encumbrances. COMPANY, BPPR, and their respective Subsidiaries, as applicable, will bear the risk of loss of or damage to EVERTEC Equipment (ordinary wear and tear excepted) from any cause except to the extent caused by EVERTEC or its suppliers. COMPANY, BPPR, and their respective Subsidiaries agree that they will not remove, relocate, modify, or interfere with EVERTEC Equipment, or attach EVERTEC Equipment to non-EVERTEC equipment without prior written authorization from EVERTEC.
- (b) Title to and risk of loss of any equipment purchased from EVERTEC will pass to COMPANY, BPPR, or their respective Subsidiaries as of delivery, upon which date EVERTEC will have no further obligations of any kind with respect to such purchased equipment, except as set forth in a document executed by the appropriate parties.
- (c) All ownership or leasehold interest in a party's facilities, and associated equipment used in connection with the Services, will at all times remain with that party.

(d) If any equipment, software or other COMPANY Third Party Intellectual Property or other Intellectual Property of COMPANY, BPPR, or any of their respective Subsidiaries ("COMPANY Resources") is made available by COMPANY or BPPR to EVERTEC under this Master Agreement: (i) COMPANY, BPPR, or their appropriate respective Subsidiaries, hereby grants EVERTEC a non-transferable and non-exclusive license to use such COMPANY Resources solely in the manner necessary to provide the Services under this Master Agreement, subject to any restrictions and other terms in the BPPR Pass Through Licenses or that otherwise may be provided in this Master Agreement or by COMPANY or BPPR to EVERTEC in writing; and (ii) EVERTEC will not use or access the COMPANY Resources for any purpose other than providing the Services under this Master Agreement.

## 2.12 Import/Export Control.

- (a) The parties acknowledge that equipment, products, software, and technical information (including technical assistance and training) provided under this Master Agreement may be subject to import or export laws, conventions or regulations, and any use or transfer of the equipment, products, software, and technical information must be in compliance with all such laws, conventions and regulations. The parties will not use, distribute, transfer, or transmit the equipment, products, software, or technical information (even if incorporated into other products) except in compliance with such laws, conventions and regulations. If requested by either party, the other party agrees to sign written assurances and other documents as may be required to comply with such laws, conventions and regulations.
- (b) In the event any necessary import or export license cannot be obtained within six (6) months after making an application, no party will have further obligations with respect to providing or purchasing and, if applicable, COMPANY, BPPR, or one of their respective Subsidiaries will return the equipment, products, software, or technical information that is the subject matter of the unsuccessful application.

### 2.13 Contingency Planning.

- (a) Each party acknowledges that it is responsible for maintaining in effect at all times during the Term appropriate Business Continuity Plans for their respective Systems, personnel, and operations in compliance with applicable Legal Requirements and Industry Standards.
  - (b) EVERTEC represents and warrants that it:
  - (i) has, as of the Effective Date, and shall at all times during the Term maintain updated Business Continuity Plans designed to minimize the risks associated with an Event affecting EVERTEC's ability to continue to provide the Services in compliance with this Master Agreement;
    - (ii) will implement its applicable Business Continuity Plans upon the occurrence of an Event; and

- (iii) shall resume the affected Services within the Recovery Time Objectives and Recovery Point Objectives established as of the Effective Date, including those provided in Disaster Recovery Services Addendum or other applicable SOWs and Schedules, or as may subsequently be agreed to between the parties for the Services.
- (c) EVERTEC may continue using its alternate site in Puerto Rico solely for Services covered by that site as of the Effective Date until the parties agree in an SOW to migrate those Services to another site outside of Puerto Rico. EVERTEC will use alternate sites located outside of Puerto Rico for any additional Services added to the Business Continuity Plan.
- (d) The Business Continuity Plans must include a Disaster Recovery Plan and a Disaster Recovery Strategy. Such Business Continuity Plans must address, at a minimum, responsibilities and required skills for Personnel and Third Party service providers; solutions to foreseeable disruptions, including those emanating from cyber threats; escalation thresholds; immediate steps to protect Personnel and customers and minimize damage; prioritization and procedures to recover functions, services, and processes; critical information protection (e.g., physical, electronic, hybrid, and use of off-site storage); logistical arrangements (e.g., housing, transportation, or food) for Personnel at the recovery locations; network equipment, connectivity and communication needs, including entity-owned and personal mobile devices; Personnel at alternate Data Center sites, including arrangements for those permanently located at the alternate Data Center facility; remote access from home; scope and frequency of testing; and resumption of a normalized state for business processes.
- (e) EVERTEC will review and, if necessary to reflect changes in the Services and to remain in compliance with the requirements in this Master Agreement, update its Business Continuity Plans at least once each year. Such updates shall (i) take into account any audits and test results, as well as changes to applicable Legal Requirements and Industry Standards; and (ii) not increase RTO or RPO times without BPPR's prior written consent.
- (f) EVERTEC will provide BPPR with a summary of its Business Continuity Plans in effect on the Effective Date and will promptly provide BPPR with an updated summary upon any amendment thereof. Upon request, EVERTEC will also provide BPPR with access to the full Business Continuity Plans for review on-site or via another secure method agreed by the parties. Without BPPR's prior written consent, EVERTEC will not make any amendments to its Business Continuity Plans that adversely affect BPPR's rights under this Master Agreement or the Recovery Time Objectives or Recovery Point Objectives agreed to with BPPR.
- (g) EVERTEC will test the operability of its Business Continuity Plans, including all Systems and operations that support the Services, in the main and alternate Data Center sites, as well as the integrity of its applicable backup processes, at least once each year. Within thirty (30) days after any Business Continuity Plan tests are conducted, EVERTEC will provide a summary of the results to BPPR; provided that (i) provision of such summary by EVERTEC shall be subject to, and limited by, applicable Legal Requirements, (ii) EVERTEC may redact from such results information relating to EVERTEC's internal business or to EVERTEC's other customers (to the extent the information does not also relate to COMPANY or BPPR), and (iii) EVERTEC shall not be obligated to provide any information if doing so would reasonably be expected to result in the waiver of any attorney-client or other legal privilege.

- (h) Upon COMPANY's or BPPR's reasonable request, EVERTEC shall make available to COMPANY or BPPR one or more Representatives who are knowledgeable about the Business Continuity Plans to respond to questions concerning EVERTEC's Business Continuity Plans, the manner in which they are tested and the manner in which they would be implemented in an Event. In the event that COMPANY or BPPR determines that applicable Legal Requirements require additional disaster recovery services or modifications to EVERTEC's existing Business Continuity Plans, EVERTEC agrees to cooperate with COMPANY and BPPR in assuring that its Business Continuity Plans comply with applicable Legal Requirements in accordance with Sections 8.1 and 8.2.
- (i) In addition to its own Business Continuity Plans, EVERTEC shall verify that its Data Subcontractors have documented plans providing appropriate terms for the implementation, maintenance and performance of their own business continuity and disaster recovery efforts related to the portion of the Services they provide and use Best Efforts to ensure that such plans are tested at least annually.
- (j) If the Services or any portion thereof are affected by an Event, EVERTEC will notify BPPR's designated authorized representative as soon as practicable (and in any event within the timeframe required in the applicable SOW or Schedule or otherwise agreed with BPPR) after the activation of its Business Continuity Plan(s). If an Event gives rise to a business interruption that renders EVERTEC unable to provide the Services, or a critical portion thereof, and such interruption is expected to remain, or remains, unresolved for longer than the applicable RTO for the relevant Service BPPR may, in addition to other remedies provided in this Master Agreement, procure such Service from a Third Party without penalty, and upon request, EVERTEC will provide its cooperation in the transfer of the related activities and databases to another service provider without cost to BPPR for the duration of such business interruption.

#### 2.14 Procedures Manual.

- (a) Subject to Sections 2.14(b), 2.14(c) and 2.14(d), EVERTEC will prepare and maintain a current set of documents that describe how EVERTEC provides, and COMPANY, BPPR and their Subsidiaries receive, the Services provided under this Master Agreement, and how the Services are integrated with COMPANY's, BPPR's and their Subsidiaries' Systems and processes (including escalation procedures and contacts, applicable processing schedules, operations processes and manuals, user guides, applicable Specifications, equipment and software used and knowledge repositories) (collectively, the "Procedures Manual").
- (b) To the extent components of the Procedures Manual do not exist as of the Effective Date, EVERTEC will create them either as a project under an SOW or as part of the Transition Assistance provided under Section 9.5, in each case at COMPANY's or BPPR's cost. COMPANY or BPPR may request updates to the Procedures Manual on an annual basis pursuant to an SOW.
- (c) EVERTEC will provide COMPANY and BPPR with the Procedures Manual during any applicable Transition Period and otherwise upon request during the Term.

(d) With respect to proprietary EVERTEC applications, EVERTEC will provide information about how COMPANY and BPPR use the products and receive the Services (e.g., configurations, integrations, escalation procedures, and rules specific to COMPANY and BPPR) but will not be required to provide Proprietary Application Information. "<u>Proprietary Application Information</u>" means Confidential Information about EVERTEC's internal processes and procedures, software code, or other proprietary service delivery materials, in each case as they relate to proprietary EVERTEC applications.

### ARTICLE THREE- PAYMENT FOR SERVICES

### 3.1 Fees.

- (a) In consideration for EVERTEC providing COMPANY, BPPR, and their respective Subsidiaries the Services, each of COMPANY and BPPR agrees to pay, or cause one of its respective Subsidiaries to pay, EVERTEC the corresponding fees set forth in Exhibit B, subject to adjustment as set forth in Section 3.1(b). The parties agree that regardless of whether a Service is set forth in a Service Addendum or not, the pricing relative to any and all Services will be set forth in Exhibit B, which will serve as a master list of the Services and corresponding pricing. The parties agree that Exhibit B will be reviewed and updated on an ongoing basis following the Effective Date hereof. Any changes to the fees set forth in Exhibit B (other than the adjustments described in Section 3.1(b)) shall be agreed to by both parties by executing an amended and restated Exhibit B clearly denoting the date any such changes become effective.
  - (b) The fees set forth in Exhibit B shall be adjusted annually as follows:
  - (i) on September 30 of each of 2022, 2023, 2024 and 2025, the fees will be adjusted for changes in the CPI in the prior twelve (12) months; provided that the fee adjustment cannot be negative, nor can it exceed one and one half percent (1.5%) per annum; and
  - (ii) on September 30 of any year thereafter, the fees will be adjusted based on the following formula: any changes in the CPI in the prior twelve (12) months *less* 2%; provided that the fee adjustment cannot be negative, nor can it exceed two percent (2%) on any given year.
- (c) Notwithstanding anything in this Master Agreement to the contrary, (i) fees charged under the ATH Network Agreement shall be subject to CPI adjustments under such agreement and not under Section 3.1(b), it being understood, for clarity, services provided under the ATH Network Agreement shall not be included on Exhibit B, and (ii) no adjustment pursuant to Section 3.1(b) shall apply to fees charged with respect to any Service that has been provided for less than twelve (12) months prior to the applicable annual adjustment under this Master Agreement unless otherwise agreed to in writing.

- (d) All fees set forth in Exhibit B or elsewhere in this Master Agreement or an SOW will be reduced by ten percent (10%) from their then current amounts effective October 1, 2025, unless an applicable SOW executed after the Effective Date specifically states that will not be subject to this adjustment. For clarification, the reduced fees will remain subject to the adjustments set forth in Section 3.1(b)(ii) above in subsequent years.
- 3.2 Terms of Payment. EVERTEC will send an invoice directly to COMPANY, on or before the fifteenth (15th) day of the month following the month in which the Services are rendered, reflecting the fees and other charges to COMPANY, BPPR, and their respective Subsidiaries for the preceding month. COMPANY, BPPR, or one of their respective Subsidiaries, as stated on the invoice, will pay to EVERTEC all undisputed amounts due under this Master Agreement within thirty (30) days from the date of receipt of the invoice, unless otherwise agreed to by the parties for a particular Service in writing. Any undisputed amount due under this Master Agreement that is not paid when due will thereafter bear interest at an annual rate of interest equal to one and a half percent (1.5%), but in no event shall exceed the maximum rate of interest allowed under any Legal Requirement. COMPANY and BPPR agree that, if any properly submitted invoice remains unpaid and undisputed for a period exceeding sixty (60) days, EVERTEC may (i) refuse to provide the Services until such time as all past due amounts are paid in full or (ii) terminate this Master Agreement in accordance with and subject to Section 9.2(a)(ii).
- 3.3 <u>Services Rendered during Legal Holidays</u>. Unless agreed to otherwise in writing, upon request of COMPANY or BPPR, and provided EVERTEC has available resources to comply therewith, EVERTEC will provide the Services to COMPANY, BPPR, and their respective Subsidiaries during Legal Holidays for the fees set forth in <u>Exhibit B</u> or the corresponding Service Addendum, as applicable.
- 3.4 <u>Additional Services</u>. If COMPANY, BPPR, or one of their respective Subsidiaries' fail to perform their respective responsibilities under this Master Agreement, EVERTEC will promptly notify COMPANY and BPPR of the failure. If COMPANY and BPPR do not remediate the failure after receiving such notice, EVERTEC may charge COMPANY and BPPR for out-of-pocket costs EVERTEC is required to incur to perform services over and above the Services listed in <u>Exhibit B</u> of this Master Agreement as a result of COMPANY and BPPR's failure (e.g., costs for computer and personnel time, equipment, supplies, and other items). EVERTEC will obtain COMPANY or BPPR written approval prior to incurring any such expenses; <u>provided</u> that until such approval is obtained EVERTEC shall not be responsible for performing any of the affected Services.
- 3.5 <u>Out-of-pocket and Third Party Expenses</u>. With COMPANY's or BPPR's prior written approval, additional costs related to delivery and/ or collection, telecommunications (other than the telecommunication installation and maintenance services set forth in <u>Exhibit B</u>) or other incidental services, as well as necessary and reasonable services to be provided through EVERTEC by Third Parties, for COMPANY, BPPR, or one of their respective Subsidiaries' benefit, incurred during the Term and that are not contemplated in any of the established fees, costs, or charges, will be paid by COMPANY, BPPR, or one of their respective Subsidiaries, as applicable, when invoices and related documents are duly presented. All such out-of-pocket or Third Party charges and administration costs related to the Services will be billed by EVERTEC to COMPANY, BPPR, or one of their respective Subsidiaries at an amount equal to cost; <u>provided</u> that to the extent COMPANY, BPPR, or one of their respective Subsidiaries sets forth any limitations on such charges and costs, COMPANY, BPPR, and their respective Subsidiaries shall not be billed for amounts in excess of such limitations.

3.6 Taxes. EVERTEC will be solely responsible for the payment of any income tax, gross receipt tax or municipal license tax related to the fees and charges paid by the Popular Parties under this Master Agreement. If the Popular Parties are required by any Legal Requirements to withhold taxes on the fees and charges paid to EVERTEC, the Popular Parties will deduct such taxes and pay them to the appropriate taxing authority unless EVERTEC provides evidence of an exemption or waiver. The Popular Parties will provide EVERTEC proof of payment of the taxes attributable to the Services rendered under this Master Agreement as provided by applicable Legal Requirements. Notwithstanding the foregoing, the Popular Parties' failure to withhold any applicable taxes, in whole or in part, will not relieve EVERTEC of its responsibility to pay any such taxes when due. The Popular Parties will be responsible for the payment of sales and use taxes, value added, and any other tax as provided by applicable law or regulation and related to the fees and charges paid to EVERTEC. When applicable, EVERTEC will include such sales and use taxes, or any other applicable tax, as a separate line item on all applicable invoices.

## 3.7 Disputed Charges; Requests for Information.

- (a) Each of COMPANY, BPPR, and their respective Subsidiaries may withhold payment of specific charges within a given invoice that it in good faith disputes or for which it may require additional information from EVERTEC to verify the amounts being charged, <u>provided</u> that COMPANY, BPPR, or such Subsidiary delivers to EVERTEC a written statement on or before the date in which such payment is due, describing in reasonable detail (i) the specific charge or charges being disputed and the basis of the dispute, (ii) if applicable, the supporting documentation that is reasonably required for verification of the charge or charges, and (iii) the amount being withheld.
- (b) A charge will be deemed "undisputed" if COMPANY, BPPR, and their respective Subsidiaries do not deliver the aforementioned written statement within the time period provided in this <u>Section 3.7</u>.
- (c) Notwithstanding the foregoing, the parties will have the right to review invoices generated hereunder and claim any under charged or over paid amounts. The parties shall make any such claims within one hundred and twenty (120) days following the date of the invoice.
- (d) Any Dispute related to the charges or fees payable under this Master Agreement, if not settled by the parties, shall be resolved in accordance with <u>Section 1.14</u> hereof.
- 3.8 <u>Supporting Documentation</u>. EVERTEC will maintain, at its sole cost, supporting documentation for the amounts billable to, and payments made by, COMPANY, BPPR, and their respective Subsidiaries hereunder in accordance with its practices prior to the Effective Date and applicable record retention requirements. COMPANY and BPPR will have the right to request supporting documentation of any invoice submitted, paid or disputed, at any time, and EVERTEC agrees to cooperate by providing such the supporting documentation at no cost. For the sake of clarity, invoices may only be disputed during the time period provided in <u>Section 3.7</u>.

3.9 No Right to Set-Off. COMPANY, BPPR, and their respective Subsidiaries shall pay to EVERTEC the full amount of undisputed charges and any disputed charges that are resolved in favor of EVERTEC and other amounts required to be paid by COMPANY, BPPR, and their respective Subsidiaries under this Master Agreement and COMPANY, BPPR, and their respective Subsidiaries shall not set-off, counterclaim or otherwise withhold any amount owed or claimed to be owed to EVERTEC under this Master Agreement on account of any obligation owed by EVERTEC or any of its Subsidiaries, whether or not such obligation has been finally adjudicated, settled or otherwise agreed upon in writing.

# 3.10 Core API Layer SOW

- (a) The parties agree as follows, notwithstanding anything in this Master Agreement to the contrary:
- (i) Twenty percent (20%) of all fees otherwise payable to EVERTEC pursuant to the Core API Layer SOW shall be withheld until EVERTEC has delivered 100% of all Core Banking APIs (as defined in the Core API Layer SOW), has completed all database integrations required under the Core API Layer SOW, and has exposed all APIs required to be exposed on the Amazon Web Services environment as required thereunder, and BPPR has received a certification from a mutually agreed third-party regarding the delivery of 100% of all Core Banking APIs (the "Core API Completion"). If the Core API Completion has not occurred on or prior to the Core API Deadline, then (A) thirty five percent (35%) of any fees otherwise payable pursuant to the Core API Layer SOW after the Core API Deadline shall be withheld until the Core API Completion. Any amount withheld by BPPR in accordance with this Section 3.10(a)(i) shall not exceed \$500,000 and will be paid by BPPR promptly following the Core API Completion.
- (ii) Subject to Section 3.10(c), if the Core API Completion has not occurred on or prior to the Core API Deadline, then BPPR shall be entitled to be paid the following amounts from EVERTEC, each of which amounts shall begin accruing with the month immediately following the month in which the Core API Deadline occurs: (x) fifty thousand dollars (\$50,000) for each of the first three (3) months; (y) one hundred thousand dollars (\$100,000) for each of fourth (4th) through sixth (6th) months; and (z) one hundred fifty thousand dollars (\$150,000) for each month thereafter. Any amount which BPPR is entitled to be paid in accordance with this Section 3.10(a)(ii) will be applied as a credit against any amounts due to EVERTEC under this Master Agreement; provided, that after June 30, 2028, and following any termination of this Master Agreement, BPPR may elect to receive such amount as a cash payment rather than a credit in which case EVERTEC will make such cash payment within thirty (30) days of a request from BPPR.

(b) The Popular Parties' rights under this Section 3.10 are in addition to, and shall in no way limit, the Popular Parties' termination rights under Section 9.3(d)(i) in connection with a failure to meet the Core API Deadline. In no event shall the Popular Parties' exercise of its rights under this Section 3.10 or Section 9.3(d)(i) diminish or waive in any manner Popular Parties' rights under the Core API Layer SOW or EVERTEC' s obligations thereunder.

3.11 <u>Phase 2 SOW</u>. Each party shall use commercially reasonable efforts to complete, execute and deliver the Phase 2 SOW as promptly as practicable after the date hereof and in any case within sixty (60) days after the date hereof. All out-of-pocket and internal costs incurred by the parties in connection with the Phase 2 SOW shall be borne fifty percent (50%) by the Popular Parties and fifty percent (50%) by EVERTEC.

### ARTICLE FOUR- DISCLAIMER OF WARRANTIES & LIMITED LIABILITY

- 4.1 <u>DISCLAIMER OF WARRANTIES</u>. EXCEPT AS EXPLICITLY PROVIDED IN THIS MASTER AGREEMENT (INCLUDING IN ANY SERVICE ADDENDUM OR EXHIBIT):
  - (a) THE SERVICES AND ANY EQUIPMENT PROVIDED UNDER THIS MASTER AGREEMENT ARE PROVIDED ON AN "AS IS", "AS AVAILABLE" BASIS;
  - (b) THE PARTIES ACKNOWLEDGE THAT GIVEN THE SERVICES (INCLUDING ANY EQUIPMENT) MAY DEPEND TO SOME EXTENT ON COMPANY, BPPR, AND THEIR RESPECTIVE SUBSIDIARIES' OWN COMPUTER SYSTEMS, EVERTEC DOES NOT MAKE ANY WARRANTIES OF ANY KIND, EITHER EXPRESS OR IMPLIED, INCLUDING WARRANTIES OF TITLE, QUIET ENJOYMENT, QUIET POSSESSION, MERCHANTABILITY, OR FITNESS FOR A PARTICULAR PURPOSE OR NON-INFRINGEMENT.
  - (c) EVERTEC DOES NOT MAKE ANY WARRANTIES OF ANY KIND WITH RESPECT TO LOSS OR CORRUPTION OF DATA, LOSS OR DAMAGE TO EQUIPMENT AND/OR SOFTWARE, SYSTEM RESPONSE TIMES, TELECOMMUNICATION LINES OR OTHER COMMUNICATION DEVICES, QUALITY, AVAILABILITY, RELIABILITY, SECURITY ACCESS DELAYS OR ACCESS INTERRUPTIONS, NOR COMPUTER VIRUSES, BUGS OR ERRORS.
  - (d) EVERTEC DOES NOT MAKE ANY WARRANTIES THAT THE SERVICES WILL NOT BE INTERRUPTED OR ERROR FREE OR AS TO THE RESULTS THAT MAY BE OBTAINED FROM THE USE OF THE SERVICES AND EVERTEC ASSUMES NO RESPONSIBILITY OR LIABILITY IF TELECOMMUNICATION CARRIERS ARE NOT AVAILABLE AT ANY GIVEN TIME.
  - (e) EVERTEC, ITS AFFILIATES, AND THEIR RESPECTIVE REPRESENTATIVES ARE NOT LIABLE, AND EXPRESSLY DISCLAIM ANY LIABILITY FOR THE CONTENT OF ANY DATA OF COMPANY, BPPR, AND THEIR RESPECTIVE SUBSIDIARIES THAT IS TRANSFERRED BY COMPANY, BPPR, AND THEIR RESPECTIVE SUBSIDIARIES TO THE SERVICES PROVIDED BY EVERTEC.

4.2 <u>Reliance on COMPANY Provided Data</u>. In performing the Services, EVERTEC will be entitled to rely upon the data, information, instructions, or Specifications provided by COMPANY, BPPR, and their respective Subsidiaries and, therefore, will not be liable to COMPANY, BPPR, and their respective Subsidiaries in the same accord as set forth herein as a limitation of liability, should EVERTEC perform in accordance with such data, information or instructions received from COMPANY, BPPR, and their respective Subsidiaries. If any error results from incorrect input supplied by COMPANY, BPPR, and their respective Subsidiaries, COMPANY, BPPR, and their appropriate respective Subsidiaries will be responsible for discovering and reporting such error and supplying the data necessary to correct such error to EVERTEC, in which case, EVERTEC will exercise Best Efforts to correct the error at COMPANY, BPPR, and their appropriate respective Subsidiaries' sole expense.

## 4.3 Force Majeure.

- (a) Subject to Section 4.3(b) below, neither party will be liable under this Master Agreement as a result of any delay or failure in the performance of its obligations (except for payment of money) to the extent such performance is prevented, frustrated, hindered or delayed as a result of an extraordinary event that is beyond the party's reasonable control, including strikes, shortages, riots, fires, floods, storms, hurricanes, epidemics, pandemics, earthquake, acts of God, hostilities or other similar events or disasters, and actions taken by Governmental Authorities in response thereto (including any quarantine, 'shelter in place,' 'stay at home,' shut down, closure, sequester or similar Legal Requirement); provided, that the non-performing party is without material fault in causing such delay or failure (a "Force Majeure Event").
- (b) This Section 4.3 exclusion of liability will only apply if, and to the extent, the party experiencing the Force Majeure Event exercises Best Efforts to mitigate and minimize the impact of the event on its obligations and follows its Business Continuity Plan, if applicable. The non-performing party will be excused from performance of the obligations so affected (including any obligations with respect to applicable Service Levels) for as long as such circumstances prevail, and the non-performing party continues to use Best Efforts to recommence performance.
- (c) In the event of an occurrence of a *Force Majeure* Event, the affected party shall give notice of suspension to the other party's designated authorized representative as soon as reasonably practicable after the activation of the Business Continuity Plan or of becoming aware of such *Force Majeure* Event, stating the date and extent of such suspension and the cause thereof, and the affected party shall resume the performance of such obligations as soon as reasonably practicable after the removal of such cause. If a *Force Majeure* Event lasts more than fourteen (14) days, BPPR shall (i) not be required to pay for Services it does not receive; and (ii) be entitled to seek (at BPPR's own expense) an alternative service provider on a temporary basis with respect to any Service(s) that EVERTEC is unable to provide. EVERTEC shall support such temporary transition. COMPANY and BPPR may terminate any Service upon notice to EVERTEC if EVERTEC is unable to provide the Service for more than 30 days as a result of a Force Majeure Event.

4.4 Systems and/or Services Not Provided by EVERTEC. To the extent COMPANY, BPPR, or their respective Subsidiaries perform any services themselves or use their own software, hardware, communications devices, Internet services, e-mail systems or other systems or, in the alternative, retain Third Parties to provide such services and systems, the parties acknowledge and agree that terms of this Master Agreement will not be deemed to impose on EVERTEC any obligation to obtain from owners of such systems any licenses or agreements that are necessary in order for EVERTEC to interface the Services with such systems. Nor will EVERTEC have any responsibility or liability in connection with such services or systems not provided by EVERTEC. COMPANY, BPPR, or their respective Subsidiaries will be solely responsible for the installation, operation, maintenance, use, and compatibility of such systems and services. In the event that such systems or services impair COMPANY, BPPR, and their respective Subsidiaries' use of any Services: (a) COMPANY, BPPR, and their respective Subsidiaries will nonetheless be liable for payment for such Services provided by EVERTEC, and (b) any Specifications generally applicable to the Services will not apply to the extent they are affected by such impairment.

# 4.5 <u>LIMITATION OF LIABILITY</u>.

(a) EXCEPT FOR WILLFUL MISCONDUCT OR GROSS NEGLIGENCE, EVERTEC AND ITS SUBSIDIARIES SHALL NOT BE LIABLE TO COMPANY, BPPR, OR ANY OF THEIR RESPECTIVE AFFILIATES PURSUANT TO THIS MASTER AGREEMENT FOR (1) ANY SPECIAL, EXEMPLARY, PUNITIVE, CONSEQUENTIAL, REMOTE OR SPECULATIVE DAMAGES OR LOSSES, INCLUDING LOSS OF PROFITS OR ANTICIPATED PROFITS, OR ROYALTIES, INCLUDING ANY SUCH DAMAGES OR LOSSES THAT ARISE IN CONNECTION WITH ANY LOST DATA, OR (2) ANY OTHER BUSINESS OR OTHER ECONOMIC LOSS ARISING FROM OR RELATED TO ANY EQUIPMENT OR SOFTWARE OR ANY SERVICES, INCIDENTAL OR OTHERWISE, PROVIDED BY THIRD PARTIES (EXCEPT THOSE SERVICES PROVIDED BY A CONTRACTOR OR A SUBCONTRACTOR OF EVERTEC UNDER SECTION 1.16), OR (3) ANY THIRD PARTY CLAIM (EXCEPT AS OTHERWISE PROVIDED IN ARTICLE SEVEN, ARTICLE TEN, OR ANY OTHER INDEMNIFICATION OBLIGATIONS IN THIS MASTER AGREEMENT), IN EACH CASE (I) WHETHER FOR, AMONG OTHER THINGS, SUCH PARTY'S NEGLIGENCE OR MISCONDUCT, BREACH OF WARRANTY OR ANY OBLIGATION ARISING THEREFROM; (II) WHETHER LIABILITY IS ASSERTED IN, AMONG OTHER THINGS, CONTRACT OR TORT (INCLUDING NEGLIGENCE AND STRICT PRODUCT LIABILITY); (III) WHETHER OR NOT FORESEEABLE; AND (IV) WHETHER OR NOT SUCH PARTY HAS BEEN ADVISED OF THE POSSIBILITY OF ANY SUCH LOSS OR DAMAGE.

(b) EXCEPT FOR WILLFUL MISCONDUCT OR GROSS NEGLIGENCE, COMPANY, BPPR, AND THEIR RESPECTIVE SUBSIDIARIES SHALL NOT BE LIABLE TO EVERTEC OR ITS AFFILIATES PURSUANT TO THIS MASTER AGREEMENT FOR (1) ANY SPECIAL, EXEMPLARY, PUNITIVE, CONSEQUENTIAL, REMOTE OR SPECULATIVE DAMAGES OR LOSSES, INCLUDING LOSS OF PROFITS OR ANTICIPATED PROFITS, OR ROYALTIES, INCLUDING ANY SUCH DAMAGES OR LOSSES THAT ARISE IN CONNECTION WITH ANY LOST DATA; (2) ANY OTHER BUSINESS OR OTHER ECONOMIC LOSS, IN EACH CASE, ARISING FROM OR RELATED TO ANY EQUIPMENT OR SOFTWARE NOT PROVIDED BY COMPANY, BPPR, OR THEIR

RESPECTIVE SUBSIDIARIES, OR ANY SERVICES, INCIDENTAL OR OTHERWISE, PROVIDED BY THIRD PARTIES (EXCEPT THOSE SERVICES PROVIDED BY A CONTRACTOR OR SUBCONTRACTOR OF COMPANY, BPPR OR THEIR SUBSIDIARIES), OR (3) ANY THIRD PARTY CLAIM (EXCEPT AS OTHERWISE PROVIDED IN <u>ARTICLE TEN</u> OR ANY OTHER INDEMNIFICATION OBLIGATIONS IN THIS MASTER AGREEMENT), IN EACH CASE, (I) WHETHER FOR, AMONG OTHER THINGS, SUCH PARTY'S NEGLIGENCE OR MISCONDUCT, BREACH OF WARRANTY OR ANY OBLIGATION ARISING THEREFROM; (II) WHETHER LIABILITY IS ASSERTED IN, AMONG OTHER THINGS, CONTRACT OR TORT (INCLUDING NEGLIGENCE AND STRICT PRODUCT LIABILITY); (III) WHETHER OR NOT FORESEEABLE; AND (IV) WHETHER OR NOT SUCH PARTY HAS BEEN ADVISED OF THE POSSIBILITY OF ANY SUCH LOSS OR DAMAGE.

- (c) EXCEPT FOR WILLFUL MISCONDUCT OR GROSS NEGLIGENCE OR TO THE EXTENT OTHERWISE PROVIDED UNDER ANY APPLICABLE LEGAL REQUIREMENTS, EACH PARTY'S LIMIT OF LIABILITY UNDER THIS MASTER AGREEMENT WILL BE AN AGGREGATE ANNUAL LIMIT EQUAL TO THE AMOUNT OF PAYMENTS MADE TO EVERTEC BY COMPANY, BPPR, AND THEIR RESPECTIVE SUBSIDIARIES FOR THE SERVICE FOR WHICH THE LIABILITY RELATES DURING THE TWELVE MONTHS PRIOR TO THE ACT, OMISSION OR EVENT THAT GIVES RISE TO THE CLAIM FOR LIABILITY (THE "GENERAL LIABILITY CAP"). NOTWITHSTANDING THE FOREGOING, FOR (X) LOSSES RELATED TO BREACHES OF CONFIDENTIALITY, PRIVACY AND DATA PROTECTION OBLIGATIONS UNDER THIS MASTER AGREEMENT AND THE ATH NETWORK AGREEMENT AND (Y) BPPR DATA BREACH COSTS (AS DEFINED IN THE ISO AGREEMENT), EACH PARTY'S LIMIT OF LIABILITY WILL BE AN AGGREGATE ANNUAL LIMIT EQUAL TO THE GREATER OF (I) THE APPLICABLE GENERAL LIABILITY CAP MULTIPLIED BY FIVE, AND (II) \$50,000,000 (AS APPLICABLE, THE "DATA CAP").
- (d) NOTHING IN THIS MASTER AGREEMENT SHALL EXCLUDE OR LIMIT EITHER PARTY'S LIABILITY: (I) FOR DEATH OR PERSONAL INJURY CAUSED BY SUCH PARTY'S NEGLIGENCE; OR (II) FOR INDEMNIFICATION OBLIGATIONS FOR CLAIMS OF INTELLECTUAL PROPERTY INFRINGEMENT PURSUANT TO <u>ARTICLE SEVEN</u> OF THIS MASTER AGREEMENT EXCEPT TO THE EXTENT SUCH A CLAIM ARISES DUE TO ACTS OR OMISSIONS OF A THIRD PARTY UNDER AN EVERTEC PASS THROUGH LICENSE, IN WHICH CASE EVERTEC'S LIABILITY TO BPPR OR COMPANY OR ANY OF THEIR RESPECTIVE AFFILIATES SHALL NOT EXCEED THE AMOUNTS ALLOCATED TO COMPANY AND BPPR PURSUANT TO <u>SECTION 7.4(C)</u>.
- (e) EACH PARTY AGREES THAT (1) AMOUNTS PAID IN CONNECTION WITH THE DEFENSE OF THIRD PARTY CLAIMS, (2) AMOUNTS PAYABLE TO A THIRD PARTY PURSUANT TO COURT ORDER OR A SETTLEMENT ENTERED INTO IN ACCORDANCE WITH THE PROVISIONS OF THIS MASTER AGREEMENT; AND (3) FINES OR PENALTIES PAID TO A GOVERNMENTAL AUTHORITY, IN EACH CASE, IN CONNECTION WITH INDEMNIFICATION OBLIGATIONS WITH RESPECT TO THIRD PARTY CLAIMS UNDER ARTICLES SEVEN AND TEN ARE DEEMED TO BE DIRECT DAMAGES AND ARE NOT SUBJECT TO THE EXCLUSIONS IN <u>SECTIONS 4.5(A)</u> AND 4.5(B).

(f) NOTWITHSTANDING ANY OTHER PROVISION OF THIS MASTER AGREEMENT, EVERTEC SHALL BE RESPONSIBLE FOR ALL REASONABLE AND CUSTOMARY OUT OF POCKET COSTS AND EXPENSES INCURRED BY COMPANY AND BPPR AND THEIR RESPECTIVE SUBSIDIARIES ARISING OUT OF OR IN CONNECTION WITH ANY BREACH OF THE CONFIDENTIALITY OR PRIVACY OR DATA PROTECTION PROVISIONS OF THIS MASTER AGREEMENT, OR CAUSED BY EVERTEC'S FRAUD, WILLFUL MISCONDUCT OR GROSS NEGLIGENCE ("REASONABLE AND CUSTOMARY COSTS"); PROVIDED THAT SUCH COSTS SHALL BE SUBJECT TO THE DATA CAP AS SET FORTH IN SECTION 4.5(C) ABOVE. REASONABLE AND CUSTOMARY COSTS MAY INCLUDE THE FOLLOWING COSTS AND EXPENSES ASSOCIATED WITH ADDRESSING AND RESPONDING TO AN INCIDENT AND EACH SHALL BE TREATED AS DIRECT DAMAGES UNDER THIS SECTION 4.5: (I) PREPARATION AND MAILING OR OTHER TRANSMISSION OF NOTIFICATIONS TO AFFECTED INDIVIDUALS; (II) PREPARATION AND MAILING OR OTHER TRANSMISSION OF ANY OTHER COMMUNICATIONS TO CUSTOMERS, AGENTS, REGULATORS, LAW ENFORCEMENT AGENCIES OR OTHERS; (III) ESTABLISHMENT OF A CALL CENTER OR OTHER COMMUNICATIONS PROCEDURES IN RESPONSE TO SUCH VIOLATION (E.G., CUSTOMER SERVICE FAQS, TALKING POINTS AND TRAINING); (IV) COMMERCIALLY REASONABLE PUBLIC RELATIONS AND OTHER SIMILAR CRISIS MANAGEMENT SERVICES; (V) REASONABLE LEGAL AND ACCOUNTING FEES AND EXPENSES ASSOCIATED WITH INVESTIGATION OF AND RESPONSE TO SUCH EVENT; (VI) COSTS FOR COMMERCIALLY REASONABLE CREDIT REPORTING, CREDIT MONITORING, IDENTITY PROTECTION OR OTHER SIMILAR PRODUCTS AND SERVICES; (VII) CONDUCTING FORENSIC REVIEWS TO IDENTIFY THE CAUSE AND IMPACT OF THE INCIDENT, TO THE EXTENT COMPANY OR BPPR IS PERMITTED TO DO SO UNDER THIS MASTER AGREEMENT; (VIII) ADDITIONAL COMPUTER PROGRAMMING REQUIRED TO IDENTIFY OR PROTECT AFFECTED INDIVIDUALS OR OTHERWISE AS A RESULT OF THE INCIDENT; (IX) REMEDIATION ACTIVITIES REQUIRED BY A GOVERNMENTAL OR REGULATORY AUTHORITY; (X) CHARGES AND FINES ASSESSED OR IMPOSED BY PAYMENT CARD COMPANIES, INCLUDING UPON THE DATA SUBJECTS WHOSE PERSONAL INFORMATION WAS COMPROMISED AS A RESULT OF THE BREACH; (XI) GOVERNMENT FINES OR PENALTIES IMPOSED BY A GOVERNMENTAL OR REGULATORY AUTHORITY.

(g) Each party hereby waives any claim that this <u>Section 4.5</u> deprives it of an adequate remedy or cause this Master Agreement to fail of its essential purpose.

## ARTICLE FIVE- CONFIDENTIALITY, PRIVACY & DATA PROTECTION

### 5.1 Confidential Information.

(a) The parties acknowledge that in the course of their dealings each may receive (the "Receiving Party") Confidential Information of the other party (the "Disclosing Party"). As such, the parties are willing to share such Confidential Information provided that the Receiving Party protects the Confidential Information of the Disclosing Party pursuant to the terms of this Master Agreement, and the Disclosing Party will remain the proprietor of the Confidential Information and any derivative thereof.

- (b) Confidential Information (other than Intellectual Property and Personal Data, which will always be deemed to be Confidential Information) will not include information that the Receiving Party can reasonably demonstrate:
  - (1) is or becomes generally available to the public without breach of any obligation of confidentiality under this Master Agreement or any confidentiality agreement that the Receiving Party may have with a Third Party;
  - (2) was already known to or was rightfully in the possession of the Receiving Party on a non-confidential basis prior to its disclosure by the Disclosing Party;
  - (3) becomes available to the Receiving Party from a Third Party, <u>provided</u> that such Third Party is not subject to an obligation of confidentiality with the Disclosing Party;
  - (4) is independently developed by the Receiving Party without reference to or reliance upon the Confidential Information of the other Party;
  - (5) is approved in writing by the Disclosing Party for disclosure; or
  - (6) is required to be disclosed by applicable Legal Requirements or by an applicable Governmental Authority, but only to the extent so required and solely for such purpose, and the Receiving Party shall otherwise remain obligated to treat such information as Confidential Information pursuant to this <u>Article Five</u>.
- (c) In any dispute with respect to these exclusions, the burden of proving that information is not Confidential Information will be on the party making such assertion.

### 5.2 Protection of Confidential Information.

(a) The Receiving Party agrees to protect and hold all Confidential Information in strict confidence and protect all Confidential Information from unauthorized and/or inadvertent access, use, destruction or disclosure in full compliance with this Master Agreement and all applicable Legal Requirements and Industry Standards, including by implementing, maintaining and enforcing appropriate data and Systems security.

(b) Unless in receipt of a specific written exemption from the Disclosing Party or required by applicable Legal Requirements, the Receiving Party will not:

- (1) use, reproduce, disseminate, modify, retain, disclose or perform Data Processing of any of the Confidential Information for any purpose other than to perform its obligations under this Master Agreement for which the Confidential Information is being disclosed, or for such Receiving Party's internal recordkeeping purposes;
- (2) reverse engineer, sell or market any Confidential Information;
- (3) receive and maintain the Confidential Information without implementing reasonable internal controls and procedures to safeguard the security and confidentiality of such information;
- (4) disclose any of the Confidential Information other than to its Representatives who: (i) have a reasonable need-to-know in order to discharge their obligations under this Master Agreement, (ii) have confidentiality agreements substantially similar to those herein or owe a duty of confidentiality to the Receiving Party that survive the Representatives' engagement, (iii) have received appropriate notice and training regarding the Confidential Information and the Data Protection Measures, and (iv) have not been indicted and/or convicted of fraud, theft, larceny, embezzlement, any other financial or cybercrime or a crime involving moral turpitude; or
- (5) remove any proprietary rights or confidentiality legend from the Confidential Information.
- (c) The prohibition against the disclosure of Confidential Information includes disclosing the substance of the negotiations of this Master Agreement and the existence and/or the terms and conditions thereof, as well as the fact that any similarity exists between the Confidential Information and information independently developed by another Person or entity, and the parties understand that such similarity does not excuse it from abiding by its covenants or other obligations under this Master Agreement.
- (d) The Receiving Party will be fully liable for the acts and omissions of its Representatives to whom it discloses the Confidential Information.

### 5.3 Security of Personal Data.

(a) To effect the purposes of this Master Agreement, COMPANY, BPPR, or one of their respective Subsidiaries may from time to time provide EVERTEC with Personal Data in order for EVERTEC to provide the Services. EVERTEC shall comply with all Legal Requirements (including Legal Requirements regarding the protection, privacy and confidentiality of Personal Data) applicable to EVERTEC's use, storage and other processing (including Data Processing) of Personal Data.

- (b) EVERTEC agrees to take commercially reasonable actions in accordance with applicable Industry Standards to protect the confidentiality, integrity, availability and resilience of its Systems and to ensure that availability of and access to Personal Data can be restored in a timely manner after an Incident. EVERTEC's responsibilities in this regard include regularly assessing and evaluating, in accordance with applicable Industry Standards and Legal Requirements, the effectiveness of the technical and organizational measures adopted by it. In furtherance thereof, EVERTEC shall be responsible for establishing, maintaining and implementing a Data Protection Program.
- (c) In furtherance, and not in limitation, of its other obligations under this Master Agreement, and to maintain the protection and privacy of the Personal Data, EVERTEC shall: (i) not disclose any Personal Data to any Third Party other than Data Subcontractors appointed pursuant to Section 1.16; (ii) limit access to the Personal Data to its employees, Representatives and Data Subcontractors who have a need-to-know such Personal Data after adequately informing such Persons of the confidential nature of the Personal Data and in each case subject to appropriate Role-Based Access Controls (which controls (A) will be based on least privilege access principles; (B) EVERTEC will document and provide to COMPANY; (C) will be subject to COMPANY approval if they relate to access to BPPR Infrastructure; and (D) will give COMPANY approval rights over individual authorizations if they relate to access to BPPR Infrastructure); (iii) use the Personal Data only for purposes of carrying out its obligations hereunder; (iv) give prompt notification to BPPR or COMPANY, as applicable, of any unauthorized or inadvertent actual or reasonably suspected disclosure of Personal Data or of actual or reasonably suspected breaches involving the disclosure of Personal Data, as provided in Section 5.3(e); (v) upon request of COMPANY, BPPR, or one of their respective Subsidiaries promptly return or destroy all Personal Data, including all copies thereof except for copies necessary to comply with applicable Legal Requirements and records retention, including the requirements in Section 6.4 (provided that following completion of such request EVERTEC shall not be required to provide any Services that require the use of such Personal Data); and (vi) implement all reasonable security measures in accordance with applicable Industry Standards designed to (a) ensure the security, integrity and confidentiality of Personal Data, (b) protect against any anticipated threats or hazards to the security or integrity of Personal Data, and (c) protect against

### (d) Data Protection Program.

- (i) EVERTEC acknowledges that EVERTEC and the Popular Parties are subject to Legal Requirements and other guidelines regarding the implementation of security measures to safeguard the security and privacy of Personal Data.
- (ii) Without limiting Section 5.2, EVERTEC represents and warrants to COMPANY, BPPR, and their respective Subsidiaries that it has in place, and hereby undertakes to, at all times during the Term, maintain and implement the following in accordance with Industry Standards (the "Data Protection Measures"): (A) a Data Protection Program that includes administrative, technical and physical safeguards to protect the security, privacy, confidentiality, availability and integrity of Personal Data in accordance with applicable Legal Requirements and Industry Standards; (B) an Incident Response Program that provides for the Breach Notification Procedures set forth in Section 5.3(e) below; and (C) a vulnerability monitoring and patching program.

- (iii) With respect to all Services that EVERTEC provides within BPPR Infrastructure, such Data Protection Measures shall also satisfy, at a minimum, the Minimum Data Safeguards set forth in Exhibit G. With respect to all Services that EVERTEC provides outside BPPR Infrastructure, such Data Protection Measures shall also satisfy, at a minimum, the EVERTEC Measures. EVERTEC represents and warrants that it (A) has made available to COMPANY and BPPR via on-site review or other secure means of transmission its Data Protection Program, Incident Response Program and descriptions of its other Data Protection Measures (collectively, the "EVERTEC Measures"); (B) will comply with these programs and measures and (C) will upon request provide the same for review on-site or via other secure method agreed by the parties.
- (iv) EVERTEC shall not make any changes to its Data Protection Program, Incident Response Program or other Data Protection Measures in any manner that would degrade or weaken the controls thereunder, or that would otherwise have an adverse effect on the protections provided to BPPR Data, without the prior written consent of COMPANY or BPPR.
  - (e) Incident Response Program.
- (i) Without limiting Section 5.2(a), EVERTEC represents and warrants to COMPANY and BPPR that it has implemented, and hereby undertakes to maintain and update, a comprehensive response program that includes processes and procedures (including notification procedures to Affected Persons) designed to address incidents of unauthorized access involving Personal Data in accordance with applicable Legal Requirements and Industry Standards.
- (ii) If EVERTEC becomes aware of an Incident, to the extent it is not prohibited by applicable Legal Requirements, EVERTEC agrees to:
  - (1) notify COMPANY and BPPR as soon as practicable, and in any event within twenty-four (24) hours, following EVERTEC's becoming aware of an Incident. The initial notification must be made to the BPPR Chief Information Security Officer, and must include based on then-available information: (i) the available facts; (ii) the status and results of the investigation including identifying the sources and underlying causes of the Incident; (iii) the estimated effects on COMPANY, BPPR, BPPR Data and the Services; (iv) the steps already taken to mitigate, remedy and contain the Incident; (v) if applicable, the potential number of affected customers, consumers, and/or employees affected (collectively "Affected Persons"); and (vi) the name and contact information of EVERTEC's representative who will serve as COMPANY's and BPPR's primary contact;

- (2) provide COMPANY and BPPR with prompt updates to any information referenced in clause (1) above;
- (3) take all reasonable steps, at EVERTEC's sole cost and expense (unless such Incident was caused by COMPANY, BPPR, any of their subsidiaries or any contractor or subcontractor thereof), in accordance with its Data Protection Program and Incident Response Program to immediately mitigate and/or remedy, at EVERTEC's expense, such Incident and contain its further occurrence:
- (4) if the Incident was caused by EVERTEC's or its Representative's breach of this Master Agreement or gross negligence, fraud or willful misconduct, recover and reconstruct BPPR Data lost or compromised in the Incident; provided that costs incurred in connection with such recovery and reconstruction shall be subject to the Data Cap;
- reasonably coordinate and cooperate with COMPANY and BPPR representatives in any required investigation and provide periodic updates regarding the Incident, including: (i) providing access to the affected facilities and affected operations as necessary for the Popular Parties or their auditors to conduct investigations required to be conducted by the Popular Parties by Legal Requirements (provided that subject to Section 4.5 COMPANY or BPPR, as applicable, shall reimburse EVERTEC for any Losses incurred by EVERTEC in connection with any unauthorized access or intrusion to EVERTEC's Systems, or any Disabling Device introduced to EVERTEC's Systems, as a result of a breach of this Master Agreement, failure to comply with applicable EVERTEC Security Requirements, gross negligence, fraud or willful misconduct of COMPANY or BPPR in connection with its access to EVERTEC's Systems pursuant to this Section 5.2(e)(ii)(5); (ii) facilitating interviews with EVERTEC's Personnel with knowledge of the Incident; and (iii) making available all relevant records, logs, files, data reporting, forensics or audit reports and other materials required by COMPANY or BPPR to allow COMPANY and BPPR to comply with applicable Legal Requirements, provided that EVERTEC will not be required to provide COMPANY or BPPR with information belonging to, or compromising the security of, EVERTEC or its other customers;
- (6) provide reasonable cooperation with COMPANY and BPPR in any litigation, Governmental Authority inquiries, or other Third Party action arising out of or resulting from the Incident and deemed necessary by COMPANY or BPPR to protect the BPPR Data and comply with applicable Legal Requirements, at the expense of COMPANY and BPPR unless such Incident was caused by EVERTEC's or its Representative's gross negligence, fraud, willful misconduct or breach of this Master Agreement;

- except as may be required by applicable Legal Requirements, not communicate with any Third Party other than EVERTEC's Representatives, including the media, vendors, and Affected Persons, regarding the Incident (as it relates to COMPANY or BPPR) without COMPANY's or BPPR's express written consent and approval of the content of the communication. EVERTEC agrees that regarding the impact of the Incident on COMPANY or BPPR, COMPANY and BPPR will have the sole right to determine: (i) whether notice of the Incident is to be provided to any individual, Governmental Authority, law enforcement agency, consumer reporting agencies or others as required by applicable Legal Requirements or otherwise; (ii) the contents of such notice; and (iii) whether any type of remediation may be offered to Affected Persons (including any free credit monitoring service), and the nature and extent of any such remediation;
- (8) report criminal acts as and to the extent required by applicable Legal Requirements and Governmental Authorities and notify COMPANY and BPPR as soon as practicable prior to such reporting; and
- (9) maintain and preserve all relevant documents, records, and other data related to any Incident involving BPPR Data for the period required by COMPANY or BPPR.
- (f) Data Processing. The parties acknowledge that for the purposes of applicable Legal Requirements, BPPR is the Controller and EVERTEC is the Processor of Personal Data. Personal Data must be processed by EVERTEC in compliance with BPPR's documented instructions that are provided in accordance with this Master Agreement or any applicable SOW or Schedule ("BPPR Instructions") and in accordance with this Master Agreement and applicable Legal Requirements. If EVERTEC reasonably believes, based on the advice of counsel, that any BPPR Instruction violates applicable Legal Requirements, EVERTEC shall so notify BPPR. In accordance with BPPR Instructions, EVERTEC shall promptly notify BPPR if EVERTEC receives a request from any Person to exercise their rights under the applicable Legal Requirements including the right of access, rectification, restriction of Data Processing, erasure (such as the right to be forgotten), data portability, or objection to any Data Processing. EVERTEC shall use Best Efforts to provide assistance to BPPR, at BPPR's cost, in responding to such requests within the time specified under the applicable Legal Requirements.
- (g) Upon request, EVERTEC shall promptly provide to COMPANY, BPPR, and any internal or external auditor reasonably designated by COMPANY or BPPR, under applicable confidentiality obligations, (i) a copy of the aforementioned Data Protection Measures and (ii) reasonable documentation as to EVERTEC's compliance with BPPR Instructions.

- (h) For the sake of clarity, the parties acknowledge and agree that to the extent changes to EVERTEC's Data Protection Measures are required due to changes in Legal Requirements that are applicable to COMPANY, BPPR or their Subsidiaries but not to EVERTEC, such changes shall be considered Mandatory Enhancements pursuant to Section 2.8 and the costs shall be allocated accordingly.
- (i) Without limiting EVERTEC's other notice obligations under this Master Agreement, EVERTEC will notify the Chief Information Security Officer at COMPANY and BPPR at InfoSecIncident@popular.com with copy to jeff.foltz@popular.com and by phone at 787-722-3492 (or any replacement contact information provided in writing by COMPANY or BPPR to EVERTEC): (A) as soon as possible after determining a computer-security incident (as defined in 12 C.F.R. 225.301) has materially disrupted or degraded, or is reasonably likely to materially disrupt or degrade, any System for four or more hours or (B) as otherwise required by Legal Requirements.

## 5.4 Required Disclosures.

- (a) In the event that the Receiving Party or any Representative of the Receiving Party is required by any Legal Requirement or any action, proceeding, subpoena, investigation, inquiry or audit, or if the Receiving Party or any Representative of the Receiving Party reasonably believes, upon the advice of counsel, that it is required by any Legal Requirement or any action, proceeding, subpoena, investigation, inquiry or audit to disclose Confidential Information, the Receiving Party will, to the extent allowable under applicable Legal Requirements, give immediate verbal and written notification of the Legal Requirement or action, proceeding, subpoena, investigation, inquiry or audit to the Disclosing Party and, to the extent allowable under applicable Legal Requirements and at the expense of the Disclosing Party, hold the Confidential Information confidential while the Disclosing Party seeks a protective order.
- (b) In addition to the foregoing, a Receiving Party or its Representatives may provide the Confidential Information to a Governmental Authority with supervisory authority over such Receiving Party or its Affiliates if (i) such information is required under Legal Requirements to be filed with or disclosed to any such Governmental Authority, (ii) such information is requested by such a Governmental Authority or the Receiving Party determines based on advice of counsel (which may be in-house counsel) that it is necessary or appropriate to provide such Confidential Information to a Governmental Authority, (iii) the Receiving Party's counsel (which may be in-house counsel) advises that such disclosure is required in order to comply with applicable Legal Requirements or (iv) such Governmental Authority is a banking regulator; provided in each case, that the Receiving Party shall have, where applicable, taken such reasonable steps to protect the confidentiality of such information as the Receiving Party takes with respect to the protection of its own comparable confidential information in such circumstances.
- 5.5 <u>Remedies</u>. The Receiving Party acknowledges and agrees that it would be difficult to fully compensate the Disclosing Party for damages resulting from the breach or threatened breach of the foregoing provisions and, accordingly, that, in addition to any other remedies that may be available, in law, at equity or otherwise, the Disclosing Party will be entitled to seek injunctive relief, including temporary restraining orders, preliminary injunctions and permanent injunctions, to enforce such provisions without the necessity of proving actual damages or posting

a bond or any other security. This provision with respect to injunctive relief will not, however, diminish the Disclosing Party's right to claim and recover damages. Notwithstanding anything in this Master Agreement to the contrary, nothing shall limit or restrict the ability of a Receiving Party to disclose Confidential Information to banking regulators having jurisdiction over such Receiving Party.

- 5.6 <u>Term of Obligation</u>. Unless otherwise agreed in writing, the parties' obligations under this <u>Article Five</u> will survive this Master Agreement for a period of three (3) years following termination hereof. Upon termination of this Master Agreement for any reason, the Receiving Party's rights to possession and use of any Confidential Information in connection with the performance of its obligations hereunder or otherwise will terminate.
- 5.7 <u>Destruction</u>. Upon the Disclosing Party's request, the Receiving Party will promptly destroy in a secure manner following Industry Standards all Confidential Information and all copies thereof, including physical copies and copies embedded in computer files, extracts from computer files and analyses or other material based upon the Confidential Information (provided that following such request the Receiving Party shall not be required to perform any obligations under this Master Agreement that require the use of such Confidential Information). Should the Disclosing Party request evidence of the destruction of any such Confidential Information, the Receiving Party will promptly (and in all cases within thirty (30) days of such request or such other time as agreed to between the parties; provided that EVERTEC shall use Best Efforts to provide within ten (10) days) submit an attestation executed by an authorized officer of the receiving party certifying that any such Confidential Information has been destroyed. If COMPANY or BPPR receives an individual's request to delete Personal Data, COMPANY or BPPR may direct EVERTEC to delete such information (the "<u>Direction</u>"), in which case, EVERTEC shall delete such information without undue delay and certify to COMPANY or BPPR, as applicable, in writing that it has complied with the Direction (and following completion of such request EVERTEC shall not be required to provide any Services that require the use of such Personal Data). Notwithstanding the foregoing, the Receiving Party may, after informing the Disclosing Party, retain copies of the Confidential Information: (i) to comply with applicable Legal Requirements; or (ii) for purposes of record retention, backup records or audit requirements, including as set forth in <u>Section 6.4</u>); provided, however, that the confidentiality obligations (including destruction and certification) hereunder will survive with respect to any retained Confidential Information.

### ARTICLE SIX- SECURITY, BPPR DATA & RECORDS

6.1 <u>Authorized Persons</u>. Each party will designate one or more individuals (hereinafter, "<u>Authorized Persons</u>") so that as to the specific Service, the Authorized Person can (1) coordinate the Services, Data Processing or other transactions on behalf of such party; (2) receive information from the other party related to the operation of the Service, including any COMPANY-, BPPR- or EVERTEC-provided access code; and (3) give written instructions or inform the other party about any action or request for action by such party.

#### 6.2 Security Measures.

- (a) The parties warrant that they have adopted, and will assume responsibility for complying with, security measures reasonably required (including under applicable Industry Standards and Legal Requirements) for the protection of access to their Systems and to the Services by their Representatives and Authorized Persons. As such, the parties warrant that they have established a Data Protection Program in accordance with Legal Requirements to minimize the risk of unauthorized access and agree that they will take the necessary measures to maintain the confidentiality of the security procedures and any access codes, passwords, instructions or security equipment.
- (b) EVERTEC will monitor, review and update the Data Protection Measures no less than annually to ensure compliance with this Master Agreement. EVERTEC agrees that, as may be relevant to its Services, the Data Protection Measures will, among other things: (i) take into account audit results, privacy and security risks, threats or changes in applicable Legal Requirements and Industry Standards; and (ii) be timely corrected with respect to any element thereof which could reasonably be expected to pose a threat to the protection of BPPR Data. During the Term, EVERTEC will not modify any component of the Data Protection Measures or discontinue any Data Safeguards, in each case in a manner that would degrade or adversely impact the protections in place with respect to BPPR Data, without COMPANY's or BPPR's consent.
- (c) EVERTEC and its Affiliates (as applicable) shall at all times (x) use Best Efforts to maintain their Systems free from all Disabling Devices that may disrupt, damage or interfere with COMPANY's, BPPR's, and their respective Subsidiaries' (as applicable) network and/or telecommunication facilities and (y) act in accordance with Industry Standards and the provisions of this Agreement to prevent any Incident. EVERTEC and its Affiliates (as applicable) agree to (1) perform penetration tests annually and as otherwise required by the Data Protection Program of EVERTEC's or any of its Affiliates' (as applicable) networks and Systems where BPPR Data is processed, transmitted or stored and (pursuant to the frequency established in the Data Protection Program) web-applications through which BPPR Data is accessed (collectively "Penetration Tests"); and, within thirty (30) days following completion of each Penetration Test, share a summary of any such tests that are relevant to the Popular Parties (provided that (i) disclosure of such results by EVERTEC shall be subject to, and limited by, applicable Legal Requirements; (ii) EVERTEC may redact from such results information relating solely to EVERTEC's internal business or solely to EVERTEC's other customers, provided that this clause (ii) shall not give EVERTEC any right to redact any information relating to EVERTEC's provision of the Services to COMPANY, BPPR, or their respective Subsidiaries; and (iii) if the critical vulnerability is identified during the Penetration Test EVERTEC will notify COMPANY and BPPR of the vulnerability within fifteen (15) days); (2) maintain an alert status regarding the security of their computing Systems, including all vulnerabilities and security patches or corrective actions, by subscribing to an industry-recognized service, such as CERT or CIAC; (3) permit COMPANY and BPPR to review EVERTEC's patch and vulnerability programs in connection with an Audit under Section 8.4; and (4) provide COMPANY and BPPR with written certification documentation showing the remediation status of any security issues identified in such assessments on a monthly basis for critical and high risk issues and upon request from COMPANY or BPPR for medium and low risk issues. Should EVERTEC's or any of its Affiliates' (as applicable) assessment reveal inappropriate or inadequate security of their Systems that are used to store BPPR

Data or otherwise provide Services based on the applicable Industry Standard requirements for security, without compensating controls, COMPANY, BPPR, or any of their respective Subsidiaries (as applicable) may, in addition to any other remedies they may have, remove EVERTEC's or any of its Affiliates' (as applicable) access to COMPANY's, BPPR's, or any of their respective Subsidiaries' (as applicable) own Systems until EVERTEC or its Affiliates (as applicable) comply with the standards set forth in this Section 6.2(c) and provide reasonable assurance (including reasonably requested evidence) of such compliance. Additionally, EVERTEC and its Affiliates will permit COMPANY and BPPR or their designated Third Party to conduct Penetration Tests on network components and Systems used exclusively for COMPANY or BPPR and that are located in COMPANY or BPPR's service network within EVERTEC, subject to the provisions set forth on Schedule 6.2(c).

- (d) If a network connection is established between any System of COMPANY, BPPR, or one of their respective Subsidiaries, on the one hand, and the Systems of EVERTEC, or one of its Affiliates, on the other hand, COMPANY, BPPR, and their respective Subsidiaries (as applicable) shall at all times (i) use Best Efforts to maintain their computing environment connected to EVERTEC's or its Affiliates' (as applicable) Systems free from all Disabling Devices and (ii) act in accordance with this Agreement and Industry Standards to prevent any Incident, in each case, that may disrupt, damage or interfere with EVERTEC's or its Affiliates' (as applicable) Systems (including complying with all applicable EVERTEC Security Requirements). If EVERTEC reasonably suspects that any System of COMPANY, BPPR or their respective Subsidiaries is the source of any Disabling Device, unauthorized access or other Incident that may reasonably be expected to adversely impact EVERTEC or EVERTEC's other customers in any material respect, EVERTEC shall be entitled to suspend connectivity to such System until the issue has been mitigated; provided that EVERTEC promptly notifies COMPANY and BPPR of the issue in advance and provides a reasonable opportunity for them to remediate the issue unless EVERTEC reasonably determines that immediate suspension is required in order to prevent material harm to EVERTEC or its other customers. EVERTEC shall not be responsible for the performance of Services to the extent they are impacted by any such suspension permitted under this Section 6.2(d).
- (e) Each party's and their respective Affiliates' Authorized Persons agree to comply with all of the other party's requirements in relation to the security of their respective computing environments and Authorized Locations, including any subsequently agreed security plan or information processing requirements that may be embodied in any Service Addendum. Each party and their appropriate respective Affiliates will execute all documents generally required by the other party for access to such other party's computing environment and Authorized Locations. Further, if any Authorized Person of any party or their respective Affiliates, at any time during the Term, is granted remote access or connection to the other party's network, or is telecommuting in any capacity, then such person will be subject to such other party's applicable additional data security requirements.
- (f) Should the Services require access codes or other identification methods to gain access, each party will immediately notify the appropriate other party or parties in writing of any change of Authorized Person or the scope of his/her authority. Until such notification is received, each party may accept, without further inquiry, all declarations, instructions or representations made or issued by the Authorized Person. Furthermore, the parties will not assume responsibility, explicitly or implicitly, for questioning or verifying with the other parties whether the Person who uses or has access to the Service is in fact the Authorized Person or if he/she is acting in accordance with another party's internal policies and procedures.

#### (g) Data Centers

- (i) EVERTEC will perform the Services at all times during the Term from the Data Center locations and with the hosting or collocation Data Subcontractors that have been disclosed to COMPANY and BPPR as of the Effective Date, in accordance with <u>Section 1.16</u>, or in an applicable subsequent Service Addendum.
- (ii) EVERTEC may not change Data Center locations or any hosting or collocation Subcontractor without providing notice to COMPANY and BPPR in accordance with <u>Section 1.16</u>.
- (iii) Notwithstanding anything to the contrary in this Master Agreement, EVERTEC may not host or otherwise store BPPR Data at any Data Center outside the United States.
- (iv) EVERTEC shall bear the cost of any incremental expenses incurred as a result of a change in Data Center location or Subcontractor that has not been requested by COMPANY or BPPR.
- (h) If any software licensed by COMPANY, BPPR or their Subsidiaries that EVERTEC is required to host or operate as part of the Services is no longer supported and EVERTEC reasonably determines that continuing to host or operate such software poses a material security risk to any System of EVERTEC, EVERTEC shall promptly notify the Popular Parties of the same and the parties will work together in good faith to develop and execute a plan to remediate the security risk (which may include upgrading the software, segregating Systems or removing the affected Services from the scope of this Master Agreement).

# 6.3 Ownership of BPPR Data.

(a) Each of COMPANY, BPPR, and their respective Subsidiaries will remain the sole and exclusive owner of all BPPR Data, regardless of whether such data is maintained in physical, electronic or digital format or any other format, storage or processing device. All BPPR Data will, however, be subject to regulation and examination by the appropriate auditors and Governmental Authorities at the Authorized Locations to the same extent as if such information were on COMPANY, BPPR, or their respective Subsidiaries' premises. EVERTEC will notify COMPANY, BPPR, and their respective Subsidiaries as soon as reasonably possible of any formal request by any Governmental Authority to examine such information maintained by EVERTEC. COMPANY, BPPR, and their respective Subsidiaries agree that EVERTEC is authorized to provide all such information when properly required to do so by a Governmental Authority, subject to the provisions of Section 5.3(a). EVERTEC acknowledges that it will not have or acquire any rights in or to any BPPR Data or Confidential Information upon termination of this Master Agreement.

(b) EVERTEC will, subject to its internal control and security procedures, permit each of COMPANY, BPPR, and their respective Subsidiaries to have or obtain (by physical, electronic, digital or other means) access to its BPPR Data, including where appropriate, access through COMPANY, BPPR, or their respective Subsidiaries' computer terminals and equipment. EVERTEC will furnish COMPANY, BPPR, and their respective Subsidiaries with such written instructions, manuals or other documentation as will be necessary to such operation and access by COMPANY, BPPR, and their respective Subsidiaries.

## 6.4 Records Retention.

- (a) Each party will maintain its respective records related to the Services in a proper, complete and accurate fashion, and in compliance with all Legal Requirements applicable to each of them and in the case of EVERTEC with the requirements provided in Section 6.4(b).
- (b) EVERTEC will retain all BPPR Data, transactions data and records for the duration of the retention periods in accordance with the longer of (i) the retention periods in effect in practice as of the Effective Date; (ii) relevant Statement of Work or (iii) as established by COMPANY or BPPR following the date of this Master Agreement by written notice to EVERTEC.
- (c) COMPANY and BPPR acknowledge that any new retention periods or changes in existing retention periods requested by COMPANY or BPPR after the Effective Date may result in commercially reasonable additional charges and/or increases in the Service fees as provided under Section 2.7(b).
  - 6.5 <u>Data Backup</u>. Except as may be specifically provided in a Service Addendum or Statement of Work, EVERTEC:
- (a) subject to EVERTEC's obligations under Section 5.7, EVERTEC will maintain a backup of BPPR Data that is in a production environment in accordance with the backup practices used as of the Effective Date, which at a minimum must support the Recovery Point Objectives for the applicable Service (in physical or other electronic or digital media), from which lost or damaged items or data can be reconstructed within the Recovery Point Objectives established for the Services;
- (b) shall periodically identify all facilities and Data Centers where BPPR Data will be at rest or backed-up or where Data Processing will take place, and
- (c) shall, at its own cost and expense (subject to the Data Cap), reconstruct any BPPR Data or reports destroyed, lost or otherwise rendered unusable due to any breach of this Master Agreement or gross negligence or willful misconduct by EVERTEC or its Representatives.

# ARTICLE SEVEN - INTELLECTUAL PROPERTY

7.1 <u>Title</u>. To the extent EVERTEC uses its own Intellectual Property to provide the Services under this Master Agreement, EVERTEC warrants that it is the owner of all right, title, and interest in and to such Intellectual Property, and neither such Intellectual Property nor the authorized use thereof, to EVERTEC's best knowledge, infringes any Intellectual Property right,

or other proprietary right, of any other Person. As such, the parties agree that, subject to the applicable Legal Requirements and to existing agreements with Third Parties, or except as otherwise expressly agreed to between the parties in writing, EVERTEC is and will remain the owner of its Intellectual Property and all derivative works based thereon (which, for the avoidance of doubt, shall not include any Intellectual Property that is transferred pursuant to the Asset Purchase Agreement) or derivative works based thereon developed by or on behalf of BPPR or its Subsidiaries) and that no title to or ownership of EVERTEC's Intellectual Property or any part thereof is hereby granted to COMPANY, BPPR, or their respective Subsidiaries. Should EVERTEC use Third Party Intellectual Property to provide the Services, then EVERTEC warrants that it is duly licensed or otherwise has the right to use such Third Party Intellectual Property to provide the Services. The parties agree that, subject to the applicable Legal Requirements and to existing agreements with Third Parties, or except as otherwise expressly agreed to among the parties in writing, each of COMPANY, BPPR, and their respective Subsidiaries is and shall remain as the owner of its Intellectual Property, and all derivative works based thereon.

7.2 General. Each party acknowledges that (a) in providing the Services to COMPANY, BPPR, and their respective Subsidiaries, EVERTEC is not transferring any right, title or interest in EVERTEC's Intellectual Property, or any part or component thereof, to COMPANY, BPPR, or their respective Subsidiaries, provided that, for clarity, the foregoing will not limit anything to the contrary in the Asset Purchase Agreement; and (b) in receiving the Services, COMPANY, BPPR, and their respective Subsidiaries are not granting any right, title or interest in their respective Intellectual Property, or any part or component thereof, to EVERTEC or its Subsidiaries.

# 7.3 Developments.

- (a) Except as otherwise agreed to in writing, any services, technology, processes, methods, software and/or enhancements to EVERTEC Intellectual Property or any EVERTEC Third Party Intellectual Property used or developed for purposes of delivering the Services (collectively, the "IP Developments"), whether developed solely by EVERTEC or jointly by EVERTEC and any other party, including any IP Developments requested or suggested by COMPANY, BPPR, or their respective Subsidiaries or a Client, will be the sole property of EVERTEC and will not be considered "works-made-for-hire"; provided, however, that the foregoing shall not constitute a grant in, to or under any right, title or interest any Intellectual Property, or any part or component thereof, of the COMPANY, BPPR, or their respective Subsidiaries, to EVERTEC or any other party. Except as otherwise agreed to in writing, COMPANY, BPPR, and their respective Subsidiaries will not acquire any ownership right, Intellectual Property right, claim or interest in EVERTEC's Intellectual Property, including as to any IP Developments. To the extent COMPANY or BPPR or one of their respective Subsidiaries does nevertheless acquire such ownership right, Intellectual Property right, claim or interest, each of COMPANY and BPPR hereby assign (and shall cause its applicable Subsidiaries to assign) all of its right, title and interest in and to the IP Developments to EVERTEC.
- (b) The parties agree that any enhancements or other modifications to or derivative works of COMPANY Intellectual Property or COMPANY Third Party Intellectual Property, whether developed solely by EVERTEC or jointly by EVERTEC and any other party in connection with the provision of the Services or otherwise will be the sole property of COMPANY,

BPPR, or one of their respective Subsidiaries and will be considered "works-made-for hire". Except as otherwise agreed to in writing, EVERTEC will not acquire any ownership right, Intellectual Property right, claim or interest in COMPANY Intellectual Property or COMPANY Third Party Intellectual Property. To the extent EVERTEC acquires any ownership right, Intellectual Property right, claim or interest in any enhancements to COMPANY Intellectual Property or COMPANY Third Party Intellectual Property, EVERTEC hereby assigns (and shall cause its applicable Subcontractors to assign) all of its right, title and interest in and to such enhancements and other materials to COMPANY, BPPR, or one of their respective Subsidiaries, as applicable.

- (c) Except as otherwise agreed to in writing, any Intellectual Property created or developed by EVERTEC as an independent product will be EVERTEC Intellectual Property and the sole property of EVERTEC and will not be considered a "work-made-for-hire".
- (d) EVERTEC hereby grants to COMPANY, BPPR, and their respective Subsidiaries, a perpetual, irrevocable, non-exclusive, sublicensable, worldwide, paid-up right and license to access, use, copy, maintain, modify, enhance, perform, display, distribute and create derivative works of any EVERTEC Intellectual Property and EVERTEC Third Party Intellectual Property that EVERTEC includes in (i) the Procedures Manual or (ii) in any enhancements or modifications to or derivative works of the COMPANY Intellectual Property or COMPANY Third Party Intellectual Property.
- (e) "EVERTEC Third Party Intellectual Property" means Third Party Intellectual Property that is licensed and made available under this Master Agreement by EVERTEC. "COMPANY Third Party Intellectual Property" means Third Party Intellectual Property that is licensed and made available under this Master Agreement by COMPANY or BPPR or any of their respective Subsidiaries.

# 7.4 Intellectual Property Infringement.

- (a) Subject to Section 7.4(d), EVERTEC agrees to defend, indemnify and hold harmless COMPANY, BPPR, and their Affiliates and Subsidiaries and their respective Representatives, Third Party information providers, Subcontractors and permitted assigns and successors in interest ("IP Indemnified Parties") from and against all Losses incurred or suffered by, or asserted against, such IP Indemnified Party directly or indirectly in relation to or arising from any claim, action or other proceeding from a Third Parties alleging that any of the Services or EVERTEC's or its Affiliates' Systems or Intellectual Property, or the use thereof by or on behalf of the IP Indemnified Party, infringes or misappropriates any Intellectual Property right of a Third Party.
- (b) If EVERTEC receives notice of an infringement claim or otherwise concludes that the Services or its or its Affiliates' Intellectual Property, or the use thereof by or on behalf of the COMPANY, BPPR, and their respective Subsidiaries, may infringe the proprietary rights (including Intellectual Property rights) of a Third Party, EVERTEC may in its sole discretion (i) procure the right for COMPANY, BPPR, and their respective Subsidiaries to continue using the affected Service or Intellectual Property; (ii) modify the affected Service or Intellectual Property to make it non-infringing; or (iii) replace the affected Service or Intellectual Property with a functional equivalent without substantial degradation in functionality, utility or performance.

(c) If EVERTEC receives notice of an infringement claim against an IP Indemnified Party related to an EVERTEC Pass Through License, if requested by BPPR or COMPANY, EVERTEC shall use Best Efforts to pursue its rights, if any, for indemnification or other monetary remedies against the applicable Third Party licensor in accordance with the terms set forth in the EVERTEC Pass Through License. Monetary awards actually received by EVERTEC, less the amount of all fees, claims, costs and expenses (including attorneys' fees) incurred by EVERTEC in connection with pursuing such claim against a Third Party provider for infringement, shall be allocated between EVERTEC and its other customers, on the one hand, and the COMPANY, BPPR and their respective Subsidiaries, on the other, in proportion to the damages and costs sustained by such party due to the EVERTEC Pass Through License.

(d) EVERTEC will have no liability for any claim of infringement and thus no obligation to defend and indemnify COMPANY, BPPR, and their respective Subsidiaries under this Section 7.4 if such infringement claim is based on (i) COMPANY, BPPR, and their respective Subsidiaries' use of a superseded or altered (other than alterations by or on behalf of, or with the written authorization or instruction of EVERTEC) release of the affected Service or Intellectual Property or any portion thereof, including COMPANY, BPPR, and their respective Subsidiaries' failure to use updates or new releases made available by EVERTEC but only to the extent that such claim would have been avoided but for such failure; (ii) any modification by COMPANY, BPPR, and their respective Subsidiaries' use of the affected Service or Intellectual Property, but only to the extent that such claim would have been avoided, but for such modification; (iii) COMPANY, BPPR, and their respective Subsidiaries' use of the affected Service or Intellectual Property in violation of this Master Agreement; (iv) COMPANY, BPPR, and their respective Subsidiaries' use, operation or combination of the affected Service or Intellectual Property with information, software, specifications, instructions, data, materials or items not supplied or approved by EVERTEC, but only to the extent that such claim would have been avoided but for such combined use; or (v) use of the affected Service or Intellectual Property in a manner in violation of, or not as intended by, the accompanying and provided documentation.

(e) Furthermore, EVERTEC's obligation to defend COMPANY, BPPR, and their respective Subsidiaries under this Section 7.4 is subject to all of the following conditions: (i) COMPANY, BPPR, or their respective Subsidiaries must notify EVERTEC promptly in writing after the claim is asserted or threatened (provided the failure of COMPANY, BPPR or a Subsidiary to so notify EVERTEC will relieve EVERTEC of its obligations under this Section 7.4 only to the extent that EVERTEC can demonstrate damages attributable to such failure); (ii) COMPANY, BPPR, or their respective Subsidiaries must give EVERTEC sole control over its defense or settlement; and (iii) COMPANY, BPPR, and their respective Subsidiaries must provide EVERTEC with reasonable assistance in defending the claim for which EVERTEC will reimburse COMPANY, BPPR, and their respective Subsidiaries for any reasonable out-of-pocket expenses that COMPANY, BPPR, or their respective Subsidiaries incur in providing such assistance.

- (f) COMPANY and BPPR agree to notify EVERTEC promptly in writing if any other type of Third Party claim is brought against COMPANY, BPPR, or their respective Subsidiaries regarding EVERTEC's Intellectual Property. EVERTEC may, at its option, choose to treat these claims as being covered by this Section 7.4.
- (g) This Section 7.4 states EVERTEC's entire liability and COMPANY's and BPPR's exclusive remedies with respect to any Third Party infringement and trade secret misappropriation claims with respect to the Services; provided, however, that the foregoing shall not limit any right of the COMPANY or BPPR, or their respective Affiliates, pursuant to the Asset Purchase Agreement or any ancillary agreement thereto (excluding this Master Agreement).

# ARTICLE EIGHT- REGULATORY COMPLIANCE, AUDIT & SERVICE REVIEWS

# 8.1 Compliance.

- (a) EVERTEC acknowledges that EVERTEC will be solely responsible for monitoring and interpreting (and for complying with) Legal Requirements applicable to EVERTEC, and as such, hereby warrants that EVERTEC will comply with applicable Legal Requirements, present and future, relating to the conduct and operation of its business.
- (b) COMPANY and BPPR acknowledge that each of COMPANY and BPPR will be solely responsible for monitoring and interpreting (and for complying with, to the extent such compliance requires no action by EVERTEC) Legal Requirements applicable to COMPANY and its Subsidiaries and BPPR and its Subsidiaries, as applicable, and as such, hereby warrants that it and its Subsidiaries will comply with all applicable Legal Requirements, present and future, relating to the conduct and operation of its or such Subsidiaries' business.
- (c) EVERTEC represents that it has, and at all times during the Term will, at its own expense, maintain any and all licenses, permits, approvals, certificates, endorsements and authorizations required by any professional association or the applicable Governmental Authorities that are required for it to provide the Services for the benefit of COMPANY and BPPR. Upon thirty (30) days prior written notification to EVERTEC, COMPANY and BPPR shall have the right to inspect copies of all such licenses, permits, approvals, certificates, endorsements or authorizations.

# 8.2 Legal Requirements.

(a) COMPANY and BPPR shall be responsible for monitoring and interpreting (and for complying with, to the extent such compliance requires no action required to be undertaken by EVERTEC pursuant to this Master Agreement) the Legal Requirements applicable to them. COMPANY and BPPR may, as part of the instructions used to establish the Specifications, include requirements that they deem necessary in order to comply with applicable Legal Requirements, and EVERTEC will select the technical parameters within EVERTEC's Systems that will apply to COMPANY, BPPR, and their respective Subsidiaries. EVERTEC shall be responsible for determining that such selections are consistent with COMPANY's or BPPR's instructions, as applicable. COMPANY and BPPR, as applicable, shall be responsible for determining that the instructions provided to EVERTEC are consistent with the Legal Requirements and with the terms and conditions of any agreements between COMPANY, BPPR, and their respective Subsidiaries and its Clients.

- (b) Subject to <u>Sections 8.1</u> and <u>8.2(a)</u>, at the expense of COMPANY and BPPR, EVERTEC shall work with COMPANY and BPPR in developing Systems and implementing Specifications for the Services to enable COMPANY, BPPR, and their respective Subsidiaries to comply with Legal Requirements applicable to the Services being provided by EVERTEC to COMPANY, BPPR, and their respective Subsidiaries.
  - (c) The parties acknowledge and agree that the performance of the Services may be subject to regulation by Governmental Authorities.
- (d) Upon the request and expense of COMPANY and BPPR, EVERTEC will take all commercially reasonable steps to cooperate in good faith with COMPANY and BPPR in any investigations, reports (including, any suspicious activity reports), filings or other documents or actions COMPANY or BPPR may take or complete to comply with applicable federal anti-money laundering and bank secrecy laws and regulations.
- 8.3 <u>BSA/AML/OFAC</u>. EVERTEC represents and warrants that it (a) is in compliance with the Office of Foreign Assets Control sanctions and regulations promulgated under the authority granted by the Trading with the Enemy Act, 12 U.S.C. § 95 (a) *et seq.*, the International Emergency Economic Powers Act, 50 U.S.C. § 1701, *et seq.*; and the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, as each may be amended from time to time and all rules and regulations promulgated thereunder applicable to EVERTEC and (b) is not: (i) listed in the Annex to, or otherwise subject to the provisions of, that certain Executive Order No. 13224 on Terrorist Financing, effective September 24, 2001, and relating to Blocking Property and Prohibiting Transactions With Persons Who Commit, Threaten to Commit or Support Terrorism; and/or (ii) named as a "specifically designated national (SDN)" on the most current list published by the U.S. Treasury Department Office of Foreign Assets Control. EVERTEC further represents that it will not act directly or indirectly, in contravention of any applicable anticorruption or anti-money laundering law.

#### 8.4 Audit.

- (a) Each of the parties agrees that it will keep current and accurate books of accounts and records, in accordance with its standard operating procedures, with respect to the transactions effected pursuant to this Master Agreement. During the Term, each party shall permit the other party's designated auditors and Governmental Authorities to review its books and records with respect to such transactions, upon prior written notice, during normal business hours.
- (b) EVERTEC, COMPANY, and BPPR acknowledge and agree that the performance of the Services may be subject to regulation by Governmental Authorities. EVERTEC agrees to use Best Efforts to cooperate with any audit or examination of the Services or COMPANY, BPPR, or their respective Subsidiaries, whether by a Governmental Authority or internal or external auditors of COMPANY, BPPR, or their respective Subsidiaries ("Audit"). Except for Audits required to be conducted by a Governmental Authority or by Legal Requirement, and for Audits conducted to verify remediation of a material breach of any provision of this Master

Agreement uncovered in a previous Audit (a "For Cause Audit"), no more than one Audit shall be conducted in any twelve-month period. Furthermore, EVERTEC agrees to provide any information or material lawfully and reasonably requested during an Audit, and permit such auditing parties to inspect or audit EVERTEC with respect to its provision of the Services; provided, however, that (i) each of COMPANY and BPPR agrees to reimburse EVERTEC for all reasonable costs incurred by EVERTEC to fulfill Audit requests from COMPANY, BPPR or its Subsidiaries under this Section 8.4b) (other than For Cause Audits); (ii) prior to reviewing any such information or material provided by EVERTEC, any Third Party (other than a Governmental Authority) shall execute a customary confidentiality agreement with EVERTEC; and (iii) nothing in this Section 8.4 shall require EVERTEC to provide information that would be expected to result in the waiver of any attorney-client or other legal privilege or disclosure of the confidential information of EVERTEC's other customers that is not related to COMPANY or BPPR or their Subsidiaries. COMPANY or BPPR requests for detail to support invoices or copies of any policies, procedures, summaries, audit reports or other information or access that EVERTEC is required to provide under this Master Agreement (including under Section 9.3) will not be considered "Audits" for purposes of the frequency limitation and cost reimbursement provisions above.

- (c) In connection with each Audit, EVERTEC will provide COMPANY, BPPR, and their respective Subsidiaries and auditors, upon COMPANY's or BPPR's written request, at a time, frequency and place reasonably agreed to between the parties, with access to Personnel and information relating to the Services to permit COMPANY, BPPR, and their respective Subsidiaries and auditors to (w) review EVERTEC's performance of Services; (x) validate EVERTEC's compliance with this Master Agreement; (y) comply with applicable Legal Requirements; and (z) conduct third party management and oversight discussions with management and personnel of EVERTEC concerning the status and conduct of Services and work being performed under this Master Agreement. EVERTEC's obligation in this regard will include providing to COMPANY and BPPR the following reports (each, an "Audit Report") on an annual basis:
  - (i) Service Center Review reports (as further described in <u>Section 8.7</u>);
  - (ii) Business Continuity Plan and business resumption test results (as further described in Section 2.13);
  - (iii) Data Protection Program documentation (as further described in Section 5.3(f);
  - (iv) Penetration Tests (as further described in Section 6.2(c));
  - (v) vulnerability and patch management implementation reports; and
  - (vi) responses to vendor questionnaires and other standardized information gathering requests.
- (d) To the extent directed by a Governmental Authority or required by a Legal Requirement, EVERTEC and its Affiliates will provide COMPANY and BPPR or their respective Subsidiaries or auditors with access to their facilities and Systems and related Personnel to address the request from the Governmental Authority or comply with the Legal Requirement.

- (e) To the extent an Audit reveals areas of material concern to COMPANY or BPPR, the parties will agree on a plan to address the deficiency in an adequate and timely manner, it being agreed that the costs of implementing any aspect of such plan shall be borne by EVERTEC to the extent that the deficiency is attributable to actions or omissions on the part of EVERTEC or any Subcontractor. EVERTEC will promptly execute any such plan and track and report on its activities and results obtained under the plan until the parties agree the area of concern has been remediated.
- 8.5 <u>IT Infrastructure Network Diagram</u>. Upon COMPANY's or BPPR's written request, EVERTEC will provide a network diagram that outlines EVERTEC's information technology network infrastructure and all equipment used in relation to the fulfillment of its obligations under this Master Agreement.
- 8.6 <u>Data Flow Diagram</u>. Upon COMPANY's or BPPR's written request, EVERTEC will provide a graphical representation of the flow of electronic BPPR Data within and between network segments and information systems as well as across an institution's perimeter up to Third Parties, if applicable ("<u>DFD</u>"). A DFD must capture the main components of an information system, how data moves in and out of the system, protection layers (encryption) applied to data in its different states, user interaction points and their functions, and external and internal network dependencies such as Third Party or fourth party connections or services.

# 8.7 Service Center Reviews.

- (a) On an annual basis during the Term, EVERTEC shall engage a reputable independent certified public accounting firm of recognized national or regional standing (the "Firm"), to conduct an examination of its IT operations and application controls for the Services provided to COMPANY, BPPR, and their respective Subsidiaries, in accordance with applicable Industry Standards and applicable Legal Requirements. In addition to IT operations and application controls, the Firm should complete an examination of EVERTEC's controls related to security, availability, processing integrity, confidentiality or privacy. The aforesaid examinations shall be conducted for the relevant System and organization controls ("SOC 1" and "SOC 2") criteria in accordance with Statement on Standards for Attestation Engagements no. 18 ("SSAE 18") or any applicable successor standard, the findings and recommendations of which shall be set forth in a report (the "Service Center Review").
  - (i) The SOC 1 Service Center Review shall (A) include a Type II Service Auditor's Report under SSAE 18, (B) cover, at a minimum, a period of nine (9) months, (C) be dated as of September 30 of the year in question, and (D) be delivered to COMPANY and BPPR on or before December 15 of such year. EVERTEC management will provide a bridge letter for the remaining period of the calendar year that is not covered by the SOC 1 Type II report no later than January 31st of the following year.
  - (ii) The SOC 2 Service Center Review shall (A) include a Type II Service Auditor's Report under SSAE 18, (B) cover, at a minimum, a period of twelve (12) months, (C) be dated as of September 30 of the year in question, and (D) be delivered to COMPANY and BPPR by February 15 of the following year.

- (iii) All such Service Center Reviews shall be deemed to be EVERTEC's Confidential Information hereunder, provided that EVERTEC shall promptly share such Service Center Reviews with BPPR upon request. EVERTEC may reduct information from any Service Center Reviews shared with BPPR if and solely to the extent necessary to protect the confidential information of other customers of EVERTEC or the information of EVERTEC that is not related to either BPPR or EVERTEC's provision of Services to BPPR.
- (iv) Additionally, EVERTEC will use its Best Efforts to obtain and provide to COMPANY and BPPR SOC 1 and 2 Type II reports or other equivalent control reports for each Data Subcontractor that stores, transfers or processes Personal Data. If a Data Subcontractor proposes to limit EVERTEC's rights in this regard in any new contract or contract renewal after the Effective Date, EVERTEC will notify COMPANY and BPPR reasonably in advance of the contract execution date of the issue and use Best Efforts to address any concerns relating to the issue that COMPANY or BPPR may raise with EVERTEC.
- (b) EVERTEC will review the scope of the Service Center Review to be conducted by the Firm on an annual basis with BPPR and incorporate reasonable input and additional requirements reasonably requested by BPPR. If BPPR requires additional scope in a Service Center Review that is (i) incremental to the scope in EVERTEC's standard Service Center Review; and (ii) solely attributable to BPPR (and not to be shared with any other EVERTEC customers) ("Incremental BPPR Requirements"), BPPR will reimburse EVERTEC for any additional amounts payable by EVERTEC to the Firm to include the Incremental BPPR Requirements.
- (c) If the Service Center Review contains any recommendations, EVERTEC shall, at its sole cost and expense, promptly take all reasonable actions necessary to comply with such recommendations on a timeline that will take into account the nature and materiality of the recommendations. Upon request, EVERTEC will advise COMPANY and BPPR on its progress in completing such actions.
- (d) If, at any time during the Term, COMPANY or BPPR has reasonable material concerns regarding (i) the scope of the Service Center Review, (ii) any material qualification in the aforesaid report, and/or (iii) EVERTEC's operational and application controls and such concerns are not addressed in the Service Center Review to COMPANY's or BPPR's reasonable satisfaction (as applicable), COMPANY or BPPR (as applicable) shall so notify EVERTEC and EVERTEC shall promptly meet with COMPANY or BPPR in an effort to resolve such concern.
- (e) For all other requests by COMPANY or BPPR to review all or a portion of the Services and EVERTEC's operation controls relating thereto, outside the scope of an Audit or SSAE 18, the parties will agree in writing to the terms and conditions applicable to such review, including the scope of the review and any expenses related thereto. If, at any time, whether or not

in connection with the Service Center Review, EVERTEC becomes aware of a significant issue of concern related to the operational and application controls operating at EVERTEC for the Services provided to the COMPANY, BPPR, or any of its respective subsidiaries, EVERTEC shall promptly notify and subsequently meet with COMPANY or BPPR in an effort to address such concern.

## ARTICLE NINE- TERM & TERMINATION

9.1 <u>Term</u>. This Master Agreement will commence on the Effective Date and will end on September 30, 2028 (the "<u>Initial Term</u>"), unless earlier terminated in accordance with the provisions of this Master Agreement.

## 9.2 Termination for Cause.

- (a) This Master Agreement or any of the Services, individually or collectively, may be terminated by either the Popular Parties, on the one hand, or EVERTEC, on the other, if the other party or parties (as applicable):
  - (i) commits a Material Breach of this Master Agreement (or series of breaches that together constitute a Material Breach), which breach is not cured within thirty (30) days following receipt of notice specifying the nature and extent of such breach; <u>provided</u>, <u>however</u>, that if such breach is not reasonably susceptible of cure within such thirty (30) day period, such period will be extended and the party will not be in default hereunder so long as it commences such cure within such thirty (30) day period and diligently pursues such cure and such failure is cured within ninety (90) days following the receipt of such notice;
  - (ii) fails to pay any properly submitted invoice providing for material amounts in the aggregate that are undisputed for a period exceeding sixty (60) days following COMPANY and BPPR's receipt of notice from EVERTEC of such non-payment; or
    - (iii) makes any assignment of this Master Agreement, except as expressly provided herein.
  - (b) For purposes of this Section 9.2 and notwithstanding anything in this Master Agreement to the contrary:
  - (i) "Material Breach" means a breach, or series of breaches, of a party's duties or obligations (other than a failure to make a payment pursuant to this Master Agreement) that if left uncured for ninety (90) days following receipt of notice from COMPANY or BPPR detailing the relevant deficiency or failure, (A) would result in a Material Adverse Effect on such notifying party and its Subsidiaries (taken as a whole) or BPPR and its Subsidiaries (taken as a whole), as applicable, or (B) would reasonably be expected to have an adverse effect on the notifying party that is material, individually or in the aggregate, measured on a scale relative to the notifying party's rights and privileges and reasonably expected benefits under this Master Agreement, provided that this clause (B) shall apply solely after September 30, 2025 and solely in the case of termination by COMPANY or BPPR.

- (ii) "Material Adverse Effect" means, with respect to any Person, any fact, event, change, effect, development, condition or occurrence that has a materially adverse effect on or with respect to any business, assets, liabilities, financial condition, or results of operations of such Person.
- (c) Notwithstanding anything in this Master Agreement to the contrary, any breach or default under a particular Service Addendum will give the non-breaching party or parties the right to terminate that particular Service Addendum, subject to the cure periods set forth under Section 9.2(a)(i), and will not automatically operate as a default under any other Service Addendum.

# 9.3 Termination without Cause.

- (a) Except as provided in a Service Addendum or SOW signed after the Effective Date, the Popular Parties may terminate any of the Services, Service Addendums or SOW in whole or in part at any time at their sole discretion with one hundred eighty (180) days prior written notice.
- (b) If COMPANY or BPPR terminate a component of the Services, and that component does not have a discrete service code, unit rate or other charging methodology applicable to it: (i) the parties will discuss in good faith whether a corresponding reduction to the charges is appropriate, but (ii) EVERTEC will not be required to reduce the charges as a result of such termination unless the parties agree to a reduction in writing. Notwithstanding the preceding sentence, the Parties shall negotiate in good faith to agree on discrete service codes for each component of the Services following the date hereof.
- (c) EVERTEC may terminate any Services, Service Addendums or SOWs that have been unused or unconsumed by any Popular Party for longer than ninety (90) days by providing sixty (60) days prior written notice (an "EVERTEC Cancellation Notice"), provided such ninety (90) day period will not apply with respect to (i) Services that are used on a seasonal basis (e.g., annual processing functions) or (ii) Services that are scheduled for use on a later date following implementation or other circumstances. The termination will not be effective if the applicable Popular Party sends written notice to EVERTEC within thirty (30) days after receipt of an EVERTEC Cancellation Notice with reasons as to why the relevant Service, Service Addendum or SOW should not be terminated notwithstanding lack of usage or consumption.
- (d) In connection with the Core API Layer SOW, subject to Section 3.10(c), the Popular Parties shall have the right to terminate this Master Agreement effective as of September 30, 2025, in whole or in part (including, for the avoidance of doubt, the Popular Parties' obligations with respect to Section 2.2), in the event that:
  - (i) if the Core API Deadline is on or before June 30, 2023 (the "Base Date"), the Core API Completion does not occur within eighteen (18) months of the Core API Deadline (the "Core API Delay Window");

- (ii) if the Core API Deadline is after the Base Date, the Core API Completion does not occur within the Core API Delay Window less the number of months between the Base Date and the Core API Deadline; provided, that the Core API Delay Window shall in no event be reduced to less than twelve (12) months after the Core API Deadline; and
- (iii) notwithstanding anything to the contrary in clause (i) and (ii) above, if the Core API Deadline is June 30, 2025 as a result of the failure of Virtusa Corporation to complete the Virtusa Work Order, as contemplated by Section 3.10(a), the Core API Completion does not occur within eighteen (18) months of the Core API Deadline.
- (e) The termination of any or all Services, Service Addendums or SOWs under this <u>Section 9.3</u> shall not result in the automatic termination of this Master Agreement or the Popular Parties' obligations to pay Annual Minimums under <u>Section 2.2</u> (except as provided otherwise in <u>Section 2.2</u> or <u>Section 9.3(d)</u>).

# 9.4 Effect upon Termination.

- (a) Unless indicated otherwise in the written termination notice, upon the termination of any Service, all remaining Services will continue in full force and effect; <u>provided</u>, <u>however</u>, that should this Master Agreement be terminated, all Services in effect as of the date of termination will also terminate.
- (b) Except as otherwise provided in this Master Agreement, upon termination, all further obligations of the parties pursuant to this Master Agreement or the particular Service that was terminated, whichever the case may be, will terminate (except the obligation of COMPANY, BPPR, and their appropriate respective Subsidiaries to make a payment for any unpaid and properly invoiced amounts, in accordance with Section 3.2 and the obligation of EVERTEC to provide Transition Assistance) without further liability of any party to the others; provided, however, that termination will not release the party that terminates from any liability which at the time of termination had already accrued to the non-terminating party or parties, including EVERTEC's obligation to continue to provide the Services during the Transition Period, if so requested by COMPANY or BPPR, and the parties' obligations to comply with the provisions of this Master Agreement related to any Services being provided during the Transition Period.
- (c) No party shall be liable to the others for damages of any kind solely as a result of terminating this Master Agreement in accordance with its provisions.
- (d) Furthermore, any such termination will be without prejudice to any rights or remedies any party may have arising out of any breach of any material representation, warranty, covenant or condition by any other party hereto.

#### 9.5 Transition Assistance.

- (a) "<u>Termination Notice Date</u>" means, as applicable for each Affected Service, the earliest of (i) twelve (12) months prior to expiration of the Initial Term, (ii) the date of any termination of this Master Agreement and all Services hereunder pursuant to <u>Section 9.2</u> hereof, (iii) the date of notice of termination or non-renewal of any individual Service or Services under this Master Agreement with or without the termination of this Master Agreement, including terminations under <u>Section 9.3</u> and any reductions or removal of portions or volumes of Services, and (iv) the date of a Release Event pursuant to <u>Section 3.1</u> of the Technology Agreement.
- (b) "Affected Services" means any Services that are not renewed or that are terminated pursuant to a particular notice of non-renewal or termination, as applicable.
- (c) For each Affected Service, a "<u>Transition Period</u>" commences on the applicable Termination Notice Date and continues until the earlier of (i) the 18-month anniversary of the effective date of termination or expiration of the Affected Services and (ii) the date on which all Affected Services have been completely transitioned to COMPANY, BPPR, or another third-party provider designated by COMPANY or BPPR (a "<u>Successor Supplier</u>"), subject to any extension of the Transition Period as described in <u>Section 9.5(e)</u>.
- (d) During each Transition Period, EVERTEC will (i) provide COMPANY, BPPR, and their respective Subsidiaries and Successor Suppliers transition support and assistance including providing COMPANY, BPPR, and their respective Subsidiaries and Successor Suppliers (x) information and cooperation necessary to effect the transition, with business continuity, of the applicable Affected Services and (y) the support and assistance described in Schedule 9.5 and (ii) continue to provide the applicable Affected Services to COMPANY or BPPR, their respective Subsidiaries, subject to the terms and conditions (including payment terms) that are in effect as of the applicable Termination Notice Date ((i) and (ii) collectively, "Transition Assistance"). During the Transition Period, COMPANY, BPPR, and their respective Subsidiaries and EVERTEC shall use their Best Efforts to completely transition each Affected Service requested by the COMPANY, BPPR, and their respective Subsidiaries to COMPANY, BPPR, and their respective Subsidiaries and Successor Suppliers prior to the end of the Transition Period.
- (e) If COMPANY or BPPR, or their respective Subsidiaries or Successor Suppliers, as applicable, have used and continue to use Best Efforts to completely transition the applicable Affected Services and reasonably continue to need such Transition Assistance following the 18-month anniversary of the effective date of termination or expiration of the Affected Services, the Transition Period shall be extended until such Service or Services shall be completely transitioned to COMPANY or BPPR or their respective Subsidiaries or Successor Suppliers (the "Extended Transition Period"), subject to the parties' agreement on an updated fee arrangement if the Extended Transition Period extends beyond March 31, 2030.
- (f) On September 30, 2028, hourly rates for any programming/development necessary to provide the Transition Assistance described in Section 9.5(d)(i) requested by COMPANY or BPPR will be adjusted by the lesser of (i) the aggregate change in CPI for the period from September 30, 2022 through September 29, 2028 (not to exceed 25%); and (ii) the difference between the sum of the change in annual CPIs (not to exceed 5%) during each one-year period commencing on September 30, 2022 and ending on September 29, 2028 less the sum of CPI adjustment actually applied during those years pursuant to Section 3.1(b) herein; provided that the fee adjustment cannot be negative.

- (g) In furtherance of and as a part of such Transition Assistance and as further described in <u>Schedule 9.5</u>, EVERTEC shall assist COMPANY, BPPR, and their respective Subsidiaries to develop a plan for the complete transition of all Services requested by the COMPANY, BPPR, or their respective Subsidiaries from EVERTEC to COMPANY, BPPR, and their respective Subsidiaries or a Successor Supplier, on a reasonable schedule, which shall give consideration to COMPANY, BPPR, and their respective Subsidiaries' need for the orderly continuation of such Services.
- (h) Prior to providing any Transition Assistance, EVERTEC shall deliver to COMPANY, BPPR, and their respective Subsidiaries a good faith estimate of all projected expenses and charges payable for the Transition Assistance; provided that:
  - (i) to the extent the Transition Assistance consists of Services that were provided pursuant to this Master Agreement, the expenses and charges for such Services shall be the price for such Services under this Master Agreement), which expenses and charges shall be paid by COMPANY, BPPR, or their appropriate respective Subsidiaries in accordance with the provisions of this Master Agreement; and
  - (ii) to the extent the Transition Assistance consists of additional programming, development or other assistance that is incremental to the Services that were provided pursuant to this Master Agreement, the expenses and charges for personnel resources will be determined using the programming/development time and materials rates then in effect under this Master Agreement; provided, if COMPANY or BPPR requests assistance that requires specialized skills that (A) EVERTEC does not make available to COMPANY or BPPR under this Master Agreement; (B) are only available from third parties at a higher cost than the costs EVERTEC incurs to engage third parties for other skills under this Master Agreement and (C) COMPANY or BPPR, as applicable, determines not to engage a third party or parties directly, EVERTEC may increase the time and materials rates to reflect the incremental costs EVERTEC must incur to procure the specialized skills requested by COMPANY or BPPR.
- (i) The quality of the Services provided by EVERTEC, and EVERTEC's performance of the Services, will not be materially degraded during the period EVERTEC is providing Transition Assistance. EVERTEC shall not make any changes to the EVERTEC Personnel that are critical to providing Services in accordance with this Master Agreement during the Transition Period or reassign any such critical EVERTEC Personnel away from performing Services under this Master Agreement during the Transition Period except as mutually agreed to by the parties in writing.
- (j) EVERTEC will, to the extent requested by COMPANY, BPPR, or their respective Subsidiaries: (i) assign to COMPANY, BPPR, their respective Subsidiaries and/or the Successor Supplier leases for some or all of the EVERTEC leased equipment that is located at facilities of COMPANY, BPPR, or their respective Subsidiaries, including applicable maintenance agreements that is located at facilities of COMPANY, BPPR, or their respective Subsidiaries, and COMPANY, BPPR, their respective Subsidiaries and the Successor Supplier shall assume the obligations under such leases that relate to periods after such date (to the extent EVERTEC is able to obtain any required consents for such assignments); and (ii) sell to COMPANY, BPPR, their

respective Subsidiaries or the Successor Supplier, at EVERTEC's then current book value, some or all of the equipment owned by EVERTEC or its Affiliates that is located at facilities of COMPANY, BPPR, or their respective Subsidiaries, including applicable maintenance agreements. EVERTEC will also provide all user and other documentation relevant to such Equipment in EVERTEC's possession. COMPANY, BPPR, their respective Subsidiaries or the Successor Supplier (as applicable). EVERTEC will be operationally responsible for obtaining any consents from Third Parties necessary to obtain any licenses or transfer any Equipment to COMPANY, BPPR, their respective Subsidiaries or Successor Supplier; provided that COMPANY or BPPR, as applicable, shall be responsible for any out-of-pocket costs incurred by EVERTEC in connection therewith.

- (k) The provisions of this Master Agreement shall apply with respect to all Services (including Transition Assistance) provided during the Transition Period. Nothing contained herein shall obligate COMPANY, BPPR, or their respective Subsidiaries to receive Transition Assistance from EVERTEC.
- (1) Notwithstanding anything to the contrary in this Master Agreement, the provisions of this <u>Section 9.5</u>, including <u>Schedule 9.5</u>, shall survive the expiration or termination of this Master Agreement (or any portion thereof) for any reason.
- (m) As and when reasonably requested by COMPANY, BPPR, and their respective Subsidiaries during the Term, EVERTEC will provide to COMPANY, BPPR, and their respective Subsidiaries such information and other cooperation regarding the performance of the Services and the manner in which the Services are integrated with COMPANY's, BPPR's and their Subsidiaries' Systems and processes as would be reasonably necessary to enable COMPANY, BPPR, and their respective Subsidiaries or potential third-party providers to conduct due diligence and prepare an informed plan to obtain such services from a third party provider. The Popular Parties and their respective Subsidiaries may disclose such information and other information about the Services to such third parties, provided such third parties may use it solely for the purposes described in this Section 9.5(m) and will be subject to confidentiality obligations substantially similar to those in this Master Agreement. This Section 9.5(m) will not, however, require EVERTEC to provide Proprietary Application Information.

## ARTICLE TEN- INSURANCE AND INDEMNIFICATION

#### 10.1 Insurance.

- (a) EVERTEC agrees to maintain adequate insurance coverage from reputable providers in amounts as set forth below. All such insurance policies will be carried at EVERTEC's own expense. Except as EVERTEC, COMPANY and BPPR may otherwise agree, EVERTEC will maintain in full force and effect during the Term insurance as follows:
  - (1) Statutory Workers' Compensation Insurance or a workers' compensation and employers' liability insurance, as applicable under the applicable state statutory law of the relevant jurisdiction, covering all its employees and including occupational disease coverage.

- (2) Commercial General Liability Insurance covering against claims for bodily injury, death, property damage, personal injury, advertising injury and products/completed operations with limits in an amount of not less than one million dollars (\$1,000,000) per occurrence. The annual aggregate limit shall not be less than two million dollars (\$2,000,000). For this policy, the Insurance Services Office Inc. ("ISO") Commercial General Liability occurrence coverage form CG 00 01 or its equivalent will be used.
- (3) Automobile Liability Insurance with limits not less than 1 million dollars (\$1,000,000) each occurrence combined single limit of liability. This insurance shall include Third Party bodily injury and property damage liability including owned and not owned and hired automobile coverage. For this policy, the ISO form number CA 00 or its equivalent will be used.
- (4) Umbrella Insurance with limits not less than five million dollars (\$5,000,000) covering excess of loss over primary liability insurance policies, including Comprehensive General Liability, Comprehensive Automobile Liability and Employers Liability insurance policies, where applicable under state Legal Requirements.
- (5) Fidelity and Crime insurance with limits not less than fifteen million dollars (\$15,000,000) each occurrence and annual aggregate including Electronic & Computer Crime, Unauthorized Computer Access coverage, and employee dishonesty coverage which shall be extended to loss of Third Parties.
- (6) Technology professional liability and cyber insurance (Errors and Omissions) with limits not less than fifteen million dollars (\$15,000,000) each occurrence and annual aggregate covering any damages caused by an error, omission or any negligent acts related to the services to be provided under this Master Agreement by EVERTEC and/or its Subcontractors, agents, officers or employees and including coverage for computer programming and electronic data processing services, network security liability, content injury, privacy injury, regulatory proceedings, dependent loss and Third Party custodian. This insurance shall include coverage for Third Party claims and for losses, unauthorized disclosures, access or use of BPPR Data in any format, including coverage for accidental loss, theft, unauthorized disclosure access or use of BPPR Data.
- (7) Employment Practices Liability Insurance with limits not less than one million dollars (\$1,000,000) per claim and in general aggregate providing protection against inappropriate workplace conduct, including: wrongful termination, discrimination, harassment, retaliation, defamation, invasion of privacy, failure to promote, deprivation of a career opportunity, and negligent evaluation.

- (b) <u>Certificates of Insurance</u>. Certificates of insurance ("<u>COI</u>") evidencing all coverage described in this <u>Section 10.1(b)</u> shall be furnished to COMPANY or BPPR (i) upon execution of this Master Agreement; (ii) annually, within forty-five (45) days following the insurance policy renewal; (iii) whenever the insurer or the insurance coverage changes; or (iv) upon COMPANY's or BPPR's request and will include the following:
  - (1) The COI will require thirty (30) days' advance written notice to COMPANY and BPPR before cancellation of, material change to, potential exhaustion of aggregate limits of, or non-renewal of the required insurance coverage(s).
  - (2) The COI will state that the respective EVERTEC insurance will respond on a primary basis without contribution from any other insurance of COMPANY, BPPR, or their respective Subsidiaries.
- (c) EVERTEC shall maintain the coverages required in this <u>Section 10.1(c)</u> from insurance carriers that have an AM Best rating of A-VII or better for risks insured in Puerto Rico and an AM Best rating of A VIII for risks insured outside Puerto Rico; provided that if an insurance carrier's rating changes to a rating below A-VII or A VIII (as applicable), EVERTEC shall use Best Efforts to promptly replace such insurance carrier to one that has a AM Best rating of A-VII or A VIII (as applicable) and in all cases shall replace it prior to the next annual renewal of the applicable coverage.
- (d) Insurance deductibles or retentions shall not exceed three million dollars (\$3,000,000) per insurance policy unless approved in writing by COMPANY and BPPR.
- (e) All liability insurance, except for Crime, Employment Practices Liability, Workers' Compensation, Professional Liability and Cyber Liability required under this Master Agreement must include an additional insured endorsement, without containing any special limitations on the scope of the endorsement, specifying COMPANY and BPPR and their officers, employees and agents as additional insureds, including additional insured status with respect to liability arising out of ongoing operations and completed operations, but only with respect to EVERTEC's activities to be performed under this Master Agreement.
- (f) EVERTEC shall waive rights of subrogation which EVERTEC or any insurer of EVERTEC may acquire against COMPANY or BPPR by virtue of the payment of any Loss; and EVERTEC will obtain any endorsement that may be necessary to affect this waiver of subrogation, but this provision applies regardless of whether or not COMPANY or BPPR has received a waiver of subrogation endorsement from EVERTEC or EVERTEC's insurer(s); provided, however, that (x) any such waivers shall not be extended to cover losses caused by acts of any COMPANY and BPPR or their affiliates, subsidiaries or parent companies and their agents or employees, which are not connected with the operations being conducted by EVERTEC or any third-party hired by EVERTEC and their principals so stipulated and then only, in so far as is necessary to meet the requirements of this Master Agreement; and (y) any such endorsements shall not extend, modify, increase limits of, or otherwise alter the coverage provided by any policy applicable within this Master Agreement.

- (g) If any of the required insurance is on a claims made basis and does not include an extended reporting period of at least 36 months, EVERTEC shall maintain either tail coverage or continuous claims made liability coverage, for a minimum of 36 months following the termination of this Master Agreement.
- (h) If EVERTEC outsources to any Third Party (including a Subcontractor) any part of the services it renders or the products it supplies to COMPANY or BPPR, then EVERTEC's required insurance policies must be endorsed to provide coverage for liabilities that may arise through such Third Party or Subcontractor to the same extent as if the work were performed, or the goods provided, by EVERTEC. In the alternative, EVERTEC must submit to COMPANY and BPPR proof of the Third Party's or Subcontractor's insurance under the same coverage and limits as required from EVERTEC. Any claims arising from the work or product provided by the Third Party or Subcontractor will be paid first through EVERTEC's insurance and, to the extent not covered, then by such Third Party's or Subcontractor's insurance.
- (i) The insolvency, bankruptcy or failure of any insurance company providing coverage as required herein shall not relieve either party of its obligations to maintain adequate insurance as provided in this <u>Section 10.1</u>.
- 10.2 <u>Indemnity</u>. Each party (the "<u>Indemnifying Party</u>") hereby agrees to indemnify, defend, protect and hold harmless the other parties, their Affiliates and their respective Representatives, suppliers, Third Party information providers, Subcontractors and permitted assigns and successors in interest (collectively, the "<u>Indemnified Party</u>") from and against any Losses incurred or suffered by, or asserted against, such Indemnified Party directly or indirectly in relation to or arising from:
- (a) subject to Section 2.3 of <u>Exhibit C</u> hereof with respect to any failure by EVERTEC to meet its Service Levels, any breach of this Master Agreement, including any breach of representation or warranty, by the Indemnifying Party;
  - (b) any claim brought by any Third Party against an Indemnified Party based on the Indemnifying Party's use or provision of the Services;
  - (c) the Indemnified Party's compliance with the Indemnifying Party's Specifications or instructions;
- (d) acts or omissions of the Indemnifying Party and its Representatives in connection with the installation, maintenance, presence, use or removal of equipment or software not provided by the Indemnified Party;
- (e) with respect to the COMPANY or BPPR as the Indemnifying Party, claims by Third Parties that EVERTEC's use of Intellectual Property provided by COMPANY or BPPR infringe the Third Party's Intellectual Property rights, except to the extent the infringement claim is based on (i) use of the Intellectual Property for any purpose not permitted in this Master Agreement; or (ii) enhancements or other modifications made by EVERTEC to the Intellectual Property, other than at the specific written direction of COMPANY, BPPR or any of their respective Subsidiaries or Affiliates;

- (g) with respect to the EVERTEC as the Indemnifying Party, a Data Breach occurring due to a breach of this Master Agreement, gross negligence, fraud or willful misconduct by EVERTEC or its Representatives; provided that EVERTEC's indemnification obligation under this <u>Section 10.2(g)</u> with respect to Data Breaches shall be subject to the Data Cap;
  - (h) the Indemnified Party's use or provision of the Services or data supplied by the Indemnifying Party; and
- (i) claims against the Indemnified Party for damage to, or loss of use of property of Third Parties and/or injury or death of any person to the extent that such damage, injury or death is caused by the negligent act or omission of the Indemnifying Party.

Notwithstanding anything in this Section 10.2 to the contrary, (i) with respect to EVERTEC as the Indemnifying Party, this Article Ten shall not apply to claims relating to infringement of any Intellectual Property right of a Third Party, which claims shall be exclusively the subject of Section 7.4, (ii) with respect to COMPANY and BPPR as the Indemnifying Party, (A) any claims related to infringement of any Intellectual Property right of a Third Party shall be exclusively the subject of Section 10.2(e); and (B) BPPR's indemnification obligation and liability with regard to infringement claims related to a BPPR Pass Through License shall be subject to and limited solely to the following: (x) if COMPANY or BPPR receives notice of an infringement claim against an EVERTEC Indemnified Party related to a BPPR Pass Through License, if requested by EVERTEC, COMPANY or BPPR shall use Best Efforts to pursue its rights, if any, for indemnification or other monetary remedies against the applicable Third Party licensor in accordance with the terms set forth in the BPPR Pass Through License; and (y) monetary awards actually received by COMPANY or BPPR, less the amount of all fees, claims, costs and expenses (including attorneys' fees) incurred by COMPANY or BPPR in connection with pursuing such claim against a Third Party provider for infringement, shall be allocated between COMPANY and BPPR and its respective Affiliates, on the one hand, and EVERTEC, on the other, in proportion to the damages and costs sustained by such party due to the BPPR Pass Through License.

- 10.3 Indemnification Procedures. With respect to claims covered by Section 10.2 above, the following procedures will apply:
- (a) *Notice*. Promptly after receipt by an Indemnified Party of notice of the commencement or threatened commencement of any civil, criminal, administrative or investigative action or proceeding involving a claim in respect of which the Indemnified Party will seek indemnification pursuant to this <u>Article Ten</u>, the Indemnified Party will notify the Indemnifying Party of such claim in writing. The failure of Indemnified Party to so notify an Indemnifying Party will relieve Indemnifying Party of its obligations under this <u>Section 10.3</u> to the extent that Indemnifying Party can demonstrate damages attributable to such failure. Within fifteen (15) days following receipt of written notice from the Indemnified Party relating to any claim, but no later than fifteen (15) days before the date on which any response to a complaint or summons is due, the Indemnifying Party will notify the Indemnified Party in writing if the Indemnifying Party elects to assume control of the defense and settlement of that claim (a "Notice of Election").

- (b) Procedure Following Notice of Election. If the Indemnifying Party delivers a Notice of Election relating to any claim within the required notice period, the Indemnifying Party will be entitled to have sole control over the defense and settlement of such claim; provided that (i) the Indemnified Party will be entitled to participate in the defense of such claim and to employ counsel at its own expense to assist in the handling of such claim; and (ii) the Indemnifying Party will notify the Indemnified Party before ceasing to defend against such claim, and will not compromise or settle such claim without the Indemnified Party's prior written consent if such compromise or settlement would impose a penalty or limitation upon the Indemnified Party, including, an injunction or other equitable relief, or such compromise or settlement does not include the release of the Indemnified Party from all liability arising from or relating to such claim. After the Indemnifying Party has delivered a Notice of Election relating to any claim, the Indemnifying Party will not be liable to the Indemnified Party for any legal expenses incurred by the Indemnified Party in connection with the defense of that claim. In addition, the Indemnifying Party will not be required to indemnify the Indemnified Party for any amount paid or payable by the Indemnified Party in the settlement of any claim for which the Indemnifying Party has delivered a timely Notice of Election if such amount was agreed to without the written consent of the Indemnifying Party.
- (c) Procedure Where No Notice of Election Is Delivered. If the party which is the Indemnifying Party does not deliver a Notice of Election relating to any claim within the required notice period, the Indemnified Party will have the right to defend the claim in such manner as it may deem appropriate, and the failure of the Indemnifying Party to deliver such Notice of Election will not affect the indemnification obligations of such party under this Master Agreement.
- (d) Cooperation. When seeking indemnification, the Indemnified Party will at all times reasonably cooperate with the Indemnifying Party in the defense or settlement of any claim which is subject to this <u>Article Ten</u>.
- (e) Entitlement to Payment. In the event an Indemnifying Party elects not to assume control of the defense and settlement of that claim, the Indemnified Party will be entitled to payment by the Indemnifying Party upon the Indemnified Party's settlement of the claim or the adjudication of liability, whichever first occurs.

## ARTICLE ELEVEN- SERVICES FOR CLIENTS

- 11.1 <u>Scope of Services</u>. EVERTEC acknowledges that the Services are intended to enable COMPANY, BPPR, and their respective Subsidiaries to manage effectively certain of:
  - (a) COMPANY, BPPR, and their respective Subsidiaries' internal operations, and
- (b) COMPANY, BPPR, and their respective Subsidiaries' performance and delivery of services and products for COMPANY, BPPR, and their respective Subsidiaries' respective Clients (hereinafter collectively referred to as the "Popular Services and Products").

Provided EVERTEC is given the opportunity to (i) coordinate with COMPANY and BPPR and approve the Specifications for the Services, COMPANY, BPPR, or one of their respective Subsidiaries proposes to offer under a written agreement between COMPANY, BPPR, or one of their respective Subsidiaries and a Client that includes the provision of such Services ("Client Agreements"); and (ii) review and accept changes to any applicable operating manuals of COMPANY, BPPR, or their respective Subsidiaries that impact EVERTEC with respect to its provision of the Services, EVERTEC shall use Best Efforts to ensure that the Services are provided in a manner that allows COMPANY, BPPR, and their respective Subsidiaries to act in accordance with the corresponding Client Agreements and operating manuals for each of the Popular Services and Products. In the event that COMPANY, BPPR, or their respective Subsidiaries make any change to any Popular Service and Product that requires a change or modification to the Services, such change shall be processed and implemented pursuant to this Master Agreement.

11.2 Special Representations Regarding EVERTEC's Performance of Services under Service Agreements with Governmental Agencies. In connection with any Client Agreement wherein the Client is a Governmental Authority of the Commonwealth of Puerto Rico, EVERTEC shall comply with all applicable Legal Requirements relating to government procurement and contracting.

# 11.3 Client Agreements.

- (a) EVERTEC will have no liability or obligation under any Client Agreement, whether through an outsourcing arrangement or through reselling of the Services; provided, however, that EVERTEC shall be responsible to COMPANY, BPPR, and their respective Subsidiaries for any Losses caused by a breach of any of its obligations under this Master Agreement. COMPANY, BPPR, and their respective Subsidiaries will have no authority to bind EVERTEC to any terms or conditions of the Service in connection with or as part of any Client Agreement, except as otherwise provided by the parties in writing.
- (b) EVERTEC agrees to assist COMPANY, BPPR, and their respective Subsidiaries with the investigation of any claims arising under a Client Agreement. Furthermore, EVERTEC will refer to COMPANY, BPPR, and their respective Subsidiaries any complaint that is received by EVERTEC, in which case, EVERTEC will be notified of the results of any investigation performed by COMPANY, BPPR, or their respective Subsidiaries within five (5) days following the receipt of the referral for investigation.
- (c) COMPANY, BPPR, and their respective Subsidiaries are responsible for any acts or omissions by Clients that, if performed by COMPANY, BPPR, or their respective Subsidiaries, would constitute a breach of this Master Agreement. COMPANY, BPPR, and their respective Subsidiaries are responsible for all fees and expenses that are payable to EVERTEC under this Master Agreement regardless of whether COMPANY, BPPR, and their respective Subsidiaries receive payment from the Client for the Services provided. Any charge-back or credit arrangements between COMPANY, BPPR, and their respective Subsidiaries and a Client are the responsibility of COMPANY, BPPR, and their respective Subsidiaries, and do not affect COMPANY, BPPR, and their respective Subsidiaries' liability for the payment of such fees and expenses.

(d) Notwithstanding any limit of liability herein, COMPANY hereby agrees to indemnify, defend, protect and hold harmless EVERTEC, its Affiliates and their respective Representatives, suppliers, Third Party information providers, Subcontractors and permitted assigns and successors in interest (collectively, the "EVERTEC Indemnitee") from and against any Losses incurred or suffered by, or asserted against, any EVERTEC Indemnitee directly or indirectly in relation to or arising from any claim brought by any Third Party against an EVERTEC Indemnitee based on (i) a Client's use of the Services in a manner inconsistent with this Master Agreement and (ii) any breach of a Client Agreement by COMPANY, BPPR, or their respective Subsidiaries or a Client to the extent such breach is not attributable to a breach of this Master Agreement by EVERTEC.

[Signature page follows.]

**IN WITNESS WHEREOF,** the parties hereto have caused this Master Agreement to be executed by their duly authorized Representatives as of the date first written above.

Popular, Inc. Evertec Group, LLC

By:/s/ Félix FernándezBy:/s/ Luis RodriguezName:Félix FernándezName:Luis A. Rodriguez

Title: Senior Vice President, Chief Legal and Corporate

Development Officer

Banco Popular de Puerto Rico

By: /s/ Félix Fernández

Name: Félix Fernández
Title: Authorized Officer

[Signature Page to the Master Agreement]



# Popular Completes Acquisition of Key Customer Channels and Enters into Amended Commercial Contracts with Evertec

SAN JUAN, Puerto Rico - (BUSINESS WIRE) - July 1, 2022 - Popular, Inc. ("Popular" or the "Corporation") (NASDAQ: BPOP) announced today that its wholly owned subsidiary, Banco Popular de Puerto Rico ("BPPR"), has completed its previously announced acquisition of certain assets and assumption of certain liabilities used by Evertec Group, LLC ("Evertec Group"), a wholly owned subsidiary of Evertec, Inc. ("Evertec") (NYSE: EVTC), to service certain BPPR channels.

As a result of the closing of the transaction, BPPR acquired from Evertec Group certain critical channels, including BPPR's retail and business digital banking and commercial cash management applications. BPPR also entered into amended and restated service agreements with Evertec Group pursuant to which Evertec Group will continue to provide various information technology and transaction processing services to Popular, BPPR and their respective subsidiaries.

Under the amended service agreements, Popular will have greater optionality to develop and enhance technology platforms and more flexibility to select service vendors, as Evertec Group will no longer have exclusive rights to provide certain of Popular's technology services. This is expected to improve Popular's ability to meet its customer needs in a timely manner. In addition, the amended service agreements are projected to reduce service costs as a result of discounted pricing and lowered caps on contractual pricing escalators tied to the Consumer Price Index. As part of the transaction, BPPR also strengthened its relationship with Evertec in the payments business, including through the incorporation of a revenue sharing structure for BPPR in connection with its merchant acquiring relationship with Evertec.

In connection with the transaction, Popular welcomed approximately 165 employees and contractors from Evertec that support the servicing of the acquired assets, strengthening and deepening Popular's in-house technology bench.

"This transaction allows Popular to accelerate its ongoing digital transformation and continue improving our clients' experience. With the ability to manage our key customer channels and greater flexibility to choose the best technology partners, we will be able to more quickly enhance the services that we offer through our digital platforms, as well as the sales and account maintenance processes we offer through all our channels. We aspire to provide an omnichannel experience that meets all of our clients' needs and expectations," said Ignacio Alvarez, President and Chief Executive Officer of Popular, Inc.

As consideration for the transaction, BPPR delivered to Evertec Group 4,589,169 shares of Evertec common stock valued at closing at \$169 million (based on Evertec's stock price on June 30, 2022 of \$36.88), resulting in an after-tax gain of approximately \$112 million. The financial benefits of the transaction during the first full year following the closing of the transaction are expected to be offset as a result of the elimination of Popular's earnings from its investment in Evertec, which is

currently accounted for under the equity method. However, the financial effect of the transaction is expected to be accretive during future years due to, among other things, incremental merchant acquiring revenue sharing income and future price reductions in continuing services as a result of the amended service agreements with Evertec.

In terms of capital, the transaction results in a negative impact of approximately \$55 million in Popular's tangible book value as a result of the net effect of the after-tax gain of the Evertec shares used as consideration for the transaction (approximately \$112 million), minus approximately \$167 million in goodwill and other intangible assets recognized by the Corporation in connection with the transaction and the effect of purchase accounting-related adjustments.

As a result of the transfer of the shares used as consideration for the transaction, Popular's ownership stake in Evertec was reduced from approximately 16.3% to approximately 10.6% at the closing of the transaction. In connection with the transaction, Popular has agreed to further reduce its voting interest in Evertec to no more than 4.5%, whether through selling shares of Evertec common stock or a conversion of such shares into non-voting preferred stock within 90 days of the closing of the transaction. The subsequent reduction of Popular's voting interest in Evertec to no more than 4.5% would result in \$177 million in after-tax gains as a result of mark-to-market accounting, assuming a value per share of \$36.88, Evertec's share price on June 30, 2022.

Popular expects to sell down its stake in Evertec to no more than 4.5% and intends to return to shareholders, via common stock repurchases, any after-tax gains resulting from such sale, subject to the receipt of regulatory approvals. Assuming Popular reduces its remaining stake in Evertec to 4.5% via sale and deploys the resulting after-tax gains (\$102 million, assuming a value per share of \$36.88, Evertec's share price on June 30, 2022) to execute common stock repurchases (assuming a value per share of \$76.93, Popular's share price on June 30, 2022), the aggregate pro forma impact of the transaction, the effect of mark-to-market accounting on the remainder of Popular's stake in Evertec and the execution of the contemplated share repurchase would result, during 2023, in an accretion of \$0.16 and \$1.64 in earnings per share and tangible book value per share, respectively.

# About Popular, Inc.

Popular, Inc. (NASDAQ: BPOP) is the leading financial institution in Puerto Rico, by both assets and deposits, and ranks among the top 50 U.S. bank holding companies by assets. Founded in 1893, Banco Popular de Puerto Rico, Popular's principal subsidiary, provides retail, mortgage and commercial banking services in Puerto Rico and the U.S. Virgin Islands. Popular also offers in Puerto Rico auto and equipment leasing and financing, investment banking, broker-dealer and insurance services through specialized subsidiaries. In the mainland United States, Popular provides retail, mortgage and commercial banking services through its New York-chartered banking subsidiary, Popular Bank, which has branches located in New York, New Jersey and Florida.

#### **Cautionary Note Regarding Forward-Looking Statements**

This press release contains "forward-looking statements" within the meaning of the U.S. Private Securities Litigation Reform Act of 1995, including without limitation those regarding Popular's acquisition of certain assets and assumption of certain liabilities from Evertec and the transactions described in this press release (the "Transaction") and the Corporation's business, financial condition, results of operations, plans, objectives and future performance. These statements are not guarantees of future performance, are based on the current expectations of Popular's management and, by their nature, involve risks, uncertainties, estimates and assumptions. Potential factors, some of which are beyond the Corporation's control, could cause actual results to differ materially from those expressed in, or implied by, such forward-looking statements. Risks and uncertainties include, without limitation, the effect of competitive and economic factors, and our reaction to those factors, the adequacy of the allowance for loan losses, delinquency trends, market risk and the impact of interest rate changes, capital market conditions, capital adequacy and liquidity, the effect of legal and regulatory proceedings, new accounting standards on the Corporation's financial condition and results of operations, the scope and duration of the COVID-19 pandemic (including the appearance of new strains of the virus), actions taken by governmental authorities in response thereto, and the direct and indirect impact of the pandemic on the Corporation, our customers, service providers and third parties. Other factors include Popular's ability to successfully transition and integrate the assets acquired as part of the Transaction, related operations, employees and third party contractors, unexpected costs, including, without limitation, costs due to exposure to any unrecorded liabilities or issues not identified during the due diligence investigation of the Transaction or that are not subject to indemnification or reimbursement by Evertec; operational risks that may affect Popular and other risks arising from the acquisition of the acquired assets or by adverse effects on relationships with customers, employees and service providers and business and other risks arising from the extension of Popular's current commercial agreements with Evertec. All statements contained herein that are not clearly historical in nature, are forward-looking, and the words "anticipate," "believe," "continues," "expect," "estimate," "intend," "project" and similar expressions, and future or conditional verbs such as "will," "would," "should," "could," "might," "can," "may" or similar expressions, are generally intended to identify forward-looking statements.

More information on the risks and important factors that could affect Popular's future results and financial condition is included in our Annual Report on Form 10-K for the year ended December 31, 2021 and in the Quarterly Report on Form 10-Q for the quarter ended March 31, 2022. Our filings are available on Popular's website (www.popular.com) and on the Securities and Exchange Commission website (www.sec.gov). Popular assumes no obligation to update or revise any forward-looking statements or information which speak as of their respective dates.

#### **Contacts**

Popular, Inc.

Investor Relations: Paul J. Cardillo, 212-417-6721 Investor Relations Officer pcardillo@popular.com

or

Media Relations:
María Cristina González Noguera, 917-804-5253
Executive Vice President and Chief Communications & Public Affairs Officer mc.gonzalez@popular.com

Financial (English): P-EN-FIN

# **Document and Entity Information**

Jul. 01, 2022

**Document And Entity Information [Line** 

**Items**]

Amendment Flag false

Entity Central Index Key 0000763901

Document Type 8-K

<u>Document Period End Date</u>

Entity Registrant Name

Jul. 01, 2022

POPULAR, INC.

Entity Incorporation State Country Code PR

Entity File Number 001-34084 Entity Tax Identification Number 66-0667416

Entity Address, Address Line One 209 Muñoz Rivera Avenue

Entity Address, City or Town Hato Rey

Entity Address, Country PR Entity Address, Postal Zip Code 00918 City Area Code (787)Local Phone Number 765-9800 Written Communications false **Soliciting Material** false Pre Commencement Tender Offer false Pre Commencement Issuer Tender Offer false

Common Stock [Member]

**Entity Emerging Growth Company** 

**Document And Entity Information [Line** 

**Items**]

Security 12b Title Common Stock (\$0.01 par value)

Trading SymbolBPOPSecurity Exchange NameNASDAQ

Cumulative Preferred Stock [Member]

**Document And Entity Information [Line** 

**Items**]

Security 12b Title 6.125% Cumulative Monthly Income Trust Preferred

false

Securities

Trading SymbolBPOPMSecurity Exchange NameNASDAQ

```
{
"da36340d8k.htm": {
"da36340d8k.htm": {
"axisGtandard": 1,
"contextCount": 3,
"dta": {
"definitionLink": {
"local": {
"local"
                                              ),
"inline": {
  "local": [
  "d335340d8k.htm"
                                                               presentationLink": {
"local": [
"bpop-20220701_pre.xml"
]
                                              ),
"schema": {
"local": [
"bpop-20220701.xsd"
                                        "miceaf": [
"peop-o202001.sad"
"peop-o202001.sad"
"peop-o202001.sad"
"micep://www.sbd..org/2003/sbdr-instance-2003-12-31.sad",
"micep://www.sbd..org/2003/sbdr-instance-2003-12-31.sad",
"micep://www.sbd..org/2003/sbdr-instance-2003-12-31.sad",
"micep://www.sbd..org/2003/sbdr-id-2003-2-31.sad",
"micep://www.sbd..org/2003/sbd-id-2003-2-31.sad",
"micep://www.sbd..org/2003/sbd-id-2003-2-31.sad",
"micep://www.sbd..org/2003/sbd-id-2003-2-31.sad",
"micep://www.sbd..org/2003/sbd-id-2003-2-31.sad",
"micep://www.sbd..org/2004/sbd-id-2003-2-31.sad",
"micep://wbd..org/2004/sbd..org/dbd-id-2003-2-31.sad",
"micep://wbd..org/dbd-id-2003-2-31.sad",
"micep://wbd..org/dbd-id-2003/sbd-id-2001-0-31.sad",
"micep://sbd.icab..org/sbd-id-2003/sbd-id-2001-0-31.sad",
"micep://sbd.icab..org/sbd-id-2003/sbd-id-2003-0-301.sad",
"micep://sbd.icab..org/sbd-id-2003/sbd-id-2003-0-301.sad",
"micep://sbd.icab..org/sbd-id-2003/sbd-id-2003-0-301.sad",
"micep://sbd.icab..org/sbd-id-2003-0-301.sad",
"micep://sbd.icab..org/sbd-id-2003-0-301.sad",
"micep://sbd.icab..org/sbd-id-2003-0-301.sad",
"micep://sbd.icab..org/sbd-id-2003-0-301.sad",
"micep://sbd.icab..org/sbd-id-2003-0-301.sad",
"micep://sbd.icab..org/sbd-id-2003-0-301.sad",
"micep://sbd.icab..org/sbd-id-2003-0-301.sad",
"micep://sbd.icab..org/sbd-id-2003-0-301.sad",
"micep://sbd.icab..org/sbd-id-2003-0-301.sad",
                            ],
"elementCount": 29,
"entityCount": 1,
"hidden": [
"http://xbrl.sec.gov/dei/2021": 2,
"total": 2
                       "Teleground and control of the contr
                                                         "sablit: "false"
"jourphys! tolonoment,"
"labefault: "true",
"labe
                                                    "html"
"baskeder" "633346981.htm",
"contentRe": "duration_2022-07-01_to_2022-07-01",
"contentRe": "duration_2022-07-01_to_2022-07-01",
"class": "sec-102",
"lass": "sec-102",
"s
                                  ],
"localname": "DocumentAndEntityInformationLineItems",
"souln": "http://www.popular.com/20220701",
"presentation": [
"http://www.popular.com//20220701/taxonomy/role/Docum
                                                         l,
"xbrltype": "stringItemType"
                                              ],

**Thompo DocumentAndEmittyInformationTable*; {

**sunt_ref*: []

**sung*: [

**sung*: [

**role*: [

**slad*: "Document And Entity Information [Table]",

**creatable*: "Document And Entity Information [Table]",
                                                         ],
"localname": "DocumentAndEntityInformationTable",
"nsur!": "http://www.popular.com/20220701",
"presentation": [
"http://www.popular.com//20220701/taxonomy/role/D
                                                         l,
"xbrltype": "stringItemType"
                                              The day and the state of the st
                                                    ),
"localname": "AmendmentFlag",
"nauri": "http://xxrl.sec.gov/dei/2021",
"presentation": [
"http://www.popular.com//20220701/taxonomy
],
"xxrltype": "booleanItemType"
                                              "abritype": "booleanttemType"

| "dei CityAreaCode": {
    "auth ref": [];
    "lang": {
    "se-u-i;
    "decumentation": "Area code of city",
    "label": "City Area Code",
    "terseLabel": "City Area Code",

                                                         ),
"localname": "CityAreaCode",
"nsuri": "http://xbrl.sec.gov/dei/2021",
"presentation": [
"http://www.popular.com//20220701/taxono
                                              "Moretype" "normalizestringitemryp)
"del Coverabtract": {
"del Coverabtract": {
"lang" |
"lang" |
"slap" |
"close: {
"documentation": "Cover page.",
"label": "Cover [Abstract]",
"terselabel": "Cover [Abstract]",
                                                         },
"localname": "CoverAbstract",
"nauri": "http://xbrl.sec.gov/dei/2021",
"xbrltype": "stringItenType"
```

```
),
"localname": "DocumentPeriodEndDate",
"nsuri": "http://xbrl.sec.gov/dei/2021",
"presentation": [
"http://www.popular.com//20220701/taxon
     "xbrltype": "dateItemType"
     "MilityPe": "cursivening"

(dis,)concentTyPe": [
"such; est": [],
"such; e
     ;
"localname": "DocumentType",
"nsuri": "http://xbrl.sec.gov/dei/2021",
"presentation": [
"http://www.popular.com//20220701/taxon/
                                                                                                                                                                                                                                                                                                                                                                    mentDocumentAndEntityInformation*
        ],
"xbrltype": "submissionTypeItemType"
                      "Fole": (
"documentation": "Address Line 1 such as Attn, Building Name, Street Name",
"label": "Entity Address, Address Line One",
"terselabel": "Entity Address, Address Line One",
        ],
"localname": "EntityAddressAddressLinel",
"nsuri": "http://xbrl.sec.gov/dei/2021",
"presentation": [
"http://www.popular.com//20220701/taxonor
        l,
"xbrltype": "normalizedStringItemType"
     , dei EntityAddressCityOrTown": {
  "auth ref": [],
  "lang": [
  "en-us": {
  "documentation": "Name of the
                      "role": {
    "documentation": "Name of the City or Town",
    "label": "Entity Address, City or Town",
    "terselabel": "Entity Address, City or Town"
     ],
"iocalname": "EntityAddressCityOrTown",
"nourl": "http://xbrl.sec.gov/dei/2021",
"presentation": [
"http://www.popular.com//20220701/taxono],
"whrtype": "normalizedStringItemType"
"mer.ype : mormalization.ting.com.ype
"doi.EntityAddressCounty": {
"such_ent': []
  ),
"localname": "EntityAddressCountry",
"nsuri": "http://xbrl.sec.gov/dei/2021",
"presentation": [
"http://www.popular.com//20220701/taxonomy]
                                                                                                                                                                                                                                                                                                        role/DocumentDocumentAndEntityInformation
  "mortypes" "countrycode tearype"
"dad EntityddressPortalipCode"; {
"such eq": [,"
"lang": [,"
"rola": [
"rola": [
"clos": [,"
"documentation": "Code for the postal or zip code",
"labal": "Entity Address, Postal zip code",
"resealche": "Entity Address, Postal zip code",
"exerables": "Entity Address, Postal zip code",
     ],
"localname": "EntityAddressPostalZipCode",
"nsuri": "http://xbrl.sec.gov/dei/2021",
"presentation": [
"http://www.popular.com//20220701/taxonom]
        l,
"xbrltvpe": "normalizedStringItemTvpe"
     'dei_EntityCentralIndexKey": [
"auth_ref": [
"r5"
"isp": [
"inap": [
"inap":
     ),
"localname": "EntityCentralIndexKey",
"nsuri": "http://xbrl.sec.gov/dei/2021",
"presentation": [
"http://www.popular.com//20220701/taxon/
     1, "xbrltvpe": "centralIndexKevItemTvpe"
     /
dei_EntityEmergingGrowthCompany": {
   "auth_ref": {
    "r5"
  ].
"Hang": {
"Ann's: {
"An
     },
"localname": "EntityEmergingGrowthCompany",
"nsur!": "http://xbrl.sec.gov/dei/2021",
"presentation": [
"http://www.popular.com//20220701/taxonomy
     l,
"xbrltype": "booleanItemType"
     outspar : Toursantempye

dom, Entityllabeber: {
    "muth_ser" [],
    "aud'; [],
    "rola";
    "rola";
    "rola";
    "rola";
    "rola";
    "comission file number. The field allows up to 17 characters. The prefix may contain 1-3 digits, the seq
    "label";    "Entity File Number",
    "resealabel";    "Entity File Number",
    "resealabel";    "Entity File Number",

        },
"localname": "EntityFileNumber",
"nsuri": "http://xbrl.sec.gov/dei/2021",
"presentation": [
"http://www.popular.com//20220701/taxono
        l,
"xbrltype": "fileNumberItemType"
     our type 'limentori tentype
ded, EntityEncoprationStateCountryCode": [
"muth ref": []
"lang": [
"role": [
"documentation": "Two-character EDGAR code representing '
"label": TEntity Incorporation State Country Code",
"tersetabel": "Entity Incorporation State Country Code",
"tersetabel": "Entity Incorporation State Country Code"
  ],
"iocalname": "EntityIncorporationStateCountryCode",
"mauri": "http://xbrl.sec.gov/de1/2021",
"presentation:
"http://www.popular.com//20220701/taxonomy/role/Doc
]"sbrltype": "edgarStateCountryStemSype"
     },
"localname": "EntityRegistrantName",
"nsuri": "http://xbrl.sec.gov/dei/2021",
"presentation": [
"http://www.popular.com//20220701/taxon],
     l,
"xbrltype": "normalizedStringItemType"
```

```
"auth_ref": [
"r5"
"ES"

**Samp"; |

**Samp"; |

**Cole*; |

           },
"localname": "EntityTaxIdentificationNumber",
"nsurf": "http://xbri.sec.gov/dei/2021",
"presentation": [
"http://www.popular.com//20220701/taxonomy/s
   |
| "localname": "LocalPhoneNumber",
| "nauri": "http://xbrl.sec.gov/dei/2021",
| "presentation": [
| "http://www.popular.com//20220701/taxono
| 1.
           1,
"xbrltype": "normalizedStringItemType"
       ),
"dei PreCommencementIssuerTenderOffer": [
"auth_ref": [
"22"
       ]."
**Lang": {
**Lang": {
**Lang": {
**Lang": {
**Lang": {
**Lang: {
**Commencement langer Theology of the Form B-K filing is intended to satisfy the filing obligation of the regist
**Labbl": *Pro Commencement Langer Theology Office",

**Labbl": *Pro Commencement 
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     unications pursuant to Rule 13e-4(c) under the Exchange Act.",
           },
"localname": "PreCommencementIssuerTenderOffer",
"nsurf": "http://xbrl.sec.gov/dei/2021",
"presentation":
"http://www.popular.com//20220701/taxonomy/role,
           l,
"xbrltype": "booleanItemType"
   SHIRLY TWO COMMENCEMENT TWO COMMENCEMENT AND ALSO COMMENT THE COMMENCEMENT THE COMMENT 
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     unications pursuant to Rule 14d-2(b) under the Exchange Act."
           ),
"localname": "PreCommencementTenderOffer",
"nsuri": "http://xbrl.sec.gov/dei/2021",
"presentation": [
"http://www.popular.com//20220701/taxonom
   "dei_Security12bTitle": {
   "auth_ref": [
   "r0"
"go" |
"Anang" |
       },
"localname": "Security12DTitle",
"nsur!": "http://sbtl.sec.gov/dei/2021",
"presentation": [
   "http://www.popular.com//20220701/taxonol],
"sbrltype": "securityTitleItemType"
   "Zi"
"lang":[
"en-un":[
"Fold":[
"document:[
"document." decurity Enchange on which a security is registered.",
"document." "docurity Enchange Name".
"crselabel": "Security Enchange Name"
           ),
"localname": "SecurityExchangeName",
"nsuri": "http://xbrl.sec.gov/dei/2021",
"presentation": [
"http://www.popular.com//20220701/taxon
           1,
"xbrltype": "edgarExchangeCodeItemType"
           "auth_ref": [
       "Panap": {
    "sun-up": {
        "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {
            "sun-up": {

                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            as soliciting material pursuant to Rule 14a-12 under the Exchange Act.",
           ],
"localname": "SolicitingMaterial",
"nsuri": "http://xbrl.sec.gov/dei/2021",
"presentation": [
"http://www.popular.com//20220701/taxon].
           l,
"xbrltype": "booleanItemType"
           Texts.ppe : Toursentemppe
ded.TextIngSpund)**: [
**such prf* [],
**lang**: [
**rola*:
**rola*: [
**
       ],
"localname": "TradingSymbol",
"nsur!": "http://xbrl.sec.gov/dei/2021",
"presentation": [
"http://www.popular.com//20220701/taxonom
],
"xbrltype": "tradingSymbolItemType"
"Set "Lines": [
"Annual [

       ),
"localname": "WrittenCommunications",
"nsur!": "http://whrl.sec.gov/dei/2021",
"presentation": [
"http://www.popular.com//20220701/taxon/],
"xbrltype": "booleanItemType"
   "mbrityme": "bocieanitemrype"

"us-gaap ClassofStockDomain": {
  "auth rest": [],
  "lang": {
  "sole": [
  "sole": [
  "label": "Class of Stock [Domain]",
  "terseLabel": "Class of Stock [Domain]",

           ],
"localname": "ClassOfStockDomain",
"nsurf": "http://fasb.org/us-gaap/2021-01-31",
"presentation": [
"http://www.popular.com//20220701/taxonomy/role]
               l,
"xbrltype": "domainItemType"
           ,uu-gaap_CommonStockNember": {
  "auth_ref": {},
  "ang": {
  "en-ua": {
  "sole": {
  "label": "Common Stock [Member]",
  "terselabel": "Common Stock [Member]",

               },
"localname": "CommonStockMember",
"nsuri": "http://fasb.org/us-gaap/2021-01-31",
"presentation": [
```

```
"http://www.popular.com//2012/01/taxonomy/role/DocumentDocumentAndEntityInformation"
| "http://www.popular.com//2012/01/taxonomy/role/DocumentDocumentAndEntityInformation"
| "statitype" - "domaintitwPreferredStockNeeber" |
| "statitype" - "domaintitwPreferredStockNeeber",
| "statitype" - "domaintitwPreferredStockNeeber",
| "statitype" - "domaintitwPreferredStockNeeber",
| "statitype" - "domaintityPreferredStockNeeber",
| "statitype" - "statitype //domaintityPreferredStockNeeber",
| "statitype / "domaintityPreferredStockNeeber",
| "statitype / "domaintityPreferredStockNeeber",
| "statitype / "statitype / domaintityPreferredStockNeeber / "statitype / domaintityP
```